### MODIFICATION NO. 4 TO OHIO COMMUNITY SCHOOL CONTRACT BY and BETWEEN Educational Service Center of Lake Erie West ("Sponsor" or "ESCLEW") AND Pathway School of Discovery ("Governing Authority" OR "School")

**WHEREAS**, the ESCLEW and the Governing Authority entered into an amended and restated Community School Contract ("Contract") effective on July 1, 2012; and

WHEREAS, the ESCLEW and the Governing Authority agree to the following modifications;

**NOW THEREFORE**, the parties modify the Contract as follows:

### 1. Article I, Section 1.7.

- a. Replace subsection (b) with the following: "Monitor and evaluate the academic and fiscal performance and the organization and operation of the School on at least an annual basis, which evaluation shall be based on the performance requirements set forth in **Attachment 11.6**, state report cards, and any other analysis conducted by the Ohio Department of Education or the Sponsor;"
- b. Add the following subsection: "Offer other activities, as determined by the Sponsor, specifically designed to benefit the school;
- c. Add the following subsection: "Report on the amounts and types of expenditures made to provide monitoring, oversight, and technical assistance to sponsored schools, pursuant to the specific requirements of R.C. 3314.025.

The rest of Section 1.7 remains as originally written in the Contract.

2. Article II, Section 2.1 shall be replaced in its entirety with the following:

"Governing Authority Members. The Governing Authority (its Board of Directors, Directors, or Board) must contain at least five Directors (members). All Governing Authority members must be pre-approved by the Sponsor. No Director may serve on the Board if restricted from doing so by R.C. 3314.02(E) or any other law, rule, or regulation. The Sponsor shall be promptly notified in writing of any changes in members, including names of resignations and changes to contact information, within five (5) business days of such change. Attached as Attachment 2.1 are the names, home or work addresses (not the address of the School), and e-mail addresses of the current members of the Governing Authority and a description of the process by which the members of the Governing Authority shall be selected and removed in the future, if not in the Code of Regulations."

3. Article II, Section 2.2. Replace the first sentence with the following: "All members shall complete training on an annual basis on public records and open meetings law. Members new to this School's Board must complete a minimum of five (5) hours of Board training, at least two (2) hours of which are on public records and open meetings law, within three (3) months of being elected or appointed to the Board." The rest of Section 2.2 remains as originally written in the Contract.

- 4. Article II, Section 2.6. At the end, add the following sentence: "The Chief Administrative Officer, other administrative employees of the School, and all individuals performing supervisory or administrative services for the School under a contract with the operator shall complete training on an annual basis on public records and open meetings law." The rest of Section 2.6 remains as originally written in the Contract.
- 5. Article III, Section 3.2 shall be replaced in its entirety with the following:

"Management by Third Party Operators. The School has entered into a contract for management or operation of the School and its curriculum and operations with National Heritage Academies, and such fully-executed contract must be attached as Attachment 3.2. Any changes in the operator or contract shall require prior written approval of the Sponsor and incorporation as Attachment 3.2. The School shall employ an attorney, who shall be independent from the operator, for any services related to the negotiation of the School's contract with the operator or if the operator and School should become adverse to each other in any particular matter.

As the management contract is a part of this Contract, the Governing Authority understands that the Sponsor may discipline the School for any issues of any operator of the School and its administration. The Governing Authority consents to this right of the Sponsor. If the Sponsor deems, at its reasonable discretion, that due to mismanagement, poor governance, or performance, another operator is advisable or necessary, the Sponsor may request, in lieu of discipline, that the Governing Authority interview, select, and enter into a different agreement for such services and such operator or services contract shall be approved by the Sponsor as a modification to this Contract."

### 6. Article III, Section 3.5.

- a. In the first paragraph, add the following sentence: "Disposition of assets shall occur in accordance with chapter 1702 of the Revised Code, as well as R.C. 3314.015(E), 3314.074, and any other applicable laws, rules, or regulations."
- b. Replace subsection 3.5.2 with the following: "In applying the principles stated above, adherence to R.C. 3314.074(A) and (B) should be respected to the fullest extent not in conflict with the above, unless the School is exempt from those provisions pursuant to R.C. 3314.074(D)."

The rest of Section 3.5 remains as originally written in the Contract.

7. Article III, Section 3.6. Add "unless Sponsor is rated exemplary for two or more consecutive years" to the exceptions in the first sentence. The rest of Section 3.6 remains as originally written in the Contract.

## 8. Article IV, Section 4.1.

a. In the first paragraph, add section 3313.721.

b. At the end, add the following paragraph: "If the School operates a preschool program that is licensed by the Ohio Department of Education under R.C. 3301.52 to 3301.59, the School shall comply with R.C. 3301.50 to 3301.59 and the minimum standards for preschool programs prescribed in rules adopted by the state board under R.C. 3301.53."

The rest of Section 4.1 remains as originally written in the Contract.

- 9. Article V, Section 5.2. At the end, add the following sentence: "A detailed description of the facilities, as well as costs, operator addendum, and related parties of any lease or mortgage, is attached as Attachment 5.2." The rest of Section 5.2 remains as originally written in the Contract.
- 10. Article VI, Section 6.3. At the end, add the following sentence: "If the School operates a preschool program that is licensed by the Ohio Department of Education under R.C. 3301.52 to 3301.59, such operation shall be detailed in Attachment 6.3." The rest of Section 6.3 remains as originally written in the Contract.
- 11. Article VI, Section 6.11. Replace the last sentence with the following: "Each school shall comply with the plan for awarding high school credit based on demonstration of subject area competency, and beginning in the 2017-18 school year, with the updated plan that permits students enrolled in seventh and eighth grade to meet curriculum requirements based on subject area competency, as adopted by the state board of education under R.C. 3313.603(J)." The rest of Section 6.11 remains as originally written in the Contract.
- 12. Article VI, Section 6.13. At the end, add the following sentences: "The School's attendance and participation policies must be available for public inspection. The School's attendance and participation records shall be made available, upon request, to the Department of Education, Auditor of State, and Sponsor, to the extent permitted by 20 U.S.C. 1232g, the Family Educational Rights and Privacy Act (FERPA), R.C. 3319.321, and any applicable rule or regulations thereto." The rest of Section 6.13 remains as originally written in the Contract.
- 13. Article VII, Section 7.2. Add the following subparagraph:

"on an annual basis,

- i. verification of completion of annual public records and open meetings law training by each governing authority member, fiscal officer, chief administrative officer, administrative employees of the School, and all individuals performing supervisory or administrative services for the school through an operator contract;
- ii. any finding for recovery issued by the auditor of state against any member of the governing authority, the operator, or any employee of the School;
- iii. updated asset reports showing all assets purchased with public funds and all assets purchased by nonpublic funds, and the ownership of each;
- iv. disclosure statements for each member of the governing authority as filed pursuant to R.C. 3314.02(E)(7); and
- v. a detailed accounting of the nature and costs of the goods and services that the operator provides to the School, pursuant to R.C. 3314.04, if the operator receives more than twenty percent (20%) of the School's gross annual revenues;

The rest of Section 7.2 remains as originally written in the Contract.

- 14. Article IX, Section 9.1. At the end, add the following sentences: "Within five (5) business days of receiving notification from the Auditor, the School shall notify the Sponsor in writing of the time, date, and location of any scheduled meetings with the Auditor. If the governing authority contracts with an entity specializing in audits, including an attorney or accountant, that entity must be independent from the operator with which the School has contracted." The rest of Section 9.1 remains as originally written in the Contract.
- 15. Article IX, Section 9.2 shall be replaced in its entirety with the following:

"Fiscal Services. The School agrees that its fiscal officer shall be its Treasurer, Anne Sherman, contracted through the School's operator, National Heritage Academies. The fiscal services agreement between the School and its fiscal officer is attached to this Contract as Attachment 9.2. If the School and the Sponsor have waived the requirement of employing or contracting with the fiscal officer, pursuant to R.C. 3314.011(D), the current resolution waiving this requirement must be attached to this Contract in Attachment 9.2. Such resolution shall only be valid for one year, and any subsequent resolution adopted by the School must be approved by the Sponsor and submitted to the Ohio Department of Education. Any changes made to the fiscal officer or fiscal services agreement must be reported to the Sponsor within five (5) business days.

The School's fiscal officer shall complete training on an annual basis on public records and open meetings law. The School's fiscal officer must be contractually obligated to assist in all regular, special, or final audits, closing procedures and requirements listed on Attachment 3.5.3 or under R.C. 3314.023, and all Ohio Department of Education rules and procedures, even if the School closes, is not renewed, suspended, or terminated."

- 16. Article IX, Section 9.5. Replace the fourth sentence with the following: "Each year of this Contract, on or before October 31, a school budget that includes all required elements under R.C. 3314.032(C) shall be adopted by the Governing Authority and submitted to the Sponsor." The rest of Section 9.5 remains as originally written in the Contract.
- 17. Article IX, Section 9.6. At the end, add the following sentence: "All moneys borrowed from the School's operator, including facilities loans or cash flow assistance, must be accounted for, documented, and bear interest at a fair market rate." The rest of Section 9.6 remains as originally written in the Contract.
- 18. Article X, Section 10.2. Add the following sub-paragraph:

"A failure of the Governing Authority and/or School or any of its officers, directors, employees, agents, or contractors to meet the obligations of this Contract or any other contract or other obligation between or on behalf of the School and another party."

The rest of Section 10.2 remains as originally written in the Contract.

19. Article XI, Section 11.1. At the end, add the following sentence: "The Governing Authority shall employ an attorney independent from the Sponsor for any negotiation of this Contract or its amendments." The rest of Section 11.1 remains as originally written in the Contract.

### 20. Article XI, Section 11.3. Add the following sub-paragraph:

"For purposes of this Contract, 'sponsor approval' may mean written approval by the ESCLEW Governing Board, the ESCLEW Governing Authority President, Superintendent, or Community Schools Center Direct, or other Sponsor agent authorized by the ESCLEW Governing Board."

The rest of Section 11.3 remains as originally written in the Contract.

21. Article XI, Section 11.6. After the first sentence, add the following: "Performance standards must include, but are not limited to, all applicable report card measures set forth in R.C. 3302.03 or R.C. 3314.017, by which the success of the School will be evaluated by the Sponsor." The rest of Section 11.6 remains as originally written in the Contract.

### 22. Article XI, Section 11.7, subsection (a).

- a. At the end of the second paragraph, add the following sentence: "If the School is not renewed for failure to meet student performance requirements stated in the Contract or for failure to meet generally accepted standards of fiscal management, the School must close permanently at the end of the then-current school year or on a date specified in the notification of termination and the School shall not enter into a contract with any other sponsor."
- b. Change the notification deadline in the third paragraph to "January 15."

The rest of Section 11.7 remains as originally written in the Contract.

### 23. Article XI, Section 11.10 shall be replaced in its entirety with the following:

"**Termination of the Contract.** The Sponsor may choose to terminate this Contract for (1) failure to meet student performance requirements stated in this Contract; (2) failure to meet generally accepted standards of fiscal management; (3) violation of any material provision of this Contract or applicable state or federal law; or (4) other good cause. Additionally, if the Sponsor has suspended the operation of this Contract under R.C. 3314.072, the Sponsor may choose to terminate this Contract prior to its expiration.

By January 15 of the year in which the Sponsor intends to terminate this Contract, the Sponsor shall notify the School of the proposed action in writing. The notice shall include the reasons for the proposed action in detail, the effective date of the termination, and a statement that the School may, within fourteen (14) days of receiving the notice, request in writing an informal hearing before the Sponsor. The informal hearing shall be held within fourteen (14) days of the receipt of a request for the hearing. Within fourteen (14) days after the informal hearing, the Sponsor shall issue a written decision either affirming or rescinding the decision to terminate this Contract.

The termination of this Contract shall be effective upon the occurrence of the later of the following events: (a) the date of the notice of termination; or (b) if an informal hearing is requested and the Sponsor affirms its decision to terminate this Contract, the effective date of the termination specified in the notice.

If the Contract is terminated for failure to meet student performance requirements stated in the contract or for failure to meet generally accepted standards of fiscal management, the School must close permanently at the end of the then-current school year or on a date specified in the notification of termination and the School shall not enter into a contract with any other sponsor."

- 24. Attachment 3.5.3 shall be replaced in its entirety with the attached.
- 25. Attachment 5.2 shall be added with the attached.
- 26. Attachment 6.13. Participation policies, including participation in blended learning program, if applicable, shall be added. The rest of Attachment 6.13 remains as originally attached to the Contract.
- 27. Attachment 9.2 shall be replaced in its entirety with the attached.
- 28. Attachment 11.6 shall be replaced in its entirety with the attached.

ALL OTHER SECTIONS, SUBSECTIONS, TERMS, OR PROVISIONS OF THE CONTRACT SHALL REMAIN IN FULL FORCE AND IN EFFECT UNLESS OTHERWISE SPECIFICALLY MODIFIED HEREIN.

**Educational Service Center of** Lake Erie West By-(Signature)

Its: Superintendent

Governing Authority of Pathway School of Discovery By: KWW A. Kobu

(Signature)

Its: President

with full authority to execute this Contract for and on behalf of **Sponsor** and with full authority to bind **Sponsor**.

Date: 10 - 19 - 16

with full authority to execute this Contract for and on behalf of **Governing Authority** and with full authority to bind **Governing Authority**.

Date: \_\_\_\_\_\_

#### Modification to Community School Contract with Educational Service Center of Lake Erie West

Be it resolved that the Pathway School of Discovery Academy Board of Directors does hereby ratify the approval of the Modification No. 4 to the Community School Contract with the Educational Service Center of Lake Erie West.

Board President:

Signature:

Kevin A. Robie Kevin a. Robie 9/21/2016

Approved on:

## ATTACHMENT 3.5.3 SUSPENSION AND CLOSING PROCEDURES

School Name:

Authorizer Name: Educational Service Center of Lake Erie West

Suspension or Closing:

To the extent that the school governing authority and/or the school's fiscal officer is unable or unwilling to execute its responsibilities in effecting an orderly suspension or closure of the school, the authorizer shall assume part or all of such tasks to the full extent possible.

| I. Initial Notifications, Student Records and School Records |   |  |  |  |
|--|---|--|--|--|
| <b>Completion Date</b>                                       | Action  |  |  |  |
|  | 1. Notify the Office of Quality School Choice that the school is suspending or closing; within 24 hours of the action, send in the community school's board resolution or official authorizer notice, including the date of closing. A school is suspended or closed for instruction when instruction is no longer taking place. A school may have a later official closing date if closing actions such as filing final expenditure reports or submitting EMIS data necessitate the school maintaining a status of Inactive in OEDS-R so that it can complete these or similar actions. If a school's closing occurs as a result of continued poor academic performance (ORC Section 3314.25 or 3314.351), often referred to as "closed by operation of law," there is no expectation that the authorizer will send in a notice, unless the school is closing prior to the end of the school year. In that instance, it is expected that the authorizer will notify the Office of Quality School Choice within 24 hours of the decision to close sooner than the end of the academic year. |  |  |  |
|  | 2. Provide a clear written timeline of the suspension/closing process.  |  |  |  |
|  | 3. Notify parents that the school is suspending/closing through a formal letter from the school Governing Authority and the authorizer superintendent or CEO within 24 hours of the action. The letter to include but not limited to: the reason for the suspension/closing of the school, options for enrolling in another community school, traditional schools or nonpublic schools and contact information.   |  |  |  |
|  | 4. Provide each parent with the location that their child's records are being delivered.  |  |  |  |
|  | 5. Provide authorizer contact information to all parents.   |  |  |  |
|  | 6. Provide each parent with the location that their child's records are being delivered.  |  |  |  |
|  | 7. Notify the teachers and staff that the school is closing.  |  |  |  |
|  | 8. Ensure that STRS and SERS contributions are current.   |  |  |  |
|  | 9. Clarify COBRA benefits and when medical benefits end. In the case in which the staff and faculty are employees of a management company, ensure that all employees have a contact person at the management company who will assist them through the transition.   |  |  |  |
|  | 10. Remind the faculty of their obligation to teach up to the date of closing or otherwise determine that the school is properly staffed up to the day of closing.  |  |  |  |
|  | 11. Ensure that each faculty member's LPDC information is current and available to the teachers.  |  |  |  |
|  | 12. Provide authorizer contact information to all staff.  |  |  |  |
|  | 13. Notify the Ohio State Teachers Retirement System and School Employees Retirement System.  |  |  |  |
|  | 14. Notify the Area Coordinator's office to schedule the student enrollment/FTE review. Preferably, the FTE review should be completed within 7 business days of the closing, if possible, or within 7 business days of the Area Coordinator's notification of the school's closing by the department.  |  |  |  |
|  | 15. Take control of and secure all school records, property and assets immediately when the school suspends or closes, to the extent possible. In the case in which the building's landlord seizes the facility and its contents, or when a governing authority reneges on its obligations or other unforeseen circumstances, legal action may be necessary. The authorizer should share any such circumstances with the Office of Quality School Choice as quickly as possible.  |  |  |  |
|  | Note: The DAS site for records retention schedules can be found here:<br>http://apps.das.ohio.gov/rims/General/General.asp. The federal records retention schedules can be found<br>here: http://www2.ed.gov/policy/gen/leg/fra.html.   |  |  |  |

| 16. Put student records in order and produce transcript materials immediately. The expectation is that records would be organized by grade level and district of residence, with the student's name and SSID clearly displayed. |
|---|
| 17. If the school has graduated students, compile a list of the names and dates of the graduates and retain that list.  |
| 18. Have copies of each list ready for the resident district receiving the records and the authorizer.  |
| 19. Have all available IEP, enrollment, and attendance records available for the completion of the FTE closure review.  |
| 20. Prepare a list of all students (name and SSID), and the resident district to which the student's records are to be delivered.   |
| 21. Have copies of each list ready for the resident district receiving the records and the authorizer.  |
| 22. Make copies of all students' records and retain the copies for use by the Auditor of State.   |
| 23. Deliver the original student records to each student's district of residence within seven business days of the school's closure. [ORC Section 3314.44]  |
| 24. Upon delivery, provide a list of the records given to the district representative, and obtain the name and signature of the individual receiving the records and the date.  |
| Note: Special education records shall be provided directly to each receiving school or school district's special education administrators for all students with disabilities.   |
| 25. Provide the authorizer with an updated list indicating delivery information.  |
| 26. Notify the Auditor of State of the school's closing/suspension and schedule a final audit.  |

Please note: The numbering does not indicate a specific sequence of steps; it is intended for ease of reference.

### **II. Disposition of Assets**

If the governing authority does not retain a treasurer to oversee the remaining financial activity, (note that this Contract requires retention of the School Treasurer for oversight of all closure and post-closure activity and responsibility) the authorizer may act as or appoint a receiver to oversee the closing of the financial records in the absence of a fiscal officer.

Keep assets separated by source of funding, state or federal, for purposes of disposition.

Federal dollars cannot be used to pay state liabilities. Account for all school property throughout the closing process by distinguishing state from federal dollars.

For Federal Title and other consolidated and competitive funds, follow EDGAR liquidation procedures in 34 CFR 80.32 including disposition for items valued at \$5,000 or greater.

For assets purchased with Federal funds that have a value of less than \$5,000 and all assets purchased with state funds, follow the school's disposition plan and track the disposition for each asset.

**Note:** Technology equipment received as part of the Erate program is federal property and cannot be sold as part of the school's assets. Specific guidance on donating the equipment can be found here: http://www.usac.org/sl/applicants/before-youre-done/equipment-transfer.aspx.

**Note:** If a community school closes and ceases to operate as a community school and the school has received computer hardware or software from the former Ohio SchoolNet commission or the former eTech Ohio commission, such hardware or software shall be turned over to the department of education, which shall redistribute the hardware and software, to the extent such redistribution is possible, to school districts in conformance with the provisions of the programs as they were operated and administered by the former eTech Ohio commission.

| <b>Completion Date</b> | Action for assets purchased with either state or federal funds   |  |  |
|------------------------|--|--|--|
| Closure only           | 27. Review the financial records of the school.  |  |  |
| Closure only           | 28. Establish check off list of assets including all inventory with proper USAS codes, state codes, and the price of each item and identify the source of funds; in the case of donated items, follow the accounting guidance. |  |  |

| Closure only    | <ul> <li>29. Establish the fair market (initial and amortized) value following generally accepted business rules and in a transparent manner. The Uniform Commercial Code offers guidelines for liquidating assets in a commercially reasonable manner for all state purchased assets and federally purchased assets that have a value of less than \$5,000 (ORC 1309.627).</li> <li>Note: Essentially, the price should be at the current price in any recognized market at the time of disposition or otherwise consistent with reasonable practices among dealers in the type of property subject to disposition. The school's governing authority's capital assets policy should also be followed.</li> <li>If an asset has \$0 market value and the school is planning to dispose of the asset at a public auction, the school should still place a minimal value on the item.</li> </ul> |  |  |  |
|-----------------|--|--|--|--|
|                 | 30. Identify staff who will have legal authority for payment processes (e.g. checks, cash, credit cards, etc.).  |  |  |  |
| Closure only    | 31. Establish disposition plan for any remaining items.  |  |  |  |
| Closure only    | 32. Notify Office of Quality School Choice, then public media (print media, radio) of the date, time, and location of any property disposition auction.  |  |  |  |
| Closure only    | 33. Provide board resolutions and minutes of any assets transferred at no cost to another school.  |  |  |  |
| Closure only    | 34. Identify any State Facilities Commission guarantees, if any.   |  |  |  |
| Closure only    | 35. Prepare documentation for disposition of the school's fixed assets.  |  |  |  |
| Completion Date | In Addition, for Assets purchased with Federal funds<br>36. For all federal program purchases that have a value of \$5,000 or greater, follow steps 26 through 30  |  |  |  |
| Closure only    | and 32 and EDGAR liquidation procedures in 34 CFR 80.32.<br>Note: Unless otherwise <i>described</i> below for the Public Charter School Program and the National School Lunch Program, all items purchased with federal funds may be sold at auction. However, for any item with a value of \$5,000 or higher, the item must be labelled on the disposition of assets record as having been purchased with federal dollars along with the purchaser information (name of school or organization and contact).  |  |  |  |
| Closure only    | <ul> <li>37. <u>Public Charter School Program, if applicable</u>.</li> <li>a. PCSP assets must first be offered to other community schools with requisite board resolutions consistent with the purpose of the PCSP.</li> <li>b. If there are no takers, then an auction sale must be held to dispose of the assets along with the state funded assets.</li> <li>c. After the above steps have been taken, any remaining assets may be offered to any public school district with documented board resolutions by the community school and the accepting district.</li> <li>d. Provide OCS with a written report of the property, and if available, a bill of sale.</li> </ul>   |  |  |  |
| Closure only    | 38. <u>National School Lunch Program</u> . Cafeteria equipment purchased with funds from the National School Lunch Program can only be liquidated through written guidance issued by the Office for Child Nutrition (OCN). Contact OCN prior to proceeding with any liquidation of Equipment.  |  |  |  |
| Completion Date | In Addition, for Assets purchased with State funds   |  |  |  |
| Closure only    | 39. In addition to establishing a fair market value, track sale of items and have supporting board resolutions for donation of items to another community or other public school or non-profit entity.   |  |  |  |
| Closure only    | 40. Consistent with ORC Section 3314.051, offer <b>real property</b> acquired from a public school district to that school district's board first at fair market value. If the district board does not accept the offer within 60 days, dispose of the property in another lawful manner.  |  |  |  |

### **III. Payments to Creditors**

For suspended schools, ensure that employees' salaries, STRS/ERS, and IRS obligations are current. For closing schools, the authorizer should consult with its legal counsel prior to implementing this section to determine if there are any outstanding federal or state claims that may need to be paid first including but not limited to federal and state taxes, Medicare, Workers Compensation, and city wage taxes. If applicable, provide documentation of any Workers Compensation claims.

Note: The authorizer is obligated to retain the school's financial records and should follow the relevant state and federal records retention schedules. The DAS site for records retention schedules can be found here: http://apps.das.ohio.gov/rims/General/General.asp. The federal records retention schedules can be found here: http://www2.ed.gov/policy/gen/leg/fra.html.

With two exceptions, noted below, utilize only state dollars, auction proceeds, foundation dollars, and any other non-federal dollars to pay creditors in the following order:

| <b>Completion Date</b> | Action   |  |  |  |  |
|------------------------|--|--|--|--|--|
|                        | 41. Retirement funds of employees of the school, such as 401(k)'s.   |  |  |  |  |
|                        | 42. STRS/SERS/retirement systems and other adjustments – see ORC 3314.074.   |  |  |  |  |
|                        | NOTE: Teachers, administrators, and other staff working for a conversion community school through a contract with the sponsoring district should be treated as employees under ORC Section 3314.074.   |  |  |  |  |
|                        | 43. Teachers and staff   |  |  |  |  |
|                        | Exception: Federal dollars can be used to pay the teacher costs and STRS/SERS/retirement costs for any employees who were paid from federal funds when the school was open.  |  |  |  |  |
|                        | 44. Private creditors or those entities that have secured a judgment against the school, including a preparation and audit costs (prepared financials).  |  |  |  |  |
|                        | Exception: Audit preparation costs can be paid from federal funds under certain conditions. Contact the  |  |  |  |  |
|                        | Office of Grants Management for specific guidance before applying any costs against federal funds to support audit costs. Please refer to Auditor of State Bulletin 2009-011.  |  |  |  |  |
|                        | 45. Any remaining funds shall be paid to the department of education for redistribution to the school districts in which the students who were enrolled in the school at the time it ceased operation were entitled to attend school under ORC section 3313.64 or 3313.65. The amount distributed to each school district shall be proportional to the district's share of the total enrollment in the community school. |  |  |  |  |
|                        | If the assets of the school are insufficient to pay all persons or entities to which compensation is owed,   |  |  |  |  |
|                        | the prioritization of the distribution of the assets to individual persons or entities within each class of payees may be determined by decree of a court in accordance section 3314.074 and Chapter 1702 of the Ohio Revised Code.  |  |  |  |  |

| IV. Prep        | IV. Preparation of Itemized Financials and Documents for FTE Review                                      |  |  |  |  |  |  |
|-----------------|--|--|--|--|--|--|--|
| Completion Date | Action   |  |  |  |  |  |  |
|                 | Review and prepare the following itemized financials:  |  |  |  |  |  |  |
|                 | 46. Year-end financial statements, notes to the financial statements and if applicable schedule of feder |  |  |  |  |  |  |
|                 | awards.  |  |  |  |  |  |  |
|                 | 47. A cash analysis (taking the previous month's recap and reconciliation of bank accounts to books)     |  |  |  |  |  |  |
|                 | for determination of the cash balance as of the closing date.  |  |  |  |  |  |  |
|                 | 48. List of investments in paper hard copy format.   |  |  |  |  |  |  |
|                 | 49. List of all payables and indicate when a check to pay the liability clears the bank.                 |  |  |  |  |  |  |
|                 | 50. List of all unused checks (collect and void all unused checks).                                      |  |  |  |  |  |  |
|                 | 51. List of any petty cash.  |  |  |  |  |  |  |
|                 | 52. List of bank accounts, closing the accounts once all transactions are cleared.                       |  |  |  |  |  |  |
|                 | 53. List of all payroll reports including taxes, retirement or adjustments on employee contract.         |  |  |  |  |  |  |
|                 | 54. List of all accounts receivable.   |  |  |  |  |  |  |
|                 | 55. List of assets and their disposition.  |  |  |  |  |  |  |
|                 | 56. FTE review complete.   |  |  |  |  |  |  |

| V. Final Payments and Adjustments |  |  |  |  |
|-----------------------------------|--|--|--|--|
| <b>Completion Date</b>            | Action   |  |  |  |
|                                   | The authorizer shall continually monitor the condition of the closed school and be prepared to receive or transmit funds on behalf of the school as directed by an appropriate agency. Receipt of funds can happen more than a year after a school's closure, and the authorizer is obliged to serve as the recipient of such funds and adjustments. |  |  |  |
|                                   | If the assets of the school are insufficient to pay all persons or entities to which compensation is owed, the prioritization of the distribution of the assets to individual persons or entities within each class of payees may be determined by decree of a court in accordance with Section 3313.074 and Chapter 1702 of the ORC.                |  |  |  |
|                                   | 57. Receive any funds or adjustments credited to the account of the closed school.   |  |  |  |
|                                   | 58. Determine if any portion of any funds or adjustments can be applied to satisfy any remaining debts; payables (any money owed to another).  |  |  |  |
|                                   | 59. Send all or the remaining portion to the Ohio Department of Education, Office of Budget and School Funding, for final disposition.   |  |  |  |

The authorizer and school governing authority representative identified below certify that all steps listed above were completed, unless otherwise noted.

| School:                            |       |  |
|------------------------------------|-------|--|
| Governing Authority Representative | Date: |  |
| Authorizer:                        |       |  |
| Authorizer Representative          | Date: |  |

The School's governing authority and the authorizer execute the Suspension or Closing Assurance Form and maintain it with copies sent to the Office of Quality School Choice. Submit the *Suspension & Closing Assurance Form* by uploading it into your named authorizer folder in the document exchange, accessed through your SAFE account, Collaboration Center.

## ATTACHMENT 5.2 FACILITIES PLAN

- 1. Detailed description of each facility used for instructional purposes, containing the address and grades served;
- 2. Annual costs associated with leasing each facility, paid for by or on behalf of the school, if applicable;
- 3. Annual mortgage principal and interest payments that are paid by the school, if applicable;
- 4. Name of the lender or landlord, identified as such, and the lender's or landlord's relationship to the operator, if any; and
- 5. If the school leases property from the operator, addendum to lease with verification from independent professional in the real estate field that, at the time the lease was agreed to, the lease was commercially reasonable.

## Attachment 5.2 Facilities Plan

Pathway School of Discovery is located at 173 Avondale Dr, Dayton, OH 45404 and serves grades K-8. The facility is 42,808 net square foot. The building has 28 classrooms, an art room, music room, media center, and gymnasium. The building is single story with a vinyl sided façade with a split face wainscot and sloped shingle roof. The building is leased from Charter Development Company, which is a subsidiary of NHA, for \$750,650 per year.

## ATTACHMENT 6.13 (PARTIAL) ATTENDANCE POLICIES

1. Participation Policies, including any policy or procedures for non-classroom learning opportunities and/or blending learning program

**NOTE:** The School's attendance and participation records shall be made available, upon request, to the Department of Education, Auditor of State, and Sponsor, to the extent permitted by 20 U.S.C. 1232g, the Family Educational Rights and Privacy Act (FERPA), section 3319.321, and any applicable rule or regulations thereto.

#### Attachment 6.13 Monitoring Attendance for Non-Classroom Based Learning

We take attendance before and after each non-classroom based learning opportunity; such as, after school tutoring, field trips, and summer learning programs.

## ATTACHMENT 9.2 FISCAL SERVICES AGREEMENT

- 1. Fiscal Services Agreement, or
- 2. Current resolution waiving the requirement for the Board to employ or contract with the fiscal officer

**NOTE:** A resolution waiving the requirement for the Board to employ or contract with the fiscal officer must be approved by the Sponsor and submitted to the Ohio Department of Education. The resolution is only valid for one year and does not waive the requirement that the school have a designated fiscal officer.

Any changes made as to a fiscal officer or a fiscal services agreement must be reported in writing to the Sponsor within five (5) business days.

#### Resolution of the Board of Directors of Pathway School of Discovery ("The School")

Motion made by: KATTY ROAPICK and seconded by: DIANE MARTIN

WHEREAS, Ohio Revised Code Section 3314.011 requires that every community school established under R.C. Chapter 3314 have a designated fiscal officer "employed by or engaged under a contract with the governing authority of the community school"; and

WHEREAS, the Governing Authority of <u>Pathway School of Discovery</u> (the "School") has designated and engaged a fiscal officer directly under a contract between the Governing Authority of the School and its management company; and

WHEREAS, the Governing Authority believes it has complied with R.C. 3314.011, and

WHEREAS, the Governing Authority may adopt a resolution waiving the requirement that it is the party responsible to employ or contract with the designated fiscal officer so long as the community school's sponsor also approves of the resolution; and

WHEREAS, the Governing Authority, for clarity purposes wishes to waive any right to have a separate contract with its fiscal officer other than through its current management agreement; and

WHEREAS, the School's sponsor may request that the Governing Authority take additional steps beyond enacting this resolution; and

WHEREAS, the Governing Authority and its management company agree that the designated fiscal officer is a licensed and bonded public officer of the School; and

WHEREAS, regardless of whether the contract designating the fiscal officer of the School contains other services or unrelated services, as a public officer of the School, the primary fiduciary duty as a fiscal officer is to the Governing Authority of the School.

**NOW THEREFORE, BE IT RESOLVED**, that  $A_{NNE} S_{HERMAN}$  is authorized and approved as the designated fiscal officer of the School and is engaged under a contract with the Governing Authority pursuant to the terms of the School's management agreement; and

**RESOLVED**, that effective February 1, 2016, the Governing Authority resolves to waive the requirement that the Governing Authority be the party responsible to separately employ its licensed fiscal officer or to contract with its designated fiscal officer in a contract separate from a contract that includes other services. The Governing Authority, does not in any way waive the fact that the fiscal officer is a public officer of the School with all of the responsibilities to the School inherent in that role; and

**RESOLVED**, that the Chair of the school's Board of Directors, or, the Principal of the School, have the authority, upon the sponsor's request, to submit an application any other information necessary to effect waiver of the requirement that the Governing Authority must employ or contract with the designated fiscal officer separately from its current management contract.

| Discussion: |  |
|-------------|--|
|             |  |

Vote: \_\_\_\_\_Yes \_\_\_\_No \_\_\_\_\_Abstaining

[Signatures on the Following Page]

Signature

Rearick Benetr Signature

ignature

Signature

Signature

#### Signature

Signature

Approved by School's Sponsor:

Ву:\_\_\_\_\_

lts: \_\_\_\_ \_\_\_\_\_

And

| Ву: |  |
|-----|--|
|     |  |
|     |  |

lts:

(Sponsor Name)

## ATTACHMENT 11.6 PERFORMANCE ACCOUNTABILITY FRAMEWORK

The community school sponsor is responsible to provide monitoring, oversight, and technical assistance to the community schools that it sponsors. This attachment describes a comprehensive performance accountability and compliance monitoring system, and it provides the information that will be used in making rigorous and standards-based renewal, revocation, and intervention decisions. At the core of these decisions, the Educational Service Center of Lake Erie West (ESCLEW) considers the well-being and interests of the students served by the community school.

This attachment includes (a) the framework by which ESCLEW will assess the performance of the school, (b) the interventions and technical assistance that ESCLEW may utilize, and (c) the academic and non-academic performance goals of the school.

### **A. Performance Assessment**

The performance assessment details many items that the Sponsor will review in its oversight of the school as part of its normal oversight or through a high stakes review. The Sponsor will conduct a high stakes review of the school upon renewal or no later than five years after the effective date of this Contract, whichever is sooner, and at any other point that the Sponsor deems appropriate, at its discretion. If an issue arises or is discovered, the Sponsor will utilize targeted intervention(s) listed in this attachment under section B, including any technical assistance necessary to assist the school.

#### **Overview – Areas of Assessment**

- Mission and Vision Statements
- Governance
- Fiscal or Financial Matters
- Operations, Recordkeeping, Reporting, and Data
- Special Education and Other Diverse Learning Needs
- Academics

#### Mission and Vision Statements

- Clear and specific mission statement
- Relevant and compelling vision statement
- Evidence of progress in meeting mission and vision
- Evidence of mission-driven decision-making in policies, curriculum, and programming

#### Governance

- Effective governance and management structure
- Oversight of school management and evaluation of management team
- Maintenance of corporate status: good standing certificate, timely filing of statement of continued existence, adherence to Code of Regulations, maintenance of corporate record-keeping, proper corporate signatures
- Maintenance of Federal tax-exempt status: timely filing of Form 990s, IRS conflict of interest adherence, no evidence of violation of Internal Revenue Code Section 501(c)(3)
- Compliance with contractual terms: governance trainings, updated governing authority roster, conflict of interest policies
- Disclosure of conflicts, signing of conflict of interest policy, and abstentions in minutes
- Compliance with Open Meetings law, including sufficient minutes
- Balance of educational and financial reporting at meetings
- Informed decision-making and operational transparency
- Any other area of operations, compliance or performance of the Contract

#### **Fiscal or Financial Matters**

- Timely submission of financial documentation to Sponsor, including:
  - o Balance sheet statement of financial position
  - Income statement statement of activities
  - Monthly budget to actual, including all revenue sources
  - Monthly separated grants reporting
  - Monthly enrollment reports
- Red flag financial concerns: significant changes in enrollment, significant changes in policy environment, loss of a particular revenue stream, facility changes, significant turnover in leadership
- Timely payment on all loans/debts
- Financial audit, including timely follow-up on findings and/or material non-compliance
- Solid business plan and clear evidence of capacity to carry out plan
- Sound management and competence in understanding of proper use of public funds
- Grounded in professional standards for sound financial operations and sustainability
- Consistency of financial reporting at governing authority meetings
- Other finance related compliance reports

#### **Operations, Record-Keeping, Reporting, and Data**

- Fair treatment, non-selective, and non-discriminatory access in admissions
- Fair and consistent treatment in discipline, with a focus on positive behavioral intervention supports
- Timely and accurate reporting to state data systems
- Appropriate attendance record-keeping and addressing excessive flagging
- Timely response to sponsor requests on interventions and routine compliance follow-ups
- Compliance with public records law, including trained public records officer
- Annual report with clear and accurate performance data

#### **Special Education and Other Diverse Learning Needs**

- Appropriate services for all students
- Appropriate access, education, support services and outcomes for students with special education and other diverse learning needs, including English language learners, migrant, and homeless students
- Compliance with policies and procedures for identification, evaluation, and referral of students with disabilities under IDEA and Section 504
- Compliance with all state and federal timelines and reporting requirements
- Proper due process notifications
- Appropriate discipline administration and discipline training for staff, including restraint, seclusion, and positive behavioral intervention supports

If the school has a majority of students with special needs, the School is expected to provide a learning environment that accomplishes the goals of these students in a manner superior to others. The Sponsor expects to see evidence of success for the special education population that exceeds progress in other environments, with methods that are innovative and effective in order to produce such success.

#### Academics

The School should be prepared to assess every measure listed below, and every component of every measure by data, in order to enhance targeted assistance. Performance standards must include, but are not limited to, all applicable report card measures set forth in R.C. 3302.03 or R.C. 3314.017. The School understands that, every year, additional measures and components may be added and graded and that an overall grade will be calculated starting in Fiscal Year 2018.

• Achievement measures "absolute achievement." Achievement includes the Performance Index. This assesses the achievement of every student (not just which students are proficient) and points are received for each level. The Achievement measure also includes Performance Indicators. The indicators show how many students have a proficient level of knowledge. Schools need to have 80% of their students reach

proficient in order to meet an indicator. Anything below a letter grade of a C, or 70%, will require targeted assistance.

- **Progress** represents the average annual level of improvement of each student in math and reading. Its components consist of Value Added Overall, Gifted Students, Students in the Lowest 20% in Statewide Achievement, and Students with Disabilities. Anything below a -1 value added gain index will require targeted assistance.
- **Gap Closing** measures the narrowing of gaps in reading, math, and graduation rates among socioeconomic, racial, ethnic, or disability subgroups. It is also called Annual Measurable Objectives (AMOs). Anything below a 70% will require targeted assistance.
- **Graduation Rate** measures the percentage of students who entered in 9th grade and graduated 4 and 5 years later. Currently, Graduation Rate has a 4 year component, for which anything below an 84% will require targeted assistance, and a five year component, for which anything below an 85% will require targeted assistance.
- **K-3 Literacy** measures the improvement in reading for grades K through 3. Currently, K-3 Literacy Improvement has one component, but may have more components in the future. This component measures the percentage of students between kindergarten and 3rd grade who went from not on-track to on-track in reading in one year. Anything below 43.2% will require targeted assistance.
- **Prepared for Success** measures whether graduates are prepared for college or a career. Prepared for Success currently has one component, which measures preparedness by looking at College Admission Tests, Dual Enrollment, Industry Credentials, Honors Diplomas, and Advanced Placement and International Baccalaureate participation and scores.
- Other Measures of Academic Success, including:
  - Other assessments conducted by school (NWEA, etc.)
  - Ohio Improvement Process
  - Assessments and interventions required by Third Grade Reading Guarantee
  - Incorporation of Common Core analytic understanding
  - Mission-specific performance measures of academic success
  - Comparison of student performance data with data of the home district building to which students would otherwise be assigned (suggested, and required upon request)
  - Post-secondary readiness, if applicable

## **B.** Technical Assistance, Interventions, and Statutory Consequences

The technical assistance and interventions listed below provide guidance and notice for the types of intervention that the Sponsor may take if the School fails to perform satisfactorily in the areas reviewed as through the performance assessment framework or if the School does not meet its contract goals. The Sponsor will make attempts to both streamline and customize interventions based on the School's operational structure and mission. The interventions may occur as part of a Corrective Action Plan or independently.

Many of the interventions listed below require the Sponsor to provide technical assistance. The Sponsor has attempted to limit financial cost to the School for other interventions in which the School must pay for the intervention. The School should budget for a certain amount of expense due to required performance interventions, programs to collect data, and staff to assess and analyze performance data, as well as increased staff for tutoring or cure of deficiencies. All interventions will require follow-up, proof of cure, and adherence to cure. Typical interventions, remedies, or cures imposed by Sponsor may be, but are not limited to the following:

#### **Mission and Vision Statements Interventions**

- Sponsor technical assistance in reaching the mission of the School and to assess the ability of the School to realize its mission, which may include aspects of culture, parent satisfaction, or other visions stated by the School in its Contract, curriculum, or programming
- Sponsor assessment of leadership with the Board or the administration and mutual plan to correct weaknesses in meeting mission

- Outside coaching or training, research, observations, or remedies to meet stated goals, at School's cost.
- Leadership training in area of desired goals and why they are not being reached or how to reach them more effectively, at School's cost
- Other tailored interventions based on the specific circumstances

#### **Governance Interventions**

- Sponsor technical assistance
- Targeted technical assistance in procedure, minutes, motions, notices, or similar governance matters
- Workout session held by Sponsor
- Recommendations of Sponsor to be followed by Board as to dysfunctions and options for resolving them
- Review and required changes of Code of Regulation, governance policies, or other rules or practices affecting governance
- Clear guidance written into policy and adherence to guidance, as to roles and responsibilities
- Required replacement of absenteeism or members hampering business or quorum
- Required agreement of Board to dismiss board member or officer violating policy, law, procedure, rules, the Code, or other customary good governance precepts
- Training using trainer suggested by or acceptable to Sponsor, at School's cost
- Other tailored interventions based on the specific circumstances

#### **Fiscal or Financial Matters Interventions**

- Sponsor technical assistance
- Enhanced and more accurate financial reports required, such as additional financial documentation, debtto- asset ratios, cash flow analyses, requirements of debt to service coverage ratios, enrollment variances and withdrawal frequencies, requirements of minimum unrestricted cash, and increased transparency enhancements for analytical purposes
- Compliance plans for recurring or material findings
- Contingency plans addressing potential funding shortfalls or other disruptive events
- Finance committee or finance expert on the board
- Pre-audit exercise with qualified Treasurer consultant, at the School's cost.
- Independent training, at the School's cost
- Other tailored interventions based on the specific circumstances

#### **Special Education and Other Diverse Learning Needs Interventions**

- Sponsor technical assistance, including but not limited to Sponsor audit of files
- Legal review of practices and procedures targeted to issues discovered, at School's cost
- State audit of files and State training, at School's cost (if any cost required)
- Training, at the School's cost
- Independent Audit, at the School's cost
- Other tailored interventions based on the specific circumstances

#### **Academic Interventions**

- Corrective Action Plan
- Sponsor technical assistance
- Targeted intervention by the Sponsor as defined below
  - Face to face meetings and increased reporting to Sponsor on progress
  - Checklists, plans to cure, and periodic data reporting
  - Revisions to the Ohio Improvement Plan
  - Required academic or professional assistance
  - Parent Engagement to support students in academics, attendance, and curriculum
  - Curriculum Mapping
  - o Study of Ohio's New Learning Standards and or the Next Generation of Assessments in Ohio
  - o OTES and OPES evaluations of staff

- Required training on the Ohio report card indicators or components
- Targeted coaching, at the School's cost
- o Targeted tutoring before or after school and/or weekends, at the School's cost

#### **Corrective Action Plans and Targeted Intervention**

Designation on the Ohio Local Report Card of a D or F in any measure, or any relative weakness in a component or measure, will require targeted intervention. The first step of that intervention will be the School's submission of all data which make up the measure, or components of that measure, and which components have caused the grade of a D or F. The School must list each component with all proof of accurate data to support it, and the School must report quarterly on academic data for measures and components identified as weaknesses.

ESCLEW may require the School to develop a Corrective Action Plan (CAP) if the School fails to meet contract goals under section C of this Attachment, has historically low achievement in any area or measure, or if there is sudden negative changes in any area or measure. ESCLEW will work with the School in developing the CAP and will utilize other appropriate entities or resources, as it deems necessary.

All academic interventions outlined in the CAP will require a thorough understanding of weaknesses, based on data derived from assessment tools selected in conjunction with the Sponsor, a thorough understanding of in which measure and which component within a measure the weakness occurs, and a targeted plan to cure that particular weakness or those weaknesses.

Once the CAP is in place, it will be monitored monthly by the Sponsor to assess the School's progress toward meeting the goals. Should the School fail to make adequate or appropriate progress, the plan may be re-evaluated and changed, with either additional interventions or supplements to the existing interventions.

#### **Renewal and Non-Renewal**

A non-renewal may occur in accordance with any cause allowed by Chapter 3314 of the Revised Code (R.C.) or the Contract. In making renewal decisions, the Sponsor will consider the school's academic performance and progress, as well as the non-academic areas listed within the Performance Accountability Framework, in conjunction with the information provided with the school's renewal application. Failure to meet contract goals or failure to appropriately address required interventions or corrective action may result in non-renewal or in a shortened contract term.

Currently, schools in which a majority of the enrolled students are children with disabilities receiving special education and related services are exempt from statutory closure on the basis of academic performance. However, the Sponsor may non-renew or progress toward termination for a special education status school if the school fails to meet its goals.

#### **Comments on Statutory Consequences**

Probation, which may lead to suspension, may be used for any cause allowed by the Ohio Revised Code, including but not limited to:

- Untimely submissions or lack of cooperation in submissions required by the State or the Sponsor for two or more times per year, or for over five business days
- Lack of compliance with laws or the Contract
- Failure to achieve academic goals, as listed in the Performance Accountability Framework
- Any fiscal matter of any level of risk
- Governance, administrative, or minor financial issues that can be adequately corrected by a Plan to Cure acceptable to Sponsor
- Potential or actual ethics or conflicts violations

Suspension, which may lead to termination, is more likely to be used for matters such as, but not limited to:

- Health or safety matters
- Consistent failure to achieve academic goals, despite corrective action plans and technical assistance

- Money owed to the State or its agencies, vendors, or the Sponsor that is delinquent, overextended, or may or does render insolvency
- Inability to make payroll
- Undisclosed debt or obligations
- Lack of accuracy in or misrepresentation, cheating, or fraud in any reporting figures, data, or testing results that is not corrected within timeframes allowed by law or which create financial, reputational, governance or academic risk
- Other matters of noncompliance, the magnitude or risk of which may be assessed by Sponsor as good cause or allowed by the Code
- Uncorrected or actual ethics or conflict violations
- Inability to meet, or lack of progress in meeting, the terms of the Contract, the laws, rules, regulations or requirements, or the reasonable requests of Sponsor geared to correct deficiencies as cited in a probation notice or corresponding corrective action plan

Termination of the Contract will occur if the School fails to develop a corrective action plan or fails to comply with or remedy the problems cited in the corrective action plan developed with the Sponsor as part of a suspension.

Permanent closure under R.C. 3314.35 or 3314.351 will occur under the following circumstances:

- For schools with grades no higher than 3, for two of the past three years, the school has either been in academic emergency, received an F in improving literacy in grades K-3, or received an overall F.
- For schools with grades 4-8, but no higher than 9, for two of the past three years, the school has either been in academic emergency and not met value-added growth, received an F for the performance index score and an F for value-added progress, or received an overall F and an F for value-added progress.
- For schools with grades 10-12 that are not drop-out recovery schools, for two of the past three years, the school has either been in academic emergency, received an F for the performance index score and not met AMOs, or has received an overall F and an F for value-added progress
- For drop-out recovery and prevention program schools, for two of the past three years and beginning on July 1, 2014, the school received a designation of "does not meet standards."

## **C. School-Specific Performance Goals and Measures**

This section of Attachment 11.6 provides school-specific academic and non-academic goals and performance measures. Goals must be specific, targeted, measurable, and rigorous. The goals must include a level or rate of increase, and they must include subgroup targets. The school recognizes that these goals may be revised if the make-up of the school significantly changes or if the standards by which the school is judged by the state change. Schools should include goals for each year of the contract or up to three years. If a school has a contract length of longer than three years, a goal evaluation will be performed after three years, and the school acknowledges that it will re-assess goals at that time, if not sconer.

ESCLEW will review the School's progress toward performance goals during site visits, on a periodic basis throughout the school year. From time to time, the ESCLEW Leadership Team may request evidence or other documentation showing progress or efforts toward attaining contract goals. If progress is not evident or if it is insufficient, ESCLEW may, at its discretion, require additional targeted assistance or intervention. If the goal is no longer applicable or realistic, ESCLEW may request that the School submit revised goals by which the School will be judged in renewal or termination decisions. The School agrees to submit such revised goals no later than three weeks after ESCLEW's request.

| School's<br>Mission<br>Statement | Working in partnership with parents and the community, the Pathway School of Discovery mission is to become one of the finest K-8 schools in the country. We offer a challenging, character-based education through a rigorous curriculum with high academic and social expectations. Our school prioritizes the academic and instructional time so each student reads, computes and writes at or above grade level. We expect our student to master basic skills and realize their full academic potential in preparation for higher education and adulthood. |               |              |  |                             |                             |
|----------------------------------|--|---------------|--------------|--|-----------------------------|-----------------------------|
|                                  | Measure  | 2014-15 Data  | 2015-16 Data | 2016-17 Goal   | 2017-18 Goal                | 2018-19 Goal                |
| Mission-<br>Specific Goal        | To achieve an overall letter grade "B"<br>or higher on the state report card.  | Not available |              | Overall Letter<br>Grade "D" in all<br>applicable<br>components | Overall Letter<br>Grade "C" | Overall Letter<br>Grade "B" |

## **Academic Performance Goals**

| Performance<br>Area | Measure   | 2014-15 Data  | 2015-16<br>Data | 2016-17 Goal   | 2017-18 Goal   | 2018-19 Goal   |
|---------------------|---|---|-----------------|--|--|--|
| Achievement         |   | 25% (4 out of<br>16)  |                 | 31.3% (5 out of 16)  | 37.5% (6 out of 16)  | 43.8% (7 out of 16)  |
|                     | Percentage and total number of<br>proficiency level indicators met  | (roughly 35% of schools in  |                 | To maintain or<br>increase the   | To maintain or<br>increase the   | To maintain or<br>increase the   |
|                     |   | the state<br>received a<br>letter grade F)  |                 | percentage of<br>indicators met  | percentage of<br>indicators met  | percentage of<br>indicators met  |
|                     | Performance index score and percent<br>increase in each category (advanced,<br>accelerated, proficient, etc.) | 82.4 out of 120<br>Advanced 7.0<br>Accelerated<br>22.7<br>Proficient 29.7<br>Basic 24.8<br>Limited 15.0<br>Untested 0.9 |                 | 83.0 out of 120<br>Advanced 7.2<br>Accelerated 22.6<br>Proficient 29.4<br>Basic 26.0<br>Limited 14.9<br>Untested 0.0 | 83.4 out of 120<br>Advanced 7.2<br>Accelerated 22.9<br>Proficient 29.6<br>Basic 26.2<br>Limited 14.2<br>Untested 0.0 | 83.7 out of 120<br>Advanced 7.3<br>Accelerated 23.1<br>Proficient 29.8<br>Basic 26.0<br>Limited 14.0<br>Untested 0.0 |
| Progress            | Value-added progress score,<br>including subgroups  | Overall: -5.98<br>Lowest 20%:<br>-1.35<br>Students with<br>Disabilities:<br>-1.68                                       |                 | Overall: -5.50<br>Lowest 20%:<br>-1.25<br>Students with<br>Disabilities:<br>-1.50                                    | Overall: -5.00<br>Lowest 20%:<br>-1.10<br>Students with<br>Disabilities:<br>-1.00                                    | Overall: -4.50<br>Lowest 20%:<br>-1.00<br>Students with<br>Disabilities:<br>-0.75                                    |

| Performance<br>Area         | Measure  | 2014-15 Data   | 2015-16<br>Data | 2016-17 Goal  | 2017-18 Goal   | 2018-19 Goal   |  |
|-----------------------------|--|--|-----------------|---|--|--|--|
| Gap Closing                 | Annual measurable objectives<br>(AMOs) percentages, including<br>subgroup and subject        | Reading<br>Multiracial 77.1<br>White 66.3<br>All students<br>65.0<br>Econ Disadv<br>59.7<br>African<br>American 56.5<br>Students/Disab<br>32.1<br><u>Math</u><br>Multiracial 68.6<br>White 61.1<br>All students<br>58.3<br>Econ Disadv<br>52.8<br>African<br>American 47.2<br>Students/Disab<br>30.2 |                 | Reading<br>Multiracial 78.6<br>White 67.6<br>All students 66.3<br>Econ Disadv 60.9<br>African American<br>57.6<br>Students/Disab 32.7<br><u>Math</u><br>Multiracial 70.0<br>White 62.3<br>All students 59.5<br>Econ Disadv 53.9<br>African American<br>48.1<br>Students/Disab<br>30.8 | Reading<br>Multiracial 80.2<br>White 69.0<br>All students 67.6<br>Econ Disadv 62.1<br>African American<br>58.8<br>Students/Disab<br>33.4<br><u>Math</u><br>Multiracial 71.4<br>White 63.6<br>All students 60.7<br>Econ Disadv 54.9<br>African American<br>49.1<br>Students/Disab<br>31.4 | Reading<br>Multiracial 81.8<br>White 70.4<br>All students 69.0<br>Econ Disadv 63.4<br>African American<br>60.0<br>Students/Disab<br>34.1<br><u>Math</u><br>Multiracial 72.8<br>White 64.8<br>All students 61.9<br>Econ Disadv 56.0<br>African American<br>50.1<br>Students/Disab<br>32.0 |  |
| Graduation<br>Rate          | Four- and five-year graduation rates   | N/A  | N/A             | N/A   | N/A  | N/A  |  |
| K-3 Literacy<br>Improvement | Percentage of students moving from<br>not on track to on track                               | 53.8%  |                 | 54.9%   | 56.0%  | 57.1%  |  |
| Prepared for<br>Success     | Component grade and percentage of<br>students participating across all six<br>measured areas | N/A  | N/A             | N/A   | N/A  | N/A  |  |

| Performance<br>Area    | Measure   | 2014-15 Data  | 2015-16<br>Data | 2016-17 Goal  | 2017-18 Goal   | 2018-19 Goal   |
|------------------------|---|---|-----------------|---|--|--|
| Student<br>Performance | Early Literacy:<br>Beginning in kindergarten, Academy<br>students will take a nationally-normed<br>standardized assessment. Each<br>year, students will score at or above<br>the 65th percentile in reading and<br>math as measured by beginning-of-<br>year to end of-year learning growth in<br>grades K-1. | Math:<br>K=99th<br>1=97rd<br>Reading:<br>K=97th<br>1=74th |                 | Beginning in<br>kindergarten,<br>Academy students<br>will take a nationally-<br>normed<br>standardized<br>assessment. Each<br>year, students will<br>score at or above<br>the 65th percentile<br>in reading and math<br>as measured by<br>beginning-of-year to<br>end of-year learning<br>growth in grades K-<br>1. | Beginning in<br>kindergarten,<br>Academy students<br>will take a<br>nationally-normed<br>standardized<br>assessment. Each<br>year, students will<br>score at or above<br>the 65th percentile<br>in reading and<br>math as measured<br>by beginning-of-<br>year to end of-year<br>learning growth in<br>grades K-1. | Beginning in<br>kindergarten,<br>Academy students<br>will take a<br>nationally-normed<br>standardized<br>assessment. Each<br>year, students will<br>score at or above<br>the 65th percentile<br>in reading and<br>math as measured<br>by beginning-of-<br>year to end of-year<br>learning growth in<br>grades K-1. |

## Non-Academic Performance Goals

| Performance<br>Area      | Measure   | 2014-15 Data                                | 2015-16 Data | 2016-17 Goal  | 2017-18 Goal  | 2018-19 Goal  |  |
|--------------------------|---|---|--------------|---|---|---|--|
| Attendance               | Percentage as measured on the local report card | Attendance<br>Rate: 94.8%                   | N/A          | Each year, the<br>school will<br>average a<br>student<br>attendance rate<br>at or above<br>93%. | Each year, the<br>school will<br>average a<br>student<br>attendance rate<br>at or above<br>93%. | Each year, the<br>school will<br>average a<br>student<br>attendance rate<br>at or above<br>93%. |  |
| Parent<br>Satisfaction   | Parent satisfaction survey                      | Overall Parent<br>Satisfaction<br>Rate: 92% | N/A          | Each year, the<br>school will have<br>an overall parent<br>satisfaction rate<br>of 80%.         |   | Each year, the<br>school will have<br>an overall parent<br>satisfaction rate<br>of 80%.         |  |
| Financial<br>Performance | Audit   | TBD   | N/A          | Each year, the<br>school will<br>receive an<br>unmodified<br>audit.                             | Each year, the<br>school will<br>receive an<br>unmodified<br>audit.                             | Each year, the<br>school will<br>receive an<br>unmodified<br>audit.                             |  |

| Performance<br>Area               | Measure                  | 2014-15 Data | 2015-16 Data | 2016-17 Goal  | 2017-18 Goal  | 2018-19 Goal  |  |
|-----------------------------------|--------------------------|--------------|--------------|---|---|---|--|
| Financial<br>Sustainability       | School budget            | TBD          | N/A          | Each year,<br>Pathway will end<br>the school year<br>with a balanced<br>budget.       | Each year,<br>Pathway will end<br>the school year<br>with a balanced<br>budget.       | Each year,<br>Pathway will end<br>the school year<br>with a balanced<br>budget.       |  |
| Governing<br>Board<br>Performance | Professional Development | N/A          | N/A          | Each Board<br>member will<br>complete one<br>Professional<br>Development<br>per year. | Each Board<br>member will<br>complete one<br>Professional<br>Development<br>per year. | Each Board<br>member will<br>complete one<br>Professional<br>Development<br>per year. |  |

### MODIFICATION NO. 3 TO OHIO COMMUNITY SCHOOL CONTRACT BY and BETWEEN Educational Service Center of Lake Erie West ("Sponsor" or "ESCLEW") AND Pathway School of Discovery ("Governing Authority" OR "School")

**WHEREAS**, the ESCLEW and the Governing Authority entered into an amended and restated Community School Contract ("Contract") effective on July 1, 2012; and

WHEREAS, the ESCLEW and the Governing Authority agree to the following modifications;

NOW THEREFORE, the parties modify the Contract as follows:

1. Article I, Section 1.7. Add the following sub-paragraph:

"Review the financial and enrollment records of the School at least once per month with the governing authority or fiscal officer and provide a written report regarding the review within ten days after the review."

The rest of Section 1.7 remains as originally written in the Contract.

- 2. Article II, Section 2.1. At the beginning, add the following sentence: "All Governing Authority members must be pre-approved by Sponsor." The rest of Section 2.1 remains as originally written in the Contract.
- 3. Article II, Section 2.4. At the end, add the following paragraphs:

"The School and Sponsor agree and state that, pursuant to 20 U.S.C. 1232g, the Family Educational Rights and Privacy Act (FERPA) and 34 C.F.R. 99, the Sponsor is an authorized representative of a state educational authority and that the School is permitted to disclose to the Sponsor personally identifiable information from an education record of a student without parental consent (or student consent, where applicable) and that the Sponsor is authorized by federal, state, and local law to conduct audits, compliance evaluations, and enforcement activities of federal and state supported education programs. Accordingly, the School agrees to grant to Sponsor's employees full and complete access as defined hereinafter to "education records," as defined by FERPA, and all documents, records, reports, databases, and other information made available to or maintained by the School or its agent(s) (including educational management companies) that are reportable to the Ohio Department of Education or its agencies or to the Ohio Auditor of State. Such information shall include, but is not limited to, the School Options Enrollment System (SOES) and the Education Management Information System (EMIS). "Full and complete access" shall include the ability to inspect and copy paper and electronic documents at the School. The School or its agent(s) shall provide usernames and passwords where applicable to enable Sponsor to have remote self-service access, in read-only format.

The Sponsor agrees to comply with FERPA and the regulations promulgated thereunder. The Sponsor warrants that it uses reasonable methods to limit Sponsor employee access to only those records in which it has legitimate educational interests and that, as required by law, the Sponsor will destroy the educational records when no longer needed for the purposes outlined in this Contract or otherwise needed under state or federal law or any applicable court order."

The rest of Section 2.4 remains as originally written in the Contract.

- 4. Article III, Section 3.4. Replace the first sentence with the following: "If the School applies for Priority 2 E-Rate funding or if it is otherwise required by law, the School must have a state-approved technology plan and complete any necessary filing procedures on or before the applicable deadlines." The rest of Section 3.4 remains as originally written in the Contract.
- 5. Article IV, Section 4.1. In the first paragraph, add sections 3301.948, 3313.6020, 3313.7112, and 3313.89, and remove chapters 3307 and 3309. In the second paragraph, add sections 2921.43 and 2921.44. The rest of Section 4.1 remains as originally written in the Contract.
- 6. Article VI, Section 6.4. Add the following sentence: "The School must submit a calendar of assessments to Sponsor prior to the start of each academic year." The rest of Section 6.4 remains as originally written in the Contract.
- 7. Article VI, Section 6.13. Add the following sentence: "The School must keep attendance records and requirements for non-classroom-based learning opportunities." The rest of Section 6.13 remains as originally written in the Contract.
- 8. Article VII, Section 7.2. Add the following subparagraph: "the number and percentage of students on IEPs, the number of those students on home instruction per the IEP, and any associated children's residential center (CRC)." The rest of Section 7.2 remains as originally written in the Contract.
- 9. Article VIII, Section 8.1. Replace the final sentence with the following: "The School shall maintain a roster and meeting dates for the Local Professional Development Committee (LPDC) and Teacher-Based Teams (TBTs), and the laws and rules governing LPDC and TBTs must be implemented by the School." The rest of Section 8.1 remains as originally written in the Contract.
- 10. Article VIII, Section 8.3. At the end, add the following sentence: "To the extent required or allowed by state or federal law, the School shall comply with chapters 3307 and 3309 of the Ohio Revised Code." The rest of Section 8.3 remains as originally written in the Contract.
- 11. Article VIII, Section 8.4. Replace the second and third sentences with the following: "The Governing Authority hereby appoints the Sponsor as a representative pursuant to R.C. 3319.39(D), for purposes of receiving and reviewing the results of criminal background checks performed under R.C. 3319.39(A)(1) for employees working at the School and authorizes its agents to communicate this information directly to the Sponsor." The rest of Section 8.4 remains as originally written in the Contract.
- 12. Article IX, Section 9.2. Replace the first sentence with the following sentences: "The School agrees that its fiscal officer shall be its Treasurer, Anne Sherman, contracted through the School's management company, National Heritage Academies. Any changes made to the fiscal officer or fiscal services agreement must be reported to the Sponsor within five (5) business days." The rest of Section 9.2 remains as originally written in the Contract.

13. Article X, Section 10.2. Add the following sub-paragraph:

"A failure of the Governing Authority and/or School or any of its officers, directors, employees, agents, or contractors to report data or information, or the reporting of improper, inaccurate, erroneous, fraudulent, negligent, or incomplete information or data to ESCLEW."

The rest of Section 10.2 remains as originally written in the Contract.

14. Attachment 9.3 shall be replaced in its entirety with the attached.

15. Attachment 9.4 shall be replaced in its entirety with the attached.

16. Attachment 11.6 shall be replaced in its entirety with the attached.

ALL OTHER SECTIONS, SUBSECTIONS, TERMS, OR PROVISIONS OF THE CONTRACT SHALL REMAIN IN FULL FORCE AND IN EFFECT UNLESS OTHERWISE SPECIFICALLY MODIFIED HEREIN.

**Educational Service Center of** Lake Erie West By: (Signature)

Governing Authority of Pathway School of Discovery

in a Robie By: (Signature)

Its: Superintendent

Its: President

with full authority to execute this Contract for and on behalf of **Sponsor** and with full authority to bind **Sponsor**.

Date:

with full authority to execute this Contract for and on behalf of **Governing Authority** and with full authority to bind **Governing Authority**.

Date: 06/17/15

## ATTACHMENT 9.3 FISCAL LICENSURE

## 1. Treasurer's License

**NOTE:** Any updates or changes to the fiscal licensure must be sent to the Sponsor within five (5) business days.

# STATE OF OHIO DEPARTMENT OF EDUCATION

5 Year School Treasurer School Treasurer License

| A | Anne Loui    | se Sherm     | an | OH320      | 2471     | 05/08/20   | )15 | 07/01/2014 | to 06/30/  | 2019 |
|---|--------------|--------------|----|------------|----------|------------|-----|------------|------------|------|
|   | THIS LICENSI | E AWARDED TO |    | EDUCATOR S | STATE ID | ISSUE DATE |     | ÉFFEC      | TIVE DATES |      |

The holder of this credential, having satisfactorily completed the requirements prescribed by The State Board of Education and the laws of Ohio, is authorized to teach the subject(s) or serve in the area(s) listed on this document for the period specified. The holder of this credential is responsible for being knowledgeable about current requirements for maintaining the credential.

Superintendent of Public Instruction

This official document was created by the Ohio Department of Education and represents a true copy of a legal educator license as referenced in Ohio Revised Code Section 3319.36.

Credential # 21125983

Employers may verify the validity of this credential by going to Educator Profile on the Ohio Department of Education's website at <u>education.ohio.gov</u> and ensuring that the unique credential number appearing on this credential matches the person's records in Educator Profile, which is the official record of educator credential history.

## ATTACHMENT 9.4 FISCAL BOND

## 1. Treasurer's Bond

**NOTE:** Any updates or changes to the bond shall be sent to the Sponsor within five (5) business days.



Crime Program Certificate of Coverage

# NATIONAL HERITAGE ACADEMIES

Certificate No. 007

Master Bond No. 106291888

This is to Certify

That the following Fiscal Agent of the Pathway School of Discovery, 173 Avondale Dr., Dayton, OH 45404-2123 is covered under a Crime Program through the master policy of the National Heritage Academies provided by the Travelers Casualty & Surety Company of America. This Bond is issued pursuant to OCR 3314.011.

The coverage and limit(s) are set forth below:

Insured: Anne L. Sherman

Terms of Coverage:

Policy Term: May 1, 2015 to August 1, 2016

Coverage:

A-Scheduled

Limit of Liability \$25,000. Deductible \$500.

Policy Term Premium: \$122.60

Travelers Casualty & Surety Company of America

Matthew VanConant, Authorized Representative
# ATTACHMENT 11.6 PERFORMANCE ACCOUNTABILITY FRAMEWORK

The community school sponsor is responsible to provide oversight and technical assistance to the community schools that it sponsors. This attachment describes a comprehensive performance accountability and compliance monitoring system, and it provides the information that will be used in making rigorous and standards-based renewal, revocation, and intervention decisions. At the core of these decisions, the Educational Service Center of Lake Erie West (ESCLEW) considers the well-being and interests of the students served by the community school.

This attachment includes (a) the framework by which ESCLEW will assess the performance of the school, (b) the interventions and technical assistance that ESCLEW may utilize, and (c) the academic and non-academic performance goals of the school.

## A. Performance Assessment

The performance assessment details many items that the Sponsor will review in its oversight of the school as part of its normal oversight or through a high stakes review. The Sponsor will conduct a high stakes review of the school upon renewal or every five years, whichever is sooner, and at any other point that the Sponsor deems appropriate, at its discretion. If an issue arises or is discovered, the Sponsor will utilize targeted intervention(s) listed in this attachment under section B, including any technical assistance necessary to assist the school.

### **Overview – Areas of Assessment**

- Mission and Vision Statements
- Governance
- Fiscal or Financial Matters
- Operations, Recordkeeping, Reporting, and Data
- Special Education and Other Diverse Learning Needs
- Academics

#### **Mission and Vision Statements**

- Clear and specific mission statement
- Relevant and compelling vision statement
- Evidence of progress in meeting mission and vision
- Evidence of mission-driven decision-making in policies, curriculum, and programming

#### Governance

- Effective governance and management structure
- Oversight of school management and evaluation of management team
- Maintenance of corporate status: good standing certificate, timely filing of statement of continued existence, adherence to Code of Regulations, maintenance of corporate record-keeping, proper corporate signatures
- Maintenance of Federal tax-exempt status: timely filing of Form 990s, IRS conflict of interest adherence, no evidence of violation of Internal Revenue Code Section 501(c)(3)
- Compliance with contractual terms: governance trainings, updated governing authority roster, conflict of interest policies
- Disclosure of conflicts, signing of conflict of interest policy, and abstentions in minutes
- Compliance with Open Meetings law, including sufficient minutes
- Balance of educational and financial reporting at meetings
- Informed decision-making and operational transparency
- Any other area of operations, compliance or performance of the Contract

#### **Fiscal or Financial Matters**

- Timely submission of financial documentation to Sponsor, including:
  - Balance sheet statement of financial position
  - Income statement statement of activities
  - Monthly budget to actual, including all revenue sources
  - o Monthly separated grants reporting
  - o Monthly enrollment reports
- Red flag financial concerns: significant changes in enrollment, significant changes in policy environment, loss of a particular revenue stream, facility changes, significant turnover in leadership
- Timely payment on all loans/debts
- Financial audit, including timely follow-up on findings and/or material non-compliance
- Solid business plan and clear evidence of capacity to carry out plan
- Sound management and competence in understanding of proper use of public funds
- Grounded in professional standards for sound financial operations and sustainability
- Consistency of financial reporting at governing authority meetings
- Other finance related compliance reports

#### **Operations, Record-Keeping, Reporting, and Data**

- Fair treatment, non-selective, and non-discriminatory access in admissions
- Fair and consistent treatment in discipline, with a focus on positive behavioral intervention supports
- Timely and accurate reporting to state data systems
- Appropriate attendance record-keeping and addressing excessive flagging
- Timely response to sponsor requests on interventions and routine compliance follow-ups
- Compliance with public records law, including trained public records officer
- Annual report with clear and accurate performance data

#### **Special Education and Other Diverse Learning Needs**

- Appropriate services for all students
- Appropriate access, education, support services and outcomes for students with special education and other diverse learning needs, including English language learners, migrant, and homeless students
- Compliance with policies and procedures for identification, evaluation, and referral of students with disabilities under IDEA and Section 504
- Compliance with all state and federal timelines and reporting requirements
- Proper due process notifications
- Appropriate discipline administration and discipline training for staff, including restraint, seclusion, and positive behavioral intervention supports

If the school has a majority of students with special needs, the School is expected to provide a learning environment that accomplishes the goals of these students in a manner superior to others. The Sponsor expects to see evidence of success for the special education population that exceeds progress in other environments, with methods that are innovative and effective in order to produce such success.

#### Academics

The School should be prepared to assess every measure listed below, and every component of every measure by data, in order to enhance targeted assistance. The School understands that, every year, additional measures and components will be added and graded and that an overall grade will be calculated starting in Fiscal Year 2015.

- **Graduation Rate** measures the percentage of students who entered in 9<sup>th</sup> grade and graduated 4 and 5 years later. Currently, Graduation Rate has a 4 year component (anything below an 84% will require targeted assistance) and a five year component (anything below an 85% will require targeted assistance).
- **K-3 Literacy** measures the improvement in reading for grades K-3. Currently, K-3 Literacy Improvement has one component, but may have more components in the future.

- **Prepared for Success** measures whether graduates are prepared for college or a career. Prepared for Success currently has one component, and it consists of College Admission Tests, Dual Enrollment, Industry Credentials, Honors Diplomas, and Advanced Placement and International Baccalaureate components.
- Achievement measures "absolute achievement." Achievement consists of the Performance Index (anything below a 70% will require targeted assistance). This assesses the achievement of every student (not just which students are proficient) and points are received for each level. The Achievement measure also consists of the Performance Indicators. The indicators show how many students have a proficient level of knowledge. Schools need to have 80% of their students reach proficient in order to meet an indicator. The indicators combine to show a letter grade and that letter grade must be a C, or 70% or more, to avoid targeted assistance.
- **Progress** represents the average annual level of improvement of each student. Its components consist of Value Added Overall, Gifted Students, Progress within the Lowest 20%, and Students with Disabilities (all must be greater than or equal to a -1 value added gain index in order to avoid targeted assistance).
- **Gap Closing** measures the narrowing of gaps in reading and math among subgroups of socio-economic, racial, ethnic, or disability. It is also called Annual Measurable Objectives (AMOs) (anything below a 70% will require targeted assistance).
- Other Measures of Academic Success, including:
  - Other assessments conducted by school (NWEA, etc.)
  - Ohio Improvement Process
  - o Assessments and interventions required by Third Grade Reading Guarantee
  - o Incorporation of Common Core analytic understanding
  - Mission-specific performance measures of academic success
  - Comparison of student performance data with data of the home district building to which students would otherwise be assigned (suggested, and required upon request)
  - Post-secondary readiness, if applicable

# **B.** Technical Assistance, Interventions, and Statutory Consequences

The technical assistance and interventions listed below provide guidance and notice for the types of intervention that the Sponsor may take if the School fails to perform satisfactorily in the areas reviewed as through the performance assessment framework or if the School does not meet its contract goals. The Sponsor will make attempts to both streamline and customize interventions based on the School's operational structure and mission. The interventions may occur as part of a Corrective Action Plan or independently.

Many of the interventions listed below require the Sponsor to provide technical assistance. The Sponsor has attempted to limit financial cost to the School for other interventions in which the School must pay for the intervention. The School should budget for a certain amount of expense due to required performance interventions, programs to collect data, and staff to assess and analyze performance data, as well as increased staff for tutoring or cure of deficiencies. All interventions will require follow-up, proof of cure, and adherence to cure. Typical interventions, remedies, or cures imposed by Sponsor may be, but are not limited to the following:

### **Mission and Vision Statements Interventions**

- Sponsor technical assistance in reaching the mission of the School and to assess the ability of the School to realize its mission, which may include aspects of culture, parent satisfaction, or other visions stated by the School in its Contract, curriculum, or programming
- Sponsor assessment of leadership with the Board or the administration and mutual plan to correct weaknesses in meeting mission
- Outside coaching or training, research, observations, or remedies to meet stated goals, at School's cost.
- Leadership training in area of desired goals and why they are not being reached or how to reach them more effectively, at School's cost
- Other tailored interventions based on the specific circumstances

### **Governance Interventions**

- Sponsor technical assistance
- Targeted technical assistance in procedure, minutes, motions, notices, or similar governance matters
- Workout session held by Sponsor
- Recommendations of Sponsor to be followed by Board as to dysfunctions and options for resolving them
- Review and required changes of Code of Regulation, governance policies, or other rules or practices affecting governance
- Clear guidance written into policy and adherence to guidance, as to roles and responsibilities
- Required replacement of absenteeism or members hampering business or quorum
- Required agreement of Board to dismiss board member or officer violating policy, law, procedure, rules, the Code, or other customary good governance precepts
- Training using trainer suggested by or acceptable to Sponsor, at School's cost
- Other tailored interventions based on the specific circumstances

### **Fiscal or Financial Matters Interventions**

- Sponsor technical assistance
- Enhanced and more accurate financial reports required, such as additional financial documentation, debtto- asset ratios, cash flow analyses, requirements of debt to service coverage ratios, enrollment variances and withdrawal frequencies, requirements of minimum unrestricted cash, and increased transparency enhancements for analytical purposes
- Compliance plans for recurring or material findings
- Contingency plans addressing potential funding shortfalls or other disruptive events
- Finance committee or finance expert on the board
- Pre-audit exercise with qualified Treasurer consultant, at the School's cost.
- Independent training, at the School's cost
- Other tailored interventions based on the specific circumstances

### Special Education and Other Diverse Learning Needs Interventions

- Sponsor technical assistance, including but not limited to Sponsor audit of files
- Legal review of practices and procedures targeted to issues discovered, at School's cost
- State audit of files and State training, at School's cost (if any cost required)
- Training, at the School's cost
- Independent Audit, at the School's cost
- Other tailored interventions based on the specific circumstances

### **Academic Interventions**

- Corrective Action Plan
- Sponsor technical assistance
- Targeted assistance by the Sponsor as defined below
- Face to face meetings and increased reporting to Sponsor on progress
- Checklists, plans to cure, and periodic data reporting
- Revisions to the Ohio Improvement Plan
- Required academic or professional assistance
- Parent Engagement to support students in academics, attendance, and curriculum
- Curriculum Mapping
- Study of Ohio's New Learning Standards and or the Next Generation of Assessments in Ohio
- OTES and OPES evaluations of staff
- Required training on the Ohio report card indicators or components
- Targeted coaching, at the School's cost
- Targeted tutoring before or after school and/or weekends, at the School's cost

### **Corrective Action Plans and Targeted Intervention**

Designation on the Ohio Local Report Card of a D or F in any measure, or any relative weakness in a component or measure, will require targeted intervention. The first step of that intervention will be the School's submission of all data which make up the measure, or components of that measure, and which components have caused the grade of a D or F. The School must list each component with all proof of accurate data to support it, and the School must report quarterly on academic data for measures and components identified as weaknesses.

ESCLEW may require the School to develop a Corrective Action Plan (CAP) if the School fails to meet contract goals under section C of this Attachment, has historically low achievement in any area or measure, or if there is sudden negative changes in any area or measure. ESCLEW will work with the School in developing the CAP and will utilize other appropriate entities or resources, as it deems necessary.

All academic interventions outlined in the CAP will require a thorough understanding of weaknesses, based on data derived from assessment tools selected in conjunction with the Sponsor, a thorough understanding of in which measure and which component within a measure the weakness occurs, and a targeted plan to cure that particular weakness or those weaknesses.

Once the CAP is in place, it will be monitored monthly by the Sponsor to assess the School's progress toward meeting the goals. Should the School fail to make adequate or appropriate progress, the plan may be re-evaluated and changed, with either additional interventions or supplements to the existing interventions.

#### **Renewal and Non-Renewal**

A non-renewal may occur in accordance with any cause allowed by Chapter 3314 of the Ohio Revised Code (ORC) or the Contract. In making renewal decisions, the Sponsor will consider the school's academic performance and progress, as well as the non-academic areas listed within the Performance Accountability Framework, in conjunction with the information provided with the school's renewal application. Failure to meet contract goals or failure to appropriately address required interventions or corrective action may result in non-renewal or in a shortened contract term.

Currently, schools in which a majority of the enrolled students are children with disabilities receiving special education and related services are exempt from statutory closure on the basis of academic performance. However, the Sponsor may non-renew or progress toward termination for a special education status school if the school fails to meet its academic goals.

#### **Comments on Statutory Consequences**

Probation, which may lead to suspension, may be used for any cause allowed by the ORC, including but not limited to:

- Untimely submissions or lack of cooperation in submissions required by the State or the Sponsor for two or more times per year, or for over five business days
- Lack of compliance with laws or the Contract
- Failure to achieve academic goals, as listed in the Performance Accountability Framework
- Any fiscal matter of any level of risk
- Governance, administrative, or minor financial issues that can be adequately corrected by a Plan to Cure acceptable to Sponsor
- Potential or actual ethics or conflicts violations

Suspension, which may lead to termination, is more likely to be used for matters such as, but not limited to:

- Health or safety matters
- Consistent failure to achieve academic goals, despite corrective action plans and technical assistance
- Money owed to the State or its agencies, vendors, or the Sponsor that is delinquent, overextended, or may or does render insolvency

- Inability to make payroll
- Undisclosed debt or obligations
- Lack of accuracy in or misrepresentation, cheating, or fraud in any reporting figures, data, or testing results that is not corrected within timeframes allowed by law or which create financial, reputational, governance or academic risk
- Other matters of noncompliance, the magnitude or risk of which may be assessed by Sponsor as good cause or allowed by the Code
- Uncorrected or actual ethics or conflict violations
- Inability to meet, or lack of progress in meeting, the terms of the Contract, the laws, rules, regulations or requirements, or the reasonable requests of Sponsor geared to correct deficiencies as cited in a probation notice or corresponding corrective action plan

Termination of the Contract will occur if the School fails to develop a corrective action plan or fails to comply with or remedy the problems cited in the corrective action plan developed with the Sponsor as part of a suspension.

Permanent closure under ORC 3314.35 or 3314.351 will occur under the following circumstances:

- For schools with grades no higher than 3, for two of the past three years, the school has either been in academic emergency, received an F in improving literacy in grades K-3, or received an overall F.
- For schools with grades 4-8, but no higher than 9, for two of the past three years, the school has either been in academic emergency and not met value-added growth, received an F for the performance index score and an F for value-added progress, or received an overall F and an F for value-added progress.
- For schools with grades 10-12 that are not drop-out recovery schools, for two of the past three years, the school has either been in academic emergency, received an F for the performance index score and not met AMOs, or has received an overall F and an F for value-added progress
- For drop-out recovery and prevention program schools, for two of the past three years and beginning on July 1, 2014, the school received a designation of "does not meet standards."

# C. School-Specific Performance Goals and Measures

This section of Attachment 11.6 provides school-specific academic and non-academic goals and performance measures. Goals must be specific, targeted, measurable, and rigorous. The goals must include a level or rate of increase, and they must include subgroup targets. Goals should be aligned with any relevant goals stated in the OIP, School Improvement process, or with existing SMART goals. The School's Comparison goal must compare the school's student performance to the state, schools serving similar populations, and/or schools in the same geographical area. The school may also include a mission-specific academic goal.

The school recognizes that these goals may be revised if the make-up of the school significantly changes or if the standards by which the school is judged by the state change. Schools should include goals for each year of the contract or up to five years. If a school has a contract length of longer than five years, a high-stakes review and goal evaluation will be performed after five years, and the school recognizes that it will re-assess goals at that time, if not sooner.

# Attachment 11.6 - Pathway Academic Performance Goals

| Performance<br>Area | Metrics  | Goal(s)  | Year 1            | Year 2            | Year 3            | Year 4            | Year 5            | Year 6            | Year 7            | Year 8            |
|---------------------|--|--|-------------------|-------------------|-------------------|-------------------|-------------------|-------------------|-------------------|-------------------|
| Achievement         | Proficiency Level<br>Indicators: Percentage of<br>students scoring at or<br>above proficient, by<br>grade and subject; Local<br>Report Card grade,<br>percent, and number of<br>indicators met | The school will<br>increase the number of<br>students scoring<br>proficient on the Ohio<br>Achievement<br>Assessments (OAA)<br>annually.                                 | +1.0%             | +1.0%             | +1.0%             | +1.0%             | +1.0%             | +1.0%             | +1.0%             | +1.0%             |
|                     | Performance Index:<br>Local Report Card<br>grade, score, and/or<br>percent increase from<br>previous year  | The school will<br>maintain or exceed a<br>performance index<br>score of 84 or make<br>annual gains.   | =/><br>84         |
| Progress            | Local Report Card grade<br>and value-added<br>progress scores, by<br>subgroup, grade, and<br>subject   | The All Students<br>subgroup at Pinnacle<br>will have a minimum<br>gain index score<br>greater than or equal to<br>-1.   | =/><br>-1         |
| Gap Closing         | Local Report Card<br>Annual Measurable<br>Objectives grade and<br>percentages, by<br>subgroup and subject  | The school will reduce<br>the students with<br>disabilities<br>achievement gap by<br>decreasing the number<br>of students failing to<br>achieve proficiency<br>annually. | Decrease<br>by 5% |
| Graduation<br>Rate  | Local Report Card grade<br>and percentages for<br>four- and five-year<br>graduation rates  | Not applicable.  | -                 | -                 | -                 | -                 | -                 | -                 | -                 | -                 |

| Performance<br>Area         | Metrics  | Goal(s)  | Year 1                         | Year 2                         | Year 3                         | Year 4                         | Year 5                         | Year 6                         | Year 7                         | Year 8                         |
|-----------------------------|--|--|--------------------------------|--------------------------------|--------------------------------|--------------------------------|--------------------------------|--------------------------------|--------------------------------|--------------------------------|
| K-3 Literacy<br>Improvement | Local Report Card<br>grade, percentage of<br>students achieving or<br>exceeding targeted<br>gains, 3 <sup>rd</sup> grade reading<br>guarantee diagnostics<br>and assessments | The school will meet or<br>exceed the statewide<br>average for K-3<br>literacy growth.   | =/><br>statewide<br>average    |
| Prepared for<br>Success     | Local Report Card<br>component grade,<br>percentage of students<br>scoring at each<br>performance level  | Not applicable.  | -                              | -                              | -                              | -                              | -                              | -                              | -                              | -                              |
| Student<br>Performance      | Other valid and reliable assessments   | Each year, the school<br>will administer a<br>nationally-normed<br>standardized<br>assessment (for<br>example, NWEA) in<br>grades K-8. The school<br>will meet or<br>outperform the national<br>average of public<br>school students in<br>beginning-to-end-of-<br>year learning growth in<br>the majority of grades<br>and subjects in reading<br>and math in grades 2-8. | 50 <sup>th</sup><br>percentile |

## Non-Academic Performance Goals

| Performance<br>Area              | Measure(s)  | Goal(s)   | Year 1     | Year 2     | Year 3     | Year 4     | Year 5     | Year 6     | Year 7     | Year 8     |
|----------------------------------|---|---|------------|------------|------------|------------|------------|------------|------------|------------|
| Attendance                       | Attendance levels<br>as measured by<br>the Local Report<br>Card, school's<br>attendance<br>system | Each year, the<br>school will<br>average a student<br>attendance rate at<br>or above 93%. | =/><br>93% |
| Post-<br>Secondary<br>Enrollment | -   | Not applicable.   | -          | -          | -          | -          | -          | -          | -          | -          |

| Performance<br>Area   | Measure(s)   | Goal(s)   | Year 1                         | Year 2                         | Year 3                         | Year 4                         | Year 5                         | Year 6                         | Year 7                         | Year 8                         |
|---|--|---|--------------------------------|--------------------------------|--------------------------------|--------------------------------|--------------------------------|--------------------------------|--------------------------------|--------------------------------|
| Area<br>Mission<br>Statement<br>Pathway's<br>Mission: Working<br>in partnership<br>with parents and<br>the community,<br>the Pathway<br>School of<br>Discovery<br>mission is to<br>become one of<br>the finest K-8<br>schools in the<br>country. We offer<br>a challenging,<br>character-based<br>education through<br>a rigorous<br>curriculum with<br>high academic<br>and social<br>expectations. Our<br>school prioritizes<br>the academic and<br>instructional time<br>so each student<br>reads, computes<br>and writes at or<br>above grade level.<br>We expect our<br>student to master<br>basic skills and<br>realize their full<br>academic<br>potential in<br>preparation for<br>higher education<br>and adulthood. | Academic growth<br>as measured by a<br>nationally-<br>normed<br>standardized<br>assessment in<br>grades K-8. | To ensure the school<br>is providing life-long<br>opportunities to<br>students through its<br>educational program,<br>we propose the<br>following academic<br>growth goal:<br>Each year, the school<br>will administer a<br>nationally-normed<br>standardized<br>assessment (for<br>example, NWEA) in<br>grades K-8. The<br>school will meet or<br>outperform the<br>national average of<br>public school students<br>in beginning-to-end-<br>of-year learning<br>growth in the majority<br>of grades and subjects<br>in reading and math in<br>grades 2-8. | 50 <sup>th</sup><br>percentile |

| Performance<br>Area                               | Measure(s)     | Goal(s)   | Year 1  | Year 2  | Year 3  | Year 4  | Year 5  | Year 6  | Year 7  | Year 8  |
|---|----------------|---|---|---|---|---|---|---|---|---|
| Parent<br>Satisfaction                            | Parent survey. | Each year, the<br>school will have<br>an overall parent<br>satisfaction rate of<br>80% with at least<br>50% of parents<br>responding to a<br>parent satisfaction<br>survey.   | =/><br>80%<br>(with at least<br>50% response<br>rate) |
| Financial<br>Performance<br>and<br>Sustainability | Audit.         | Each year, the<br>school will<br>receive an<br>unqualified audit.   | Unqualified<br>audit                                  |
| Governing<br>Board<br>Performance                 | Evaluation.    | Each year, the<br>school's Board of<br>Directors will<br>formally review<br>the performance<br>of, and provide<br>feedback to, its<br>education<br>management<br>partner. The<br>review will be<br>used to identify<br>the management<br>partner's<br>successes and<br>opportunities to<br>improve its future<br>performance. | Annual<br>evaluation                                  |

### MODIFICATION NO. 2 TO OHIO COMMUNITY SCHOOL CONTRACT BY and BETWEEN Educational Service Center of Lake Erie West ("Sponsor" or "ESCLEW") AND Pathway School of Discovery ("Governing Authority" OR "School")

**WHEREAS**, the ESCLEW and the Governing Authority entered into an amended and restated Community School Contract ("Contract") effective on July 1, 2012; and

WHEREAS, the ESCLEW and the Governing Authority agree to the following modifications;

**NOW THEREFORE**, the parties modify the Contract as follows:

1. Article I, Section 1.2. At the end, add the following paragraph:

"The School is a separate, independent, and autonomous legal entity, responsible for educational programming, staff, budgeting and finance, and scheduling, and is not related to, an agent of, or under the control of the Sponsor, notwithstanding anything required herein or under the law related to the Sponsor's duties of oversight or intervention."

The rest of Section 1.2 remains as originally written in the Contract.

2. Article IV, Section 4.1 is updated as follows:

In the first paragraph, add section 3319.46.

At the end, add the following sentence: "The School will adhere to the provisions agreed to in its Race to the Top Memorandum of Understanding (Phase II) with the Ohio Department of Education, including the requirement to adopt and implement comprehensive evaluation systems for principals consistent with the Ohio Principal Evaluation System (OPES) framework."

The rest of Section 4.1 remains as originally written in the Contract.

- 3. Article VI, Section 6.3. The second and third sentences are deleted. The rest of Section 6.3 remains as originally written in the Contract.
- 4. Article VI, Section 6.4 shall be replaced in its entirety as follows:

"Academic Proficiency and Achievement Assessments. The School must administer all required statewide proficiency or achievement assessments and any other performance standards or assessments required by law, by the Ohio Department of Education, or recommended by Sponsor. The School must ensure that all such standards, requirements, and assessments are timely and properly administered, met, and completed. Any change in assessment or method of measurement of progress must be reported in writing to the Sponsor, along with the justification for the change. The School specifically acknowledges the closure provisions of R.C. 3314.35. Academic goals, performance standards, measurement, and assessment are included in **Attachment 11.6**."

- 5. Article VI, Section 6.6. R.C. 3314.06(A) is added the list of applicable exceptions. The rest of Section 6.6 remains as originally written in the Contract.
- 6. Article VI, Section 6.8 is updated as follows:

The heading is replaced as "Assuring Student Growth."

The first sentence is deleted.

The rest of Section 6.8 remains as originally written in the Contract.

- 7. Article VI, Section 6.12. At the end of subsection (c), add the following sentence: "If the School admits out-of-state students, it shall adopt a policy and tuition schedule." The rest of Section 6.12 remains as originally written in the Contract.
- 8. Article VIII, Section 8.1. The following sentence is deleted: "Any classroom teacher hired on or after July 1, 2013 and employed to provide instruction in physical education must hold a valid license issued pursuant to R.C. 3319.22 for teaching physical education." The rest of Section 8.1 remains as originally written in the Contract.
- 9. Article IX, Section 9.1. At the end, add the following sentence: "The School shall comply with the standards for financial reporting adopted under R.C. 3301.07(B)(2)." The rest of Section 9.1 remains as originally written in the Contract.
- 10. Article IX, Section 9.5. At the end, add the following sentence: "Financial performance goals, standards, measurement, and assessment are included in Attachment 11.6." The rest of Section 9.5 remains as originally written in the Contract.
- 11. Article IX, Section 9.7. In the first sentence, "all state per pupil foundation funds" is replaced with "the total amount of payment for operating expenses." The rest of Section 9.7 remains as originally written in the Contract.
- 12. Article XI, Section 11.5. At the end, add the following sentence: "If the term of this Contract is longer than five years, the Sponsor will conduct a high-stakes review every five years." The rest of Section 11.5 remains as originally written in the Contract.
- 13. Article XI, Section 11.6 is replaced in its entirety as follows:

**"Performance Assessments, Standards, Goals, and Interventions. Attachment 11.6** sets forth in detail (a) the performance assessment framework, (b) Sponsor technical assistance, interventions, and statutory consequences, and (c) the measurable and attainable goals upon which the School shall be evaluated by Sponsor. These performance measurement criteria supplement, but do not replace, alter, or limit Sponsor's statutory rights and responsibilities, including but not limited to those of nonrenewal, probation, suspension, or termination of the Contract."

14. Article XI, Section 11.7 is updated as follows:

The heading is updated to read: "Renewal and Non-Renewal of this Contract."

At the beginning of subsection (a), add the following paragraph: "Renewal is conditioned upon the Sponsor's determination that the School has satisfactorily complied with the applicable laws and this Contract, and that the School's progress in meeting the academic, financial, and organizational goals stated in this Contract is satisfactory. The School shall submit an Application for Renewal for purposes of this determination."

The rest of Section 11.7 remains as originally written in the Contract.

15. Article XI, Section 11.9. At the end, add the following paragraph:

"If the Sponsor suspends the operation of the School pursuant R.C. 3314.03, this Contract shall become void if the Governing Authority fails to provide a proposal to remedy the conditions cited by the Sponsor as reasons for the suspension, to the satisfaction of the Sponsor, by September 30th of the school year immediately following the school year in which the operation of school was suspended."

16. Attachment 6.4 is removed.

17. Attachment 11.6 is added.

ALL OTHER SECTIONS, SUBSECTIONS, TERMS, OR PROVISIONS OF THE CONTRACT SHALL REMAIN IN FULL FORCE AND IN EFFECT UNLESS OTHERWISE SPECIFICALLY MODIFIED HEREIN.

**Educational Service Center of** Lake Erie West (Signature)

Its: Superintendent

with full authority to execute this Contract for and on behalf of Sponsor and with full authority to bind Sponsor.

Date: 8/20/14

(Signature)

Its: President

with full authority to execute this Contract for and on behalf of **Governing Authority** and with full authority to bind **Governing Authority**.

Date: 6/18/14

# ATTACHMENT 11.6 PERFORMANCE ASSESSMENTS, STANDARDS, GOALS, AND INTERVENTIONS

The community school sponsor is responsible to provide oversight and technical assistance to the community schools that it sponsors. This attachment describes a comprehensive performance accountability and compliance monitoring system, and it provides the information that will be used in making rigorous and standards-based renewal, revocation, and intervention decisions. At the core of these decisions, the Educational Service Center of Lake Erie West (ESCLEW) considers the well-being and interests of the students served by the community school.

This attachment includes (a) the framework by which ESCLEW will assess the performance of the school, (b) the interventions and technical assistance that ESCLEW may utilize if ESCLEW consultants perceive the need, and (c) the academic and non-academic performance goals of the school.

# A. Performance Assessment Framework

The performance assessment details many items that ESCLEW will review in its oversight of the school. If an issue arises, ESCLEW may utilize intervention(s) listed in this attachment, including any technical assistance necessary to assist the school.

### **Overview – Areas of Assessment**

- Mission and Vision Statements
- Governance
- Fiscal or Financial Matters
- Operations, Recordkeeping, Reporting, and Data
- Special Education and Other Diverse Learning Needs
- Academics

#### **Mission and Vision Statements**

- Clear and specific mission statement
- Relevant and compelling vision statement
- Evidence of progress in meeting mission and vision
- Evidence of mission-driven decision-making in policies, curriculum, and programming

#### Governance

- Effective governance and management structure
- Oversight of school management and evaluation of management team
- Maintenance of corporate status: good standing certificate, timely filing of statement of continued existence, adherence to Code of Regulations, maintenance of corporate record-keeping, proper corporate signatures
- Maintenance of Federal tax-exempt status: timely filing of Form 990s, IRS conflict of interest adherence, no evidence of violation of Internal Revenue Code Section 501(c)(3)

- Compliance with contractual terms: governance trainings, updated governing authority roster, conflict of interest policies
- Disclosure of conflicts, signing of conflict of interest policy, and abstentions in minutes
- Compliance with Open Meetings law, including sufficient minutes
- Balance of educational and financial reporting at meetings
- Informed decision-making and operational transparency
- Any other area of operations, compliance or performance of the Contract

### Fiscal or Financial Matters

- Timely submission of financial documentation to Sponsor, including:
  - Balance sheet statement of financial position
    - Income statement statement of activities
  - o Monthly budget to actual, including all revenue sources
  - Monthly separated grants reporting
  - Monthly enrollment reports
- Red flag financial concerns: significant changes in enrollment, significant changes in policy environment, loss of a particular revenue stream, facility changes, significant turnover in leadership
- Timely payment on all loans/debts
- Financial audit, including timely follow-up on findings and/or material non-compliance
- Solid business plan and clear evidence of capacity to carry out plan
- Sound management and competence in understanding of proper use of public funds
- Grounded in professional standards for sound financial operations and sustainability
- Consistency of financial reporting at governing authority meetings
- Other finance related compliance reports

## Operations, Record-Keeping, Reporting, and Data

- Fair treatment, non-selective, and non-discriminatory access in admissions
- Fair and consistent treatment in discipline, with a focus on positive behavioral intervention supports
- Timely and accurate reporting to state data systems
- Appropriate attendance record-keeping and addressing excessive flagging
- Timely response to sponsor requests on interventions and routine compliance followups
- Compliance with public records law, including trained public records officer
- Annual report with clear and accurate performance data

## Special Education and Other Diverse Learning Needs

- Appropriate services for all students
- Appropriate access, education, support services and outcomes for students with special education and other diverse learning needs, including English language learners, migrant, and homeless students
- Compliance with policies and procedures for identification, evaluation, and referral of students with disabilities under IDEA and Section 504
- Compliance with all state and federal timelines and reporting requirements
- Proper due process notifications

• Appropriate discipline administration and discipline training for staff, including restraint, seclusion, and positive behavioral intervention supports

#### Academics

The School should be prepared to assess every measure listed below, and every component of every measure by data, in order to enhance targeted assistance. The School understands that, every year, additional measures and components will be added and graded and that an overall grade will be calculated starting in Fiscal Year 2015.

- Graduation Rate measures the percentage of students who entered in 9<sup>th</sup> grade and graduated 4 and 5 years later. Currently, Graduation Rate has a 4 year component (anything below an 84% will require targeted assistance) and a five year component (anything below an 85% will require targeted assistance).
- K-3 Literacy measures the improvement in reading for grades K-3. Currently, K-3 Literacy Improvement has one component, but may have more components in the future.
- **Prepared for Success** measures whether graduates are prepared for college or a career. Prepared for Success currently has one component, and it consists of College Admission Tests, Dual Enrollment, Industry Credentials, Honors Diplomas, and Advanced Placement and International Baccalaureate components.
- Achievement measures "absolute achievement." Achievement consists of the Performance Index (anything below a 70% will require targeted assistance). This assesses the achievement of every student (not just which students are proficient) and points are received for each level. The Achievement measure also consists of the Performance Indicators. The indicators show how many students have a proficient level of knowledge. Schools need to have 80% of their students reach proficient in order to meet an indicator. The indicators combine to show a letter grade and that letter grade must be a C, or 70% or more, to avoid targeted assistance.
- Progress represents the average annual level of improvement of each student. Its components consist of Value Added Overall, Gifted Students, Progress within the Lowest 20%, and Students with Disabilities (all must be greater than or equal to a -1 value added gain index in order to avoid targeted assistance).
- Gap Closing measures the narrowing of gaps in reading and math among subgroups of socio-economic, racial, ethnic, or disability. It is also called Annual Measurable Objectives (AMOs) (anything below a 70% will require targeted assistance).
- Other Measures of Academic Success, including:
  - Other assessments conducted by school (NWEA, etc.)
  - Ohio Improvement Process
  - Assessments and interventions required by Third Grade Reading Guarantee
  - o Incorporation of Common Core analytic understanding
  - Mission-specific performance measures of academic success
  - Comparison of student performance data with data of the home district building to which students would otherwise be assigned (suggested, and required upon request)
  - Post-secondary readiness, if applicable

# **B.** Technical Assistance, Interventions, and Statutory Consequences

The technical assistance and interventions listed below provide guidance and notice for the types of intervention that the Sponsor may take if the School fails to perform satisfactorily in the areas reviewed as through the performance assessment framework or if the School does not meet its contract goals. The Sponsor will make attempts to both streamline and customize interventions based on the School's operational structure and mission.

Many of the interventions listed below require the Sponsor to provide technical assistance. The Sponsor has attempted to limit financial cost to the School for other interventions in which the School must pay for the intervention. The School should budget for a certain amount of expense due to required performance interventions, programs to collect data, and staff to assess and analyze performance data, as well as increased staff for tutoring or cure of deficiencies.

All interventions will require follow-up, proof of cure, and adherence to cure. Typical interventions, remedies or cures imposed by Sponsor may be, but are not limited to the following:

### **Mission and Vision Statements Interventions**

- Sponsor technical assistance in reaching the mission of the School and to assess the ability of the School to realize its mission, which may include aspects of culture, parent satisfaction, or other visions stated by the School in its Contract, curriculum, or programming
- Sponsor assessment of leadership with the Board or the administration and mutual plan to correct weaknesses in meeting mission
- Outside coaching or training, research, observations, or remedies to meet stated goals, at School's cost.
- Leadership training in area of desired goals and why they are not being reached or how to reach them more effectively, at School's cost
- Other tailored interventions based on the specific circumstances

#### Governance Interventions

- Sponsor technical assistance
- Targeted technical assistance in procedure, minutes, motions, notices, or similar governance matters
- Workout session held by Sponsor
- Recommendations of Sponsor to be followed by Board as to dysfunctions and options for resolving them
- Review and required changes of Code of Regulation, governance policies, or other rules or practices affecting governance
- Clear guidance written into policy and adherence to guidance, as to roles and responsibilities
- Required replacement of absenteeism or members hampering business or quorum
- Required agreement of Board to dismiss board member or officer violating policy, law, procedure, rules, the Code, or other customary good governance precepts
- Training using trainer suggested by or acceptable to Sponsor, at School's cost
- Other tailored interventions based on the specific circumstances

#### Fiscal or Financial Matters Interventions

- Sponsor technical assistance
- Enhanced and more accurate financial reports required, such as additional financial documentation, debt- to- asset ratios, cash flow analyses, requirements of debt to service coverage ratios, enrollment variances and withdrawal frequencies, requirements of minimum unrestricted cash, and increased transparency enhancements for analytical purposes
- Compliance plans for recurring or material findings
- Contingency plans addressing potential funding shortfalls or other disruptive events
- Finance committee or finance expert on the board
- Pre-audit exercise with qualified Treasurer consultant, at the School's cost.
- Independent training, at the School's cost
- Other tailored interventions based on the specific circumstances

### Special Education and Other Diverse Learning Needs Interventions

- Sponsor technical assistance, including but not limited to Sponsor audit of files
- Legal review of practices and procedures targeted to issues discovered, at School's cost
- State audit of files and State training, at School's cost (if any cost required)
- Training, at the School's cost
- Independent Audit, at the School's cost
- Other tailored interventions based on the specific circumstances

#### Academic Interventions

- Sponsor technical assistance
- Targeted assistance by the Sponsor as defined below
- Face to face meetings and increased reporting to Sponsor on progress
- Checklists, plans to cure, and periodic data reporting
- Revisions to the Ohio Improvement Plan
- Required academic or professional assistance
- Parent Engagement to support students in academics, attendance, and curriculum
- Curriculum Mapping
- Study of Ohio's New Learning Standards and or the Next Generation of Assessments in Ohio
- OTES and OPES evaluations of staff
- Required training on the Ohio report card indicators or components
- Targeted coaching, at the School's cost
- Targeted tutoring before or after school and/or weekends, at the School's cost

### Targeted Academic Interventions and Assistance

Reports on academic data for all measures and components of the Ohio Local Report Card will be required and submitted to Sponsor by October 15 of each school year. A meeting will be required yearly (between October1 and December 15) between the Sponsor and the School's chief academic officer or administrator(s) to assess the readiness of the School to collect data, assess data, cure weaknesses, and report to the Sponsor. If the School is not fully prepared for this meeting, the School may be subject to some level of discipline.

Any D or F in any measure, or any relative weakness in a component or measure, will require targeted intervention. The first step of that intervention will be the School's submission of all data which make up the measure, or components of that measure, and which components have caused the grade of a D or F. The School must list each component with all proof of accurate data to support it, and the School must report quarterly on academic data for measures and components identified as weaknesses.

All academic interventions will require a thorough understanding of weaknesses, based on data and tools recommended by Sponsor, a thorough understanding of in which measure and which component within a measure the weakness occurs, and a targeted plan to cure that particular weakness or those weaknesses, whether it be Graduation Rate, K-3 Literacy Improvement, Prepared for Success, Achievement, Progress or Gap Closing, or any other measure now assessed or required to be assessed in the future.

Each plan to cure the weaknesses will require bi-monthly reporting to the Sponsor with a narrative as to how the plan is achieving progress. Should the plan be showing lack of progress, after two months of data, the plan will be re-evaluated and changed, with either additional interventions or supplements to the existing interventions.

#### **Comments on Statutory Consequences**

A non-renewal may occur in accordance with any cause allowed by Chapter 3314 of the Ohio Revised Code (Code) or the Contract.

Probation may be used for any cause allowed by the Code, including but not limited to:

- Untimely submissions or lack of cooperation in submissions required by the State or the Sponsor for two or more times per year, or for over five business days
- Lack of compliance with laws or the Contract
- Lack of progress on academics
- Any fiscal matter of any level of risk
- Governance, administrative, or minor financial issues that can be adequately corrected by a Plan to Cure acceptable to Sponsor
- Potential or actual ethics or conflicts violations

Suspension, which may lead to termination, is more likely to be used for matters such as, but not limited to:

- Health or safety matters
- Consistent lack of progress on academics
- Money owed to the State or its agencies, vendors, or the Sponsor that is delinquent, overextended, or may or does render insolvency
- Inability to make payroll
- Undisclosed debt or obligations
- Lack of accuracy in or misrepresentation, cheating, or fraud in any reporting figures, data, or testing results that is not corrected within timeframes allowed by law or which create financial, reputational, governance or academic risk
- Other matters of noncompliance, the magnitude or risk of which may be assessed by Sponsor as good cause or allowed by the Code
- Uncorrected or actual ethics or conflict violations

 Inability to meet, or lack of progress in meeting, the terms of the Contract, the laws, rules, regulations or requirements, or the reasonable requests of Sponsor geared to correct deficiencies

Permanent closure under ORC 3314.35 or 3314.351 will occur under the following circumstances:

- For schools with grades no higher than 3, for two of the past three years, the school has either been in academic emergency, received an F in improving literacy in grades K-3, or received an overall F.
- For schools with grades 4-8, but no higher than 9, for two of the past three years, the school has either been in academic emergency and not met value-added growth, received an F for the performance index score and an F for value-added progress, or received an overall F and an F for value-added progress.
- For schools with grades 10-12 that are not drop-out recovery schools, for two of the past three years, the school has either been in academic emergency, received an F for the performance index score and not met AMOs, or has received an overall F and an F for value-added progress
- For drop-out recovery and prevention program schools, for two of the past three years and beginning on July 1, 2014, the school received a designation of "does not meet standards."

Currently, schools in which a majority of the enrolled students are children with disabilities receiving special education and related services are exempt from statutory closure.

## C. School-Specific Performance Goals and Measures

This section of Attachment 11.6 provides school-specific academic and non-academic goals and performance measures. The school recognizes that these goals may be revised if the make-up of the school significantly changes or if the standards by which the school is judged by the state change. Schools should include goals for each year of the contract or up to five years. If a school has a contract length of longer than five years, a high-stakes review and goal evaluation will be performed after five years, and the school recognizes that it will re-assess goals at that time, if not sooner.

#### Academic Performance Goals

| Performance<br>Area | Metrics  | Goal(s)  |
|---------------------|--|--|
| Achievement         | Proficiency Level Indicators:<br>Percentage of students<br>scoring at or above proficient,<br>by grade and subject; Local<br>Report Card grade, percent,<br>and number of indicators met | The school will increase the number of students scoring proficient on the Ohio Achievement Assessments (OAA) annually. |
|                     | Performance Index:<br>Local Report Card grade,<br>score, and/or percent increase<br>from previous year   | The school will maintain or exceed a performance index score of 84 or make annual gains.                               |

| Performance<br>Area         | Metrics   | Goal(s)  |
|-----------------------------|---|--|
| Progress                    | Local Report Card grade and<br>value-added progress scores,<br>by subgroup, grade, and<br>subject   | The All Students subgroup will have a minimum gain index score greater than or equal to -1.  |
| Gap Closing                 | Local Report Card Annual<br>Measurable Objectives grade<br>and percentages, by subgroup<br>and subject  | The school will reduce the students with disabilities achievement gap by decreasing the number of students failing to achieve proficiency annually.  |
| Graduation<br>Rate          | Local Report Card grade and<br>percentages for four- and five-<br>year graduation rates   | Not applicable.  |
| K-3 Literacy<br>Improvement | Local Report Card grade,<br>percentage of students<br>achieving or exceeding<br>targeted gains, 3 <sup>rd</sup> grade<br>reading guarantee diagnostics<br>and assessments | The school will meet or exceed the statewide average for K-3 literacy growth.  |
| Prepared for<br>Success     | Local Report Card component<br>grade, percentage of students<br>scoring at each performance ·<br>level  | No applicable.   |
| Student<br>Performance      | Other valid and reliable assessments  | Each year, the school will administer a nationally-normed standardized assessment in grades K-8. The school will meet or outperform the national average of public school students in beginning-to-end-of-year learning growth in the majority of grades and subjects in reading and math in grades 2-8. |

# Non-Academic Performance Goals

| Performance<br>Area          | Measure(s)  | Goal(s)   |
|------------------------------|---|---|
| Attendance                   | Attendance levels as<br>measured by the Local Report<br>Card, school's attendance<br>system | Each year, the school will average a student attendance rate at or above 93%. |
| Post-Secondary<br>Enrollment |   | Not applicable.   |

| Performance<br>Area                               | Measure(s) | Goal(s)  |
|---|------------|--|
|   |            | Mission: Working in partnership with parents<br>and the community, the Pathway School of<br>Discovery mission is to become one of the finest<br>K-8 schools in the country.  |
| Mission<br>Statement                              |            | To ensure the school is one of the finest K-8 schools in the country, we propose the following academic growth goal:   |
|   |            | Each year, the school will administer a nationally-normed standardized assessment in grades K-8. The school will meet or outperform the national average of public school students in beginning-to-end-of-year learning growth in the majority of grades and subjects in reading and math in grades 2-8. |
| Parent<br>Satisfaction                            |            | Each year, the school will have an overall parent<br>satisfaction rate of 80% with at least 50% of<br>parents responding to a parent satisfaction<br>survey.   |
| Financial<br>Performance<br>and<br>Sustainability |            | Each year, the school will receive an unqualified audit.   |
| Governing<br>Board<br>Performance                 |            | Each year, the school's Board of Directors will<br>formally review the performance of, and provide<br>feedback to, its education management partner.<br>The review will be used to identify the<br>management partner's successes and<br>opportunities to improve its future performance.                |
| Other School-<br>Specific Goals                   |            | Store i vez z zachodni okok, s o z z zako i  |

#### MODIFICATION NO. 1 TO OHIO COMMUNITY SCHOOL CONTRACT BY and BETWEEN Educational Service Center of Lake Erie West ("Sponsor" or "ESCLEW") AND Pathway School of Discovery ("Governing Authority" OR "School")

WHEREAS, the ESCLEW and the Governing Authority entered into an amended and restated Community School Contract ("Contract") effective on July 1, 2012; and

**WHEREAS**, the ESCLEW and the Governing Authority agree to the following modifications, which shall be effective on July 1, 2013;

NOW THEREFORE, the parties modify the Contract as follows:

1. Article II, Section 2.1. The third sentence is updated to read: "Unless otherwise permitted by law, no Director shall serve on the board of more than four other community schools in the State of Ohio."

The remainder of Section 2.1 remains as originally written in the Contract.

2. Article II, Section 2.5 shall be replaced in its entirety as follows:

"Power of Attorney. The Governing Authority hereby grants to the Sponsor a power of attorney so the Sponsor may, but is not obligated to, assume operation of the School as provided for in Section 3314.073 of the Ohio Revised Code. Provided prior written notice is delivered to all members of the Governing Authority, the Sponsor may also replace the Governing Authority or any officer of the School, if the Governing Authority or its officer(s) fully resign or a majority of its members abandon(s) its/their duties hereunder or at law. The Governing Authority confirms its consent to this power by signing below, and shall execute and deliver to the Sponsor all agreements and other documents that the Sponsor reasonably shall deem necessary or appropriate to comply with the requirements of this subparagraph. Upon any failure by the Governing Authority to comply. In addition, any failure by the Governing Authority promptly to comply. In addition, any failure by the Governing Authority promptly to comply with the requirements of this subparagraph shall be good cause for termination of this Contract."

3. Article II. At the end, add a new section as follows:

"Section 2.8. Annual Contract Review. The Governing Authority agrees to meet with the Sponsor annually to review terms and requirements of this Contract and shall reserve at least one half hour at a Governing Authority meeting for that purpose."

4. Article IV Section 4.1 is updated as follows:

In the first paragraph, remove 2313.18 and include sections 2313.19, 3313.539, 3313.609, 3313.6411.

At the end, add the following paragraph: "If the School is a recipient of moneys from a grant awarded under the federal race to the top program, Division (A), Title XIV, Sections 14005 and 14006 of the "American Recovery and Reinvestment Act of 2009," the School will pay teachers based on performance in accordance with section 3317.141 and will comply section 3319.111 of the Revised Code as if it were a district."

The remainder of Section 4.1 remains as originally written in the Contract.

5. Article V, Section 5.1 shall be replaced in its entirety as follows:

"5.1 Location of Facilities. The facility to be used for the primary location of the community school will be maintained at 173 Avondale Drive, Dayton, Ohio 45404. The School may not open an additional facility without the prior approval of Sponsor and modification of this Contract. The School agrees to comply with all laws, rules, and regulations concerning multiple facilities.

5.2 Lease or Mortgage Payments. If any School facility has been or will be leased, a copy of the fully executed lease must be provided to the Sponsor within three (3) business days of its execution. If any School facility has been or will be purchased by the School, a copy of the contract of sale and related documents must be provided to the Sponsor within three (3) business days of execution, and, after purchase, a copy of the recorded conveyance documents must immediately be provided to the Sponsor. Any lease or use of any School facility must be documented in writing. The facility will not be changed without prior written notification to the Sponsor. Any lease or mortgage payments must be consistent with the budgets given to and approved by Sponsor. In any change of facility, the Sponsor, at its sole discretion, but without obligation to do so, may request maps, plans, revised budgets showing adequate service of the debt and reserves for maintenance or repairs, and/or attorney, accountant, or financial consultant assurances or opinions regarding structure, financing or otherwise. Sponsor may object to location of a proposed facility based on a business reason or an otherwise reasonable basis, but is not obligated to control or direct the marketing or facilities decisions of the School. The Sponsor shall not be liable for the debts obligations, or business of the School, but may request any information Sponsor deems necessary to assess adequate planning for facilities."

- 6. Article V, Section 5.2 is renumbered as Section 5.3. The language of the Section remains as originally written in the Contract.
- 7. Article VI, Section 6.8 shall be replaced in its entirety as follows:

"Assuring Adequate Yearly Progress and Value-Added Growth. If the School does not make Adequate Yearly Progress and/or meet state standards for value-added on the Ohio Local Report Card, the School shall develop a school-wide plan of intervention and submit the plan to Sponsor for approval. The School shall also develop a plan of intervention for each student not found proficient or who has not made a year's growth under value-added standards, and shall make such plans available for review by Sponsor."

8. Article VI, Section 6.11. The first sentence of the Section is replaced as follows: "Prior to graduation, the School shall send its list of graduates to Sponsor." The remainder of Section 6.11 remains as originally written in the Contract.

- 9. Article VI, Section 6.13. The last sentence is removed. The remainder of Section 6.13 remains as originally written in the Contract.
- 10. Article VIII, Section 8.1. At the end, add the following sentence: "Any classroom teacher hired on or after July 1, 2013 and employed to provide instruction in physical education must hold a valid license issued pursuant to R.C. 3319.22 for teaching physical education."
- 11. Article IX, Section 9.3 shall be replaced in its entirety as follows:

**"Fiscal Licensure.** Prior to assuming the duties of fiscal officer of the School, the fiscal officer shall be licensed as provided for in R.C. 3301.074. The licensure is attached as **Attachment 9.3**. Any updates or changes to the fiscal licensure must be sent to the Sponsor within five (5) business days."

12. Article IX, Section 9.4 shall be replaced in its entirety as follows:

**"Fiscal Bond.** The School's fiscal officer shall execute a bond in an amount and with surety to be approved by Sponsor, payable to the State of Ohio, conditioned on the faithful performance of all of the official duties required of the School's fiscal officer. The bond shall be deposited with and certified by the Governing Authority, and a copy thereof filed with the county auditor. A copy of the fiscal officer bond is contained in Attachment 9.4. Any updates or changes to the bond shall be sent to the Sponsor within five (5) business days."

13. Article IX, Section 9.7 shall be replaced in its entirety as follows:

"Payment to Sponsor for Oversight. For and in consideration of one and one-half percent (1.5%) of all state per-pupil foundation funds received by the School from the State of Ohio (but only up to 3% of such funds unless otherwise allowed by law), the Sponsor shall provide the oversight required by law, at the inception of this Contract. Beginning on July 1, 2015, this percentage will automatically increase to two percent (2%) of such funds. Payments to the Sponsor shall be made by monthly automatic transfer to the general fund of the Sponsor, and the School agrees to sign documentation necessary to accomplish the same.

At the inception of this Amended and Restated Contract, the oversight fee will remain at 1.5%, and it will increase to 2% on July 1, 2015, as outlined above. Beginning eighteen (18) months thereafter, the ESCLEW reserves the right to increase the fee upon not less than sixty (60) days prior written notice, for not more than 0.5% each increase, at any rolling 18-month interval during the term of this Contract, provided the total oversight fee shall never exceed statutory limits (currently 3%).

However,

- 1. It shall not be mandatory for the ESCLEW to raise any oversight fee;
- 2. Nothing shall prohibit the ESCLEW from raising the oversight fee to any increment below 0.5% of the last fee amount, and
- 3. Should the laws, rules, or regulations change to increase oversight fees or regulate how or from whom they are paid or otherwise, this Contract shall be amended to comply with such laws at the Sponsor's written request and, the School agrees to

automatically comply with such new rules even without such amendment, at the Sponsor's written request."

14. Article XI, Section 11.9 shall be replaced in its entirety as follows:

"Suspension. The Sponsor may suspend operations of the School if (a) conditions of the School do not comply with a health and safety standard established by law for school buildings; or (b) for (1) failure to meet student performance requirements stated in this Contract; (2) failure to meet generally accepted standards of fiscal management; (3) violation of any provision of this Contract or applicable state or federal law; or (4) other good cause, if Sponsor first sends a written notice of intent to suspend explaining the reasons and provides the Governing Authority with five (5) business days to submit a remedy, and either promptly reviews and reasonably disapproves the proposed remedy, or the Governing Authority fails to submit a remedy or fails to implement the remedy.

Once the School is suspended, it must cease operations on the next business day and immediately send notice to all School employees and parents stating that the School is suspended and the reasons therefore. The School again has an opportunity to submit a proposed remedy within five (5) business days. The School may not operate while the suspension is in effect, and any such suspension shall remain in effect until Sponsor notifies the Governing Authority that it is no longer in effect. At all times during suspension, the School remains subject to nonrenewal or termination proceedings, in accordance with the law."

ALL OTHER SECTIONS, SUBSECTIONS, TERMS, OR PROVISIONS OF THE CONTRACT SHALL REMAIN IN FULL FORCE AND IN EFFECT UNLESS OTHERWISE SPECIFICALLY MODIFIED HEREIN.

**Educational Service Center** of Lake Erie West (Signature)

Its: <u>Superintendent</u>

with full authority to execute this Contract for and on behalf of **Sponsor** and with full authority to bind **Sponsor**.

Date: \_\_\_\_\_/

**Governing Authority of** Pathway School of Discover Bv: (Signature)

Its: (Title)

with full authority to execute this Contract for and on behalf of **Governing Authority** and with full authority to bind **Governing Authority**.

Date: 05/15/13

### Revised 04-18-12

### AMENDED AND RESTATED CONTRACT FOR OHIO COMMUNITY SCHOOL

 This CONTRACT is entered into by and between the Educational Service Center of Lake Erie

 West ("ESCLEW or Sponsor), formerly known as the Lucas County Educational Service Center

 (LCESC or Sponsor) and the <u>Pathway School of Discovery</u>, (School

 Governing Authority, Governing Authority, or School) by and through

 <u>Kevin Robie</u> (name), the <u>President</u> (officer) of its

 Board of Directors and governing authority, an Ohio public community school established as a non-profit corporation under Ohio Revised Code (R.C.) Chapter 1702.

WHEREAS, R.C. Chapter 3314 permits Ohio public community schools; and

WHEREAS, ESCLEW is an authorized Sponsor under R.C. Chapter 3314; and

WHEREAS, the School Governing Authority is an Ohio non-profit Corporation with its corporate principal place of business located in <u>Montgomery</u> County, Ohio; and

WHEREAS, the School Governing Authority wishes to fully state or restate its Agreement to operate an Ohio Community School;

NOW THEREFORE, the School Governing Authority and the Sponsor enter into this Contract pursuant to the following terms and conditions. All Attachments and Recitals to this Contract are incorporated by reference and made a part of this Contract.

#### <u>ARTICLE I</u> <u>Continuation of Community School</u>

1.1 <u>Continuation of Start-Up Community School.</u> The School Governing Authority and the Sponsor agree that the School Governing Authority is a continuing start-up Ohio public community school [X], or a new start up community school [\_] (check one), subject to the laws of the State of Ohio and this Contract. The School covenants and agrees to Sections 1.2 through 1.6 below.

- 1.2 <u>Non-Profit Corporation.</u> The School is established and operated as a non-profit corporation under R.C. Chapter 1702. The School shall maintain in good standing its status as a non-profit corporation. The School shall hold all rights to the name of the School and any trade names or fictitious names.
- 1.3. <u>Public Benefit Corporation.</u> The School must be an Ohio Public Benefit Corporation under R.C. 1702.01(P), if formed after April of 2003.

- 1.4 <u>**Tax Exempt Status.**</u> The School may, but is not required to, qualify as a Federal tax exempt entity under Section 501(c)(3) of the Internal Revenue Code. Should the School so qualify, a copy of its federal tax exempt status determination letter must be forwarded to the Sponsor. Any change in tax status of the School must be reported in writing to the Sponsor within three (3) business days after notice to the School, with a copy of any documentation and official/governmental notices or letters.
- 1.5 <u>Corporate Documents.</u> Attached as <u>Attachment 1.5</u> are the Certificate of Incorporation, Articles of Incorporation, Appointment of Statutory Agent, Employer ID Number, Code of Regulations, IRS Determination Letter (if any), Mission Statement and Organizational Chart of the **School**. Any changes or updates in any of these documents must be reported in writing to the **Sponsor** within three (3) business days of the effective date of such changes, along with a copy of all documentation and filings.
- 1.6 <u>Prior Status.</u> The School was not a non-public chartered or non-chartered school in existence on January 1, 1997. This representation is material, and if in error, the Sponsor may terminate this Contract.
- 1.7 **Sponsor**. The **Sponsor** shall carry out the responsibilities established by law, including:
  - (a) Monitor the **School's** compliance with the laws applicable to the **School** and with the terms of this Contract;
  - (b) Monitor and evaluate the academic and fiscal performance and the organization of the School on at least an annual basis, and, if applicable, if Sponsor so elects under R.C. 3314.35(E) or 3314.36(D), the Sponsor may also evaluate the academics of the School for a period of at least three school years as described in <u>Attachment 1.7(b)</u>, and provide the results of this evaluation to the Ohio Department of Education and the parents of students enrolled at the School;
  - (c) Provide reasonable technical assistance to the **School** in complying with this Contract and with applicable laws (provided, however, **Sponsor** shall not be obligated to give legal advice to the **School** (*See* 2.7 below));
  - (d) Take steps to intervene in the School's operation to correct problems in the School's overall performance, declare the School to be on probationary status pursuant to R.C. 3314.073, suspend operation of the School pursuant to R.C. 3314.072, or terminate or non renew this Contract pursuant to R.C. 3314.07, as determined necessary by the Sponsor; and
  - (e) Establish and/or require a plan of action to be undertaken if the **School** experiences financial difficulties or closes before the end of the school year. Such plan or requirements for such plan shall be set out by the Sponsor as and when financial difficulties arise in a customized tailored manner to address the source of difficulties.

(f) Report on an annual basis the results of the evaluation conducted under division 1.7(b) above, to the department of education and to the parents of students enrolled in the community school.

### <u>ARTICLE II</u> Governing Authority/Administration

- 2.1 Attached as Attachment 2.1 are the names, home or Governing Authority Members. work addresses (not the address of the School), home, work and cellular telephone numbers, and electronic mail addresses of the current members of the School Governing Authority and a description of the process by which the members of the School Governing Authority shall be selected in the future. The School Governing Authority (its Board of Directors "Directors" or "Board") must contain at least five Directors. No Director shall serve on the board of more than one other community school in the State of Ohio. The Sponsor shall be promptly notified of any changes in members in writing (members or Directors of the Board) including names, notices of new names, addresses, e-mail and telephone numbers, within three (3) business days of such change. No present or former member, or immediate relative of a present or former member, of the Board of Directors of the School shall be an owner, employee, or consultant of the School's Management Company or operator, unless at least one year has elapsed since the conclusion of the member's service on the Board.
- 2.2 **Training of Governing Authority Members**. All Board members new to this School's Board will be required to go through a minimum of five (5) hours of Board training within three (3) months of being elected or appointed to the Board. Such training must be approved by the Sponsor. All Board directors or officers are required to obtain a clean BCI&I and FBI background check before the effective date of the member's term. The School shall obtain written consent from each Board member to give the background check to the Sponsor. The ESCLEW shall have adequate prior written notice of all regular and special meetings, be copied with all agenda, packets, handouts and minutes of all meetings of the Board of Directors or its committees. Sponsor must receive a hard copy notice and telephone call invitation to all special meetings at least 24 hours prior to such meeting, or, as soon as scheduled, whichever comes first. The ESCLEW representative shall be invited into executive sessions unless the session involves a legal dispute with the ESCLEW.
- 2.3 <u>Chief Administrative Officer.</u> The Chief Administrative Officer of the School will be called the <u>Principal</u> (principal, director, administrator, etc.). At the inception of this Contract, the position will be held by <u>Keith Colbert</u> \_\_\_\_\_\_. Any change in the identity and/or role of the Chief Administrative Officer shall be reported within three (3) business days to the Sponsor, along with that person's recently completed criminal background check and resume.

2.4 <u>Cooperation with Sponsor Oversight.</u> The School Governing Authority and administration covenant and agree to cooperate fully with the Sponsor in all activities as

required by regulations of the Ohio Department of Education for oversight of the **School**. This includes, but is not limited to:

- Annual file update per checklist as set out by the ESCLEW.
- Preliminary site visit and certification of letter of assurances at least 12 days prior to the first day of school and at all times thereafter as determined reasonably necessary by the Sponsor.
- Monthly reviews of financials and enrollment records.
- Other appropriate requests for information from the **Sponsor**, the Ohio Department of Education or other applicable governmental agencies.
- Timely submit all required or requested data into the ESCLEW/AOIS document management system.
- Maintain daily attendance sheets, signed and verified by the teacher and Principal of the School.
- Maintain high school drop-out recovery status (if applicable) and comply with all current and future rules, regulations and assessments associated with such status.
- 2.5 **<u>Power of Attorney.</u>** The School Governing Authority hereby grants to the Sponsor, a power of attorney to carry out all provisions of applicable law and this Contract on behalf of the School Governing Authority, should it become necessary in the Sponsor's reasonable discretion, and, subject to Ohio Open Meetings law, to appoint a new Board of Directors (all but one must be unrelated to Sponsor) for cause or for breach of this Contract, to all extents permissible by law. The School Governing Authority confirms its consent to this power by signing below, and shall execute and deliver to the Sponsor all agreements, and other documents that the Sponsor reasonably shall deem necessary or appropriate to comply with this subparagraph. Upon any failure by the School Governing Authority promptly to comply with the requirements of this subparagraph, the Sponsor shall be entitled to an order of specific performance from a court of law, ordering the School Governing Authority to comply. In addition, any failure by the School Governing Authority promptly to comply with the requirements of this subparagraph shall be good cause for termination of this Contract. In order to effectuate this provision, the Code of Regulations of the School must contain a provision allowing the Sponsor to appoint (non Sponsor related) Directors, and/or dismiss any Directors, if the Sponsor deems necessary at its discretion.
- 2.6 <u>General Training.</u> The School administrator, or appropriate representative, shall participate regularly in training provided by the ESCLEW and by the Ohio Department of Education, or the approved or affiliated organization of any of the preceding entities.
- 2.7 <u>Technical Assistance and Training by Sponsor.</u> The Sponsor may provide technical assistance and training to the School and its staff at such times and to the extent that the Sponsor deems appropriate or as the then current law requires. The School Governing Authority or its administrators have an obligation to attend training and receive technical assistance at the direction of the Sponsor.

### ARTICLE III Operations

- 3.1 <u>Student Transportation.</u> The School's plan for transportation of students is provided in <u>Attachment 3.1</u>. The School Governing Authority will work to assure that transportation of students is provided in accordance with all laws, rules and regulations, including to and from career technical programs.
- 3.2 Management by Third Parties. If the School Governing Authority enters into any contract for management or operation of the School and its curriculum and operations, such fully executed contract must be attached as Attachment 3.2. Any changes in the management company or contract shall require prior written approval of the Sponsor and incorporation as Attachment 3.2. If the management company provides services to the School in excess of twenty percent (20%) of the School's gross annual revenues, then the management company must provide a detailed accounting of the nature and costs of the services it provides to the School, acceptable to the Auditor of the State of Ohio. If at some point Sponsor deems, at its reasonable discretion, that due to mismanagement, governance, or performance, another company is necessary, Sponsor may require School to interview, select and enter into an agreement for such services and such management agreement must be approved by Sponsor. Any management agreement with the School must allow the Governing Authority to have its own legal counsel, paid for by the revenues of the School, should the School and management company be adverse to each other with any particular matter.
- 3.3 <u>Non-Sectarian</u>. The School shall be nonsectarian in its programs, admission policies, employment practices, and all other operations, and will not be operated by a sectarian school or religious institution.
- 3.4 <u>**Technology Plan.**</u> The School must have a state-approved technology plan within sixty (60) days of execution of this Contract and complete the filing procedures for E-tech Ohio or any successor thereof on or before the next applicable deadlines. All technology plans must be updated and submitted as required by law, rule or regulation.
- 3.5 <u>Disposition of Assets.</u> In the event that this Contract is a) suspended and terminated or b) not renewed and not reassigned to or sponsored by another authorized sponsor, or c) the School dissolves, the operation of the School will cease as a community school. The following requirements and procedures apply regarding the School Governing Authority and the School.
  - 3.5.1 After paying or adequately providing for the payment of all known obligations of the corporation, the Directors shall distribute the remainder of the assets as follows:
    - (a) Assets held upon condition requiring return, transfer, or conveyance, which condition shall have occurred by reason of the dissolution or otherwise, shall be returned, transferred or conveyed in accordance with such requirements;

(b) In the case of a public benefit corporation:

(i) Assets held by it in trust for specified purposes shall be applied so far as is feasible and in accordance with the terms of the trust,

(ii) The remaining assets not held in trust shall be applied so far is feasible towards carrying out the purposes stated in the School's articles,

(iii) In the event and to the extent that, and in the judgment of the directors, it is not feasible to apply the assets provided in above clauses (a) and (b), the assets shall be applied as may be directed by the court of common pleas and the county and this state in which the principle office of the corporation is located.

- 3.5.2 In applying the principles stated above, adherence to R.C. 3314.074(A) and (B) should be respected to the fullest extent not in conflict with the above.
- 3.5.3 The School shall comply with and cooperate with the closing requirements summarized on <u>Attachment 3.5.3</u> and all other required procedures, including any Ohio Department of Education Closing Procedure Guidance at the pertinent time.
- 3.6 <u>Commencement of School Operations.</u> The School shall open for operation not later than September 30<sup>th</sup> of each school year, unless the mission of the School is solely to serve dropouts. If the School fails to open within eight (8) months after the execution of this Contract, this Contract may be deemed void in a writing sent by Sponsor.
- 3.7 **Policies and Parent Surveys.** The School shall send to **Sponsor**, copies of policies and handbooks, a parent involvement policy, and any parent survey measuring parent satisfaction with the operation of the School.

### <u>ARTICLE IV</u> Compliance With Laws

4.1 The School shall comply with sections 9.90, 9.91, 109.65, Compliance with Laws. 121.22, 149.43, 2151.357, 2151.421, 2313.18, 3301.0710, 3301.0711, 3301.0712, 3301.0714 (as stated in 3314.17), 3301.0715, 3313.472, 3313.50, 3313.536, 3313.608, 3313.6012, 3313.6013, 3313.6014, 3313.6015, 3313.643, 3313.648, 3313.66, 3313.661, 3313.662, 3313.666, 3313.667, 3313.67, 3313.671, 3313.672, 3313.673, 3313.69, 3313.71, 3313.716, 3313.718, 3313.719, 3313.80, 3313.801 (as if the school were a district), 3313.814, 3313.816, 3313.817, 3313.86, 3313.96, 3319.073, 3319.321, 3319.39, 3319.391, 3319.41, 3321.01, 3321.041, 3321.13-.14, 3321.17-.191, 3327.10, 4111.17, 4113.52 and 5705.391, Chapters 117., 1347., 1702., 2744., 3307, 3309, 3314., 3365., 3742., 4112., 4123., 4141., and 4167. of the Ohio Revised Code as if it were a school district. Unless prohibited by the United States Constitution, the Ohio Constitution, or other controlling law, the School will comply with these sections and chapters of the Ohio Revised Code now in effect and as hereafter amended. Certain laws listed above which are not specified therein as mandatory, are permissive, unless otherwise

specifically required under this Contract. Laws listed above which are mandatory, are also mandatory under this Contract.

The **School** shall comply with Chapter 102 of the Ohio Revised Code and section 2921.42 of the Ohio Revised Code. The **School** must have a conflict of interest policy, attached as <u>Attachment 4.1</u>.

The **School** shall also comply with R.C. 3302.04, including division (E) of that section to the extent possible, except that any action required by a school district under R.C. 3302.04 shall be taken by **Sponsor**. The **Sponsor**, however, shall not be required to take any action under R.C. 3302.04(F).

The **School** shall comply with R.C. 3313.614, and with R.C. 3313.61 and 3313.611 as qualified by R.C. 3314.03(A)(11)(f) (See 6.11 below).

4.2 <u>Compliance With Other Laws</u>. The School and the School Governing Authority may carry out any act or insure the performance of any function that is in compliance with the United States Constitution, the Ohio Constitution, Federal law, Ohio law or this Contract. The School is not exempt from federal laws, rules and regulations, or Ohio laws granting rights to parents. The School specifically acknowledges that federal laws, rules and regulations apply to its operation, including but not limited to those having to do federal grants.

## ARTICLE V Facilities

5.1 Location of Facility. The facility to be used for the community school will be maintained at <u>173 Avondale Dr.</u>, Dayton, OH 45404 . If the facility has been or will be leased, a copy of the fully executed lease must be provided to the Sponsor within three (3) business days of its execution. If the facility has been or will be purchased by the School, a copy of the contract of sale and related documents must be provided to the Sponsor within three (3) business days of execution, and after purchase, a copy of the recorded conveyance documents shall immediately be provided to the Sponsor. Any lease or use of the facility must be documented in writing. The facility will not be changed and the number of square feet used will not be reduced without prior notification to the Sponsor. Any lease or mortgage payments must be consistent with the budgets given to and approved by Sponsor. In any change of facility, the Sponsor, at its sole discretion, but without obligation to do so, may request maps, plans, revised budgets showing adequate service of the debt and reserves for maintenance or repairs, and/or attorney, accountant or financial consultant assurances or opinions regarding structure, financing or otherwise. Sponsor may object to location of a proposed facility based on a business reason or an otherwise reasonable basis, but is not obligated to control or direct the marketing or facilities decisions of the School. The Sponsor shall not be liable for the debts, obligations or business of the School, but may request any information Sponsor deems necessary to assess adequate planning for facilities.

Pathway School of Discovery By

5.2 <u>Compliance with Health and Safety Standards.</u> Any facility used for or by the School shall meet all health and safety standards established by law for community school buildings. The School shall certify all Sponsor assurances required by law, rule or regulation to be sent to the Ohio Department of Education, or, after any walkthrough or site visit. Facilities will be maintained in clean, healthy manner to the satisfaction of the Sponsor and/or as indicated by proper authorities. Copies of all current permits, inspections and/or certificates must be filed with the Sponsor. The School must keep all permits, inspections and/or certifications current and compliant.

A Certificate of Occupancy must be provided to the **Sponsor** prior to occupancy. Proof of occupancy shall be satisfied by the **School's Governing Authority** providing to the **Sponsor** any permanent, interim or temporary certificate of occupancy issued by the government agency having jurisdiction over the same.

### <u>ARTICLE VI</u> <u>Educational Program</u>

6.1 Number of Students. The School will provide learning opportunities to a minimum of twenty-five (25) students; and as applicable, for a minimum of Nine Hundred Twenty (920) hours per school year or in accordance with any applicable changes of law. The School shall serve grades K-8 and serve ages \_\_\_\_\_\_N/A\_\_\_\_. The number of students attending the School at any one time shall not exceed the number allowed by the occupancy permit (including staff), or 752 students, whichever is less. An increase in the number of students may not occur without the prior written consent of the Sponsor. An involuntary and material decrease in enrollment may be a sole good cause for nonrenewal, suspension or termination at the discretion of the Sponsor. A material decrease in enrollment for purposes of this 6.1 shall be an unplanned decrease not preapproved by Sponsor of (i) below 25 students, or, (ii) 40% or greater decrease from the average enrollment figures of the prior four or more school operating (open for students) months. ....

Pathway School of Discovery

- 6.2 <u>Continuing Operation.</u> The School agrees to continue operation by teaching the minimum number of students permitted by law. Time is of the essence in continuing operation. Failure to continue operation without interruption (except due to calamity or Act of God, defined as war, or severe weather casualty such as fire, tornado or flood), is grounds for termination of this Contract. Only upon written notification to the Sponsor, can the School calendar be materially changed. A material change shall be defined as any change of one week or more, either consecutively or cumulatively.
- 6.3 <u>Curriculum</u>. For purposes of this Contract, on <u>Attachment 6.3</u>, the focus, mission, philosophy, goals and objectives of the curriculum shall be separated from the methods used to achieve those goals and all classroom and non-classroom based learning opportunities. Any change in methods must be reported to **Sponsor**. However, changes in the focus, mission, philosophy, goals and objectives may not be changed without prior written notice to **Sponsor** containing justification and sound reasoning, and prior

approval of **Sponsor**. <u>Attachment 6.3</u> must also show how the **School's** curriculum is aligned with Ohio Content Standards. The ESCLEW specifically authorizes "Learning Opportunities" to include educational opportunities provided by the School during suspension or expulsion of the School's students, if provided for by the School's own policies. "Learning Opportunities" shall also include any opportunities provided for in a Credit Flex program of the School or as described on <u>Attachment 6.3</u>.

- 6.4 Achievement Tests, Value-Added Assessments, Other Assessments and Standards. The performance standards (requirements) and assessments which shall include statewide proficiency or achievement tests and any other standards and/or assessments required by law or recommended by the Sponsor, must be timely and properly administered, met and completed. Initial performance standards/requirements are attached as Attachment 6.4. The School shall also be evaluated based upon state standards. State standards shall be met by the School and may be changed from time to time by the Ohio Department of Education. In addition to the required testing, the School must assess and keep benchmarks acceptable to the Sponsor, of all students, in order to provide guidance for the Sponsor to review yearly progress. Such assessments and intended benchmarking are identified on Attachment 6.4. No assessment tools may be changed without notification to the Sponsor, which shall contain the justification for the change. The School specifically acknowledges the closure provisions of the Ohio Revised Code section 3314.35.
- 6.5 **<u>Racial and Ethnic Balance</u>**. The School will attempt to achieve and attempt to continue, as the case may be, racial and ethnic balance reflective of the community it serves by doing each of the items recited in <u>Attachment 6.5</u>. Notwithstanding the admissions procedures of the School, in the event that the racial composition of the enrollment of the School is violative of a federal desegregation order, the School shall take any and all corrective measures to comply with desegregation order.
- 6.6 <u>**Tuition.**</u> Subject only to any applicable exception pursuant to section 3314.26 of the Ohio Revised Code, tuition in any form shall not be charged for the enrollment of any student. Nothing in this section prevents reasonable activity or class fees as allowed by law, or the **School's** engaging in voluntary fund-raising activities or the School's ability to otherwise charge tuition or fees if expressly allowed by applicable statute.
- 6.7 <u>Suspension and Expulsion Policies.</u> A policy regarding suspension, expulsion, removal and permanent exclusion of a student that specifies among other things the types of misconduct for which a student may be suspended, expelled or removed and the due process related thereto is included in <u>Attachment 6.7</u>. The School's policy and practices pursuant to the policy shall comply with the requirements of sections 3313.66, 3313.661, and 3313.662 of the Ohio Revised Code. Those policies and practices shall not infringe upon the rights of handicapped students as provided by state and federal law, and, the School must also maintain a separate policy for the discipline of students receiving special education services (also in <u>Attachment 6.7</u>).
- 6.8 <u>Assuring Adequate Yearly Progress</u>. The School shall develop a plan of intervention for all students not found proficient or not making adequate yearly progress, and submit it to the Sponsor for approval.
- 6.9 **Disabled Students**. Upon admission of any disabled student, the **School** shall comply with federal and state laws regarding the education of handicapped students. The **School** shall provide all necessary related services, or, the **School** may contract for related services if it provides documentation to **Sponsor** identifying the providers, along with the plan to provide related services, the provider's qualifications, experience and reputation.
- 6.10 <u>School Closure.</u> The School agrees to remain open for students to attend until the end of the school year in which it is determined that the School must close. The programs provided to students in the final year of the School must continue without interruption or reduction unless program changes are approved in writing by the Sponsor. The Sponsor may, at its sole discretion, operate the School in the event the School Governing Board fails to continue until the end of the approved school year or is otherwise suspended or terminated. Provided however, Sponsor may suspend the operations or terminate the contract as otherwise indicated by law. The School Governing Authority and its administration takes sole responsibility for the ODE closing procedures listed in Attachment 3.5.3 and indemnifies, shall defend and hold harmless the Sponsor for all performance thereof.
- 6.11 High School Diplomas. Within two (2) business days from when School receives its information of which students passed all required tests for graduation, the School shall send its list of graduates to Sponsor. Sponsor shall be invited to all graduation ceremonies. The School shall comply with the requirements of the Ohio Core Curriculum and/or allowable waivers thereof. If the School is a high school awarding a diploma, the School shall comply with sections 3313.61 and 3313.611 of the Ohio Revised Code except that, by completing the curriculum adopted by the School Governing Authority the student will be deemed to have met the requirement that a person must successfully complete the curriculum specified in Title 33 of the Ohio Revised Code. Provided, however, beginning with students who enter ninth grade for the first time on or after July 1, 2010, the requirement in sections 3313.61 and 3313.611 [3313.61.1] of the Revised Code that a person must successfully complete the curriculum of a high school prior to receiving a high school diploma shall be met by completing the Ohio core curriculum prescribed in division (C) of section 3313.603 [3313.60.3] of the Revised Code, unless the person qualifies under division (D) or (F) of that section. Each school shall comply with the plan for awarding high school credit based on demonstration of subject area competency, adopted by the state board of education under division (J) of section 3313.603 [3313.60.3] of the Revised Code.
- 6.12 <u>Admissions Policy</u>. The admissions and enrollment procedures of the School are attached hereto as <u>Attachment 6.12</u> and shall be followed and may not be changed without the prior written consent of the Sponsor. At a minimum, the admission procedures at all times must:
  - (a) specify that the school will not discriminate in its admission of students to the school on the basis of race, religion, color, national origin, handicap, intellectual ability, athletic ability or measurement of achievement or aptitude [If the School limits admission to "at-risk" See (b) below];

(b) be open to any individual entitled to attend school in the State of Ohio pursuant to section 3313.64 or section 3313.65 of the Ohio Revised Code, except that admission to the school may be limited to (i) students who have obtained a specific grade level or are within a specific age group; (ii) students who meet a definition of "at-risk" that the parties to this Contract agree upon and/or; (iii) residents of a specific geographic area that the parties to this Contract agree upon.

If the number of applicants meeting admission criteria exceeds the capacity of the **School's** programs, classes, grade levels or facilities, students shall be admitted by lot from all eligible applicants, except preference shall be given to students attending the **School** the previous year and may be given to eligible siblings of such students. The lottery may be conducted by the **Sponsor**.

- (d) The **School** shall adopt a policy regarding the admission of students residing outside the district in which the **School** is located, which shall comply with the administrative procedures specified herein and shall either permit the enrollment of students who reside in districts adjacent to the district in which the **School** is located or permit the enrollment of students who reside in any other district in the state. Such a policy is included in <u>Attachment 6.12</u>.
- 6.13 <u>Attendance Policy</u>. <u>Attachment 6.13</u> must include the School's Attendance and Truancy Policies, as well as procedures for automatic withdrawal if a student misses 105 consecutive hours of learning opportunities. Attendance must be itemized and verified daily by the applicable teacher(s).

### ARTICLE VII Reporting

7.1 <u>Annual Report.</u> The School Governing Authority shall submit not later than October 30 of each year to the Sponsor and to the parents of all students enrolled in the School, or any other statutorily required parties, its financial status, and the annual report of its activities and progress in meeting the goals and standards of this Contract, its activities and standards.

- 7.2 <u>Reports to Sponsor.</u> The School Governing Authority shall report to the Sponsor the following on or before the day set by statute, rule or regulation, or, by the Sponsor, all information or documents required under applicable law, including but not limited to:
  - (a) that information required under R.C. 3314.08(B)(2)(a)-(i); and
  - (b) the number of student suspensions and expulsions and the number of students suspended and expelled; and
  - (c) the number and names of students withdrawn; and
  - (d) annually, or at the request of Sponsor, any event, occurrence or circumstance that could reasonably have a material adverse effect on the operations, properties (both real or personal), assets, condition (financial or otherwise), enrollment, prospects or reputation of the School, including but not limited to notice of breach of this Contract; breach, claimed default or violation of any covenant or term of any loan or contract with a third party; failure to maintain in good standing its non-profit corporate charter or its tax exempt status; failure to comply with any conditions of permits, licenses, certificates or other regulatory or similar obligations; threats of or claims for lawsuits, actions, investigations, arbitrations or mediations; and, the School Governing Authority's position, cure, or plan of action; and
  - (e) all items required to be reported in this Contract, or by the Ohio Department of Education, including but not limited to those listed on <u>Attachments 7.2(h)</u>; and
  - (f) all financials, budgets, assets, liabilities, or similar information and enrollment records (required monthly); and
  - (g) staff and teacher turnover; and
  - (h) any structure, governance or operational information.
- 7.3 <u>Site Visits</u>. The Sponsor shall be allowed to observe the School in operation at site visits at Sponsor's request and shall be allowed access for such site visits or other impromptu visits as Sponsor deems advisable or necessary.

### ARTICLE VIII Employees

8.1 <u>Employment of Teachers.</u> At least one (1) full-time classroom teacher or two (2) parttime classroom teachers each working more than twelve (12) hours per week must be employed by the School. The full-time classroom teachers and part-time classroom teachers teaching more than twelve (12) hours per week shall be certified or licensed in accordance with R.C. 3319.22 to 3319.31, or other applicable sections of the Ohio Revised Code. Upon employment, the School shall forward teacher qualifications to Sponsor. The School may employ non-certificated persons to teach up to twelve (12) hours per week pursuant to R.C. 3319.301, to the extent permitted by the No Child Left Behind Act. The school-wide students to full-time equivalent classroom teacher ratio shall be no more than <u>28 to 1</u>. The **School** may also employ necessary non-teaching employees. Prior to opening day, the **School** will provide the **Sponsor** with proof of Ohio certification for a sufficient number of teachers to support the stated teacher/student ratio, as well as the credentials and background checks for all staff of the **School**. All teachers and para-professionals shall meet the "highly qualified" standards as applicable and as set out in the law known as "No Child Left Behind" or its successors law(s). The laws and rules governing LPDC must be implemented by the School.

- 8.2 <u>Dismissal of Employees.</u> Subject to 11.2 below, the School Governing Authority may employ administrators, teachers and non-teaching employees necessary to carry out its mission and fulfill this Contract, so long as no contract of employment extends beyond the term of this Contract. The dismissal procedures for staff and the plan for disposition of employees if this Contract is terminated, are set out on <u>Attachment 8.2</u>.
- 8.3 <u>Employee Benefits.</u> Any health, medical or other benefits provided by the School shall be set out in <u>Attachment 8.2</u>. <u>Attachment 8.2</u> may be amended by the School from time to time, provided however, all such amendments shall be provided to Sponsor in writing within three (3) business days of amendment or change. In the event certain employees have bargained collectively pursuant to Chapter 4117 of the Ohio Revised Code, the collective bargaining agreement supersedes <u>Attachment 8.2</u> to the extent that the collective bargaining agreement provides for health and other benefits. Any collective bargaining agreement shall not, under any circumstances, be a part of this Contract.
- 8.4 The School Governing Authority must request that **Criminal Background Check.** the superintendent of the Bureau of Criminal Identification & Investigation and the Federal Bureau of Investigation conduct a criminal background records check for any applicant who has applied to the School for employment, in any position, as a person responsible for the care, custody and control of a child, including those who may be in unsupervised contact with a child. The School shall obtain written consent from such persons in order to give such background checks to Sponsor. In the alternative, if consent cannot be obtained, the School may either (a) provide an attorney opinion letter or (b) provide an affidavit from the Board President or Chairperson, in either case stating that the checks are either 1) clean or 2) reviewed and approved by legal counsel, on a timely basis. An applicant may be employed conditionally until the criminal records check is completed and the results of the criminal records check are received. If the results of the criminal records check indicate that the applicant does not qualify for employment or care, custody or control of, or unsupervised contact with children, the applicant shall be released from employment. The School shall conduct such background checks for renewals of licenses and shall follow all required reporting of misconduct to the State Superintendent of Public Instruction.

All volunteers must be notified that the **School** may require a background check of the volunteer at any time, at the **School's** request.

### ARTICLE IX <u>Finance</u>

- 9.1 **Financial Records.** The **School's** financial records will be maintained in the same manner as are financial records of school districts, pursuant to rules of the Auditor of the State, and audits shall be conducted in accordance with section 117.10 of the Ohio Revised Code.
- 9.2 <u>Fiscal Services.</u> The School agrees that its fiscal agent shall be (pick and initial only one):

| by<br>ESCLEW by<br>(initial)      | 1. | Its Treasurer,; or   |
|-----------------------------------|----|--|
| PSD by KAR<br>ESCLEW by (initial) | 2. | Its Management Company, <u>National Heritage Academies</u> ,<br>through the Management Company's Treasurer <u>Kathy</u><br><u>Schaaphok (Schmidt) (name)</u> ; |
| by<br>ESCLEW by<br>(initial)      | 3. | The <b>ESCLEW's</b> Treasurer by way of a Fiscal Services<br>Agreement. (Note requirements in Attachment 9.2)  |
| by<br>ESCLEW by<br>(initial)      | 4. | A qualified service provider named<br>(credentials and training to be provided to the <b>Sponsor</b> .   |

Should the **School** be using the **ESCLEW's Treasurer**, the Fiscal Services Agreement is as stated in <u>Attachment 9.2</u>. Payments to the **Sponsor** for fiscal services are as stated in such Fiscal Services Agreement and are in addition to the oversight fee in §9.7 below.

If 9.2, 1., 2. or 4 are chosen, the School Treasurer must be obligated to assist in all closing procedures and Ohio Department of Education closing requirements listed on Attachment and Supplement 3.5.3., and to participate in any final audit if the School closes, is non-renewed, suspended or terminated.

If the School does not use the Sponsor for its Treasurer, the School's Treasurer must be obligated to assist in all closing procedures, regular, special or final, audits, closing requirements listed on Attachment 3.5.3, and all Ohio Department of Education rules and procedures, even if the School closes, is not renewed, suspended or terminated.

9.3 **Fiscal Licensure.** Prior to assuming the duties of fiscal officer, agent and/or fiscal servicer of School, the fiscal officer, agent or service provider shall be licensed as provided for in Ohio Revised Code 3301.074 or shall have completed not less than 16 hours of continuing education classes, courses or workshops in the area of school accounting as approved by the **Sponsor**, completed an additional 24 hours of continuing education classes or workshops in the area of school accounting as approved by the **Sponsor**, and have completed or will complete 8 hours of continuing education classes, courses or workshops in the area of school accounting as approved by the **Sponsor**, and have completed or will complete 8 hours of continuing education classes, courses or workshops in the area of school accounting as approved by the

**Sponsor** during each subsequent year. The Certification of Training or licensure is attached as <u>Attachment 9.3</u>.

- 9.4 **Fiscal Bond.** Fiscal agent, officer and/or service provider shall execute a bond in an amount and with surety to be approved by Sponsor, payable to the State of Ohio, conditioned on the faithful performance of all of the official duties required of the School fiscal agent, officer or service provider. The bond shall be deposited with Sponsor, and a copy thereof, certified by Sponsor, shall be filed with the county auditor. A copy of the fiscal agent bond is contained in <u>Attachment 9.4</u>.
- 9.5. Prior to its first opening, the School must submit a balanced budget for 25 Budget. students, 50 students and 80 students. A continuing start-up school may submit its budgets based on historical data. A financial plan detailing an estimated school budget for each fiscal year of this Contract, and for five years, is attached as Attachment 9.5. Each year of this Contract, on or before June 30, a revised school budget shall be submitted to the Sponsor. The budget must detail estimated revenues and expenses. Revenues include the base formula amount that will be used for purpose of funding calculations under section 3314.08 of the Ohio Revised Code. Should the Sponsor request further breakdown of revenue or expenses, or line items for expenses or revenue not projected, the School agrees to revise or comply with such requests. Should the School be managed by a third party management company, the School Governing Authority must procure from such management company, sufficient data, at the Sponsor's reasonable discretion, to allow the Sponsor to review revenue and expenses as required or permitted by law.
- 9.6 **Borrowing Money.** The School may borrow money to pay necessary and actual expenses of the School in anticipation of receipt of any portion of the payments to be received by the School. The School may issue notes to evidence such a borrowing. A copy of all notes must be provided to the Sponsor within five (5) business days of signing. The proceeds from the notes shall be used only for the purpose for which the anticipated receipts may be lawfully expended by the School. The School may borrow money for a term not to exceed fifteen (15) years for the purpose of acquiring facilities.
- 9.7 <u>Payment to Sponsor for Oversight.</u> For and in consideration of one and one-half percent (1.5%) of all state per-pupil foundation funds received by the School from the State of Ohio (but only up to 3% of such funds unless otherwise allowed by law), the Sponsor shall provide the oversight required by law. Payments to the Sponsor shall be made by monthly automatic transfer to the general fund of the Sponsor, and the School agrees to sign documentation necessary to accomplish the same.

At the inception of this Amended and Restated Contract, the oversight fee will be set at or remain at 1.5% as described above. The ESCLEW reserves the right to increase the fee upon not less than sixty (60) days prior written notice, for not more than .5% each increase, at any rolling 18-month interval during the term of this Contract, provided the total oversight fee shall never exceed statutory limits (currently 3%).

### However,

- 1. It shall not be mandatory for the ESCLEW to raise any oversight fee; and
- 2. Nothing shall prohibit the ESCLEW from raising the oversight fee to any increment below .5% of the last fee amount, and
- 3. The School may exercise an option to waive any increase, once during the first 35-month period of this Contract only, as measured from the proposed effective date of the last Sponsor-requested raise in fee if:
  - a. the School's enrollment at the time of the notice of increase is 1,000 students or more on the most recent past month's CSADM report; or
  - b. the School's most recently officially released academic Ohio Report Card Rating is Effective or higher, and (i) there are no findings in the school's last officially released annual state audit, no special education, FTE federal grant or lunch audit findings and no special audits or other administrative audits then being conducted.
- 4. Should the laws, rules or regulations change to increase oversight fees or regulate how or from whom they are paid, or otherwise, this Contract shall be amended to comply with such laws at the Sponsor's written demand, and, the School agrees to automatically comply with such new rules even without such amendment, at the Sponsor's written demand.
- 9.8 <u>Federal Grants.</u> School must report to Sponsor the sources of funds used to acquire property and equipment used at the School, clearly delineated and tracking those acquired with federal funds, and shall report to Sponsor the disposition of assets acquired with federal funds with an explanation as to how disposed and backup for proper disposition in accordance with the applicable law and grant rules. The School also acknowledges that certain federal grants, including but not limited to the federal food programs may require bidding processes not otherwise required under state law.
- 9.9 **Fiscal Year.** The fiscal year for the **School** shall be July 1 to June 30.

### <u>ARTICLE X</u> <u>Insurance/Indemnification</u>

10.1 <u>Liability Insurance.</u> Comprehensive general liability insurance at all times will be maintained by the School Governing Authority in amounts not less than one million dollars (\$1,000,000) per occurrence and three million dollars (\$3,000,000) in the aggregate, plus an excess or umbrella policy extending coverage as broad as primary coverage in an amount no less than Five Million Dollars (\$5,000,000). The insurance coverage shall be not only for the School and the School Governing Authority, its Directors, Officers and its employees but also for the Sponsor, its Board, Superintendent and employees as additional insureds. The School Governing Authority shall also

maintain directors and officers liability/errors and omissions coverage in the amount of One Million Dollars (\$1,000,000) per occurrence. The insurance coverage must be occurrence coverage rather than claims made coverage. The School Governing Authority must obtain policies that notify the Sponsor in writing at least thirty (30) days in advance of any material adverse change to, or cancellation of, such coverage; and shall provide evidence of the same as <u>Attachment 10.1</u>. School must provide copies of all comprehensive general liability, real or personal property, directors and officers liability insurance, proof of worker's compensation payments and unemployment compensation payments and notice of lapse of any such coverage to Sponsor within three (3) business days of request, and, within three (3) business days of any change or notice to School by the applicable insurance entity.

- 10.2 <u>Indemnification.</u> The School Governing Authority and School shall defend, indemnify, save and hold harmless the Sponsor and its Board, Superintendent, officers, employees and agents from any and all claims, debts, actions, causes of actions, proceedings, judgments, mitigation costs, fees, liabilities, obligations, damages, losses, costs, or expenses (including, without limitation, attorneys', expert, accounting, auditors or other professionals' fees and court costs) of whatever kind or nature in law, equity or otherwise (collectively "Liabilities") arising from any of the following:
  - (a) A failure of the **School Governing Authority** and/or **School** or any of its officers, directors, employees, agents or contractors to perform any duty, responsibility or obligation imposed by law or this Contract;
  - (b) An action or omission by the School Governing Authority and/or School or any of its officers, directors, employees or contractors that results in injury, death or loss to person or property, breach of contract, or violation of statutory law or common law (state and federal), or Liabilities;
  - (c) Any sum that the **Sponsor** may pay or become obligated to pay on account of: (1) any inaccuracy or breach of any representation under this Contract by the School and/or Governing Authority; (2) any breach or any failure of the **School** to duly perform, comply with, or observe any term, provision, covenant, agreement, obligation, or condition under this Contract or under the law, and all agreements delivered in any way connected herewith, on the part of the **School**, to be performed, complied with, or observed; or (3) Liabilities to lenders, vendors, the State of Ohio, receivers, parents, students, their heirs or representatives, the **School** Governing Authority or to third parties in any way related to the **School** or its operations or in any way related to closure, termination or suspension of the School; and
  - (d) Any Liabilities incurred by Sponsor or any of its officers, directors, employees, agents or contractors as a result of an action or legal proceeding at law or equity brought against Sponsor by the School or the School Governing Authority unless the School or School Governing Authority obtains a final judgment or order on the merits against the Sponsor, and the right to appeal such judgment or order has been exhausted or has expired, excepting out only those liabilities directly caused by Sponsor's gross negligence or willful misconduct.

- 10.3 <u>Indemnification if Employee Leave of Absence.</u> If the Sponsor provides a leave of absence to a person who is thereafter employed by the School, the School Governing Authority and the School shall indemnify and hold harmless the Sponsor and its board members, Superintendent, employees and agents from liability arising out of any action or omission of that person while that person is on such leave and employed by the School Governing Authority.
- 10.4 **Survival**. All provisions of Article X, section 10.2 shall survive the voidance, expiration, termination, nonrenewal, suspension or abandonment of this Contract.

### ARTICLE XI General Provisions

- 11.1 <u>Contract Authorization</u>. Before executing this Contract, the School Governing Authority must pass a resolution in a properly noticed and held public meeting, authorizing execution of this Contract and authorizing one or more individuals to execute this Contract for and on behalf of the party, with full authority to bind the party.
- 11.2 <u>Termination and Cancellation of Contracts.</u> Except as otherwise permitted by this Contract, or by the Sponsor, contracts entered into by the School with third parties shall provide for the School's right to terminate upon failure of the School to remain in operation.
- 11.3 <u>General Acknowledgements.</u> The School specifically recognizes and acknowledges the following:
  - (a) The **Sponsor's** authority to assume operation of the **School** under R.C. 3314.073.
  - (b) The authority of public health and safety officials to inspect and order **School** facilities closed if not in compliance with health and safety laws and regulations in accordance with R.C. 3314.03(A)(22)(a).
  - (c) The authority of the Ohio Department of Education to suspend the operations of the **School** under R.C. 3314.072 due to the circumstances enumerated therein.
  - (d) The **Sponsor** is not liable for the acts or omissions, or the debts of the **School** pursuant to R.C. 3314.07(D) and 3314.08(J)(2), and any other applicable law limiting the liability of the **Sponsor**.
  - (e) The **Sponsor** may take steps to intervene in, correct, declare probationary status of, suspend, terminate or non-renew the status of the **School** as an Ohio Community School, and correct problems in the **School**'s performance.
  - (f) The Ohio Department of Education may take over sponsorship of the **School** in accordance with R.C. 3314.015(C).

- (g) The mandate of permanent closure under R.C. 3314.35, under the circumstances enumerated therein.
- (h) That the Auditor of the State of Ohio may deem the School "unauditable" and direct the stoppage of funding for the School, and, the Sponsor's right to terminate this Contract in such a situation.
- (i) That the Ohio Department of Education has set out its guidance for closing procedures (attached as a modified supplement to Attachment 3.5.3) and that the School hereby authorizes Sponsor to take all such action required by the ODE therein if the School administration or Board itself does not do so in a manner acceptable to the Sponsor or the ODE.
- (j) The Sponsor and its officers, directors, governing board, administrator, consultants, and employees are granted civil immunity for any action authorized under Chapter 3314, Ohio Community School Law, or the sponsorship contract, that is taken to fulfill the Sponsor's responsibility to oversee and monitor a community school, in accordance with R.C. 3314.07 or otherwise, and, the School hereby indemnifies, defends, and shall hold the Sponsor harmless from all such actions.
- 11.4 **Dispute Resolution.** Sponsor and School agree to informal mediation of any dispute not otherwise governed by mandatory administrative procedures pursuant to this Contract or the law. Such mediation shall be non-binding and the parties, if failing to agree on one mediator, shall obtain a list of three mediators from the Toledo Bar Association and each eliminate one, using the one mediator left after eliminations. All mediation will take place in Lucas County and all costs of the mediator shall be split equally between the parties.
- 11.5 <u>Term.</u> This Contract shall be for a term of <u>ten</u> (10) years effective as of, or, commencing on July 1, 2012 and ending on June 30, 2022. However, the School agrees to close voluntarily and waives all rights of due process and all claims, losses, causes of action or damages of any kind against ESCLEW if the School is closed mandatorily by statute, by another governmental agency or by operation of law. Pathway School of Discovery By KAK. ESCLEW By initials initials
- 11.6 <u>Renewal</u>. Renewal is subject to the Sponsor's determination that the School has satisfactorily complied with the applicable laws and this Contract, and that the School's progress in meeting the academic goals stated in this Contract is satisfactory.

### 11.7 Non-renewal of this Contract.

- (a) The **Sponsor** may choose not to renew this Contract at its Expiration Date for any of the following reasons:
  - (1) Failure to meet student performance requirements stated in this Contract;
  - (2) Failure to meet generally accepted standards fiscal management;

- (3) Violation of any provision of this Contract or applicable state or federal law;
- (4) Other good cause.

By February 1 of the year in which the **Sponsor** intends to not renew the Contract, the **Sponsor** shall notify the **School** of the proposed action in writing. The notice shall include the reasons for the proposed action in detail, the effective date of the non-renewal, and a statement that the **School** may, within fourteen (14) days of receiving the notice, request, in writing, an informal hearing before the **Sponsor**. The informal hearing shall be held within fourteen (14) days of the receipt of a request for the hearing. Within fourteen (14) days of the informal hearing, the **Sponsor** shall issue a written decision either affirming or rescinding the decision to not renew this Contract.

- (b) If the **School** does not intend to renew this Contract with the **Sponsor**, the **School** shall notify the **Sponsor** in writing of that fact at least one hundred eighty (180) days prior to the expiration of this Contract. In such a case, the **School** may enter into a Contract with a new sponsor in accordance with R.C. 3314.03, upon the expiration of this Contract, or, at the sole discretion of the **Sponsor**, by an assignment of this Contract before its expiration date.
- 11.8 **Probation.** The Sponsor may, in lieu of suspension or termination, declare in writing that the School is in a probationary status, after consulting with the School Governing Authority or authorized parties thereof, and specifying the conditions that warrant probation, and, after receiving the School Governing Authority's written assurances (satisfactory to Sponsor) of the actions and time frames necessary to remedy those conditions. Such probationary status shall not extend beyond the then current school year. Sponsor may proceed to suspension, termination or take-over of operations if the Sponsor finds at any time, that the Governing Authority is no longer able or willing to remedy the conditions to the satisfaction of Sponsor. Sponsor may suspend in lieu of probation at any time pursuant to 11.9 below.
- 11.9 Intent to Suspend/Suspension. The Sponsor may suspend operations of the School for (1) failure to meet student performance requirements stated in this Contract or (2) failure to meet generally accepted standards of fiscal management, or (3) violation of any provision of this Contract or applicable state or federal law, or (4) other good cause, if Sponsor sends a written notice of intent to suspend explaining the reasons and provides the School Governing Authority with five business days to submit a remedy, and promptly reviews and reasonably disapproves the proposed remedy, or, if the Governing Authority fails to submit a remedy or fails to implement the remedy.

Once the **School** is suspended it must cease operations on the next business day, immediately send notice to all **School** employees and parents stating that the **School** is suspended and the reasons therefore, and the **School** again has an opportunity to submit a proposed remedy within five business days. At all times during suspension, the **School** remains subject to nonrenewal or termination proceedings in accordance with the law.

11.10 <u>Termination of the Contract</u>. The Sponsor may choose to terminate this Contract for any of the following reasons: (1) failure to meet student performance requirements stated in this Contract, (2) failure to meet generally accepted standards fiscal management, (3)

violation of any material provision of this Contract or applicable state or federal law, or (4) other good cause.

Additionally, if the Sponsor has suspended the operation of this Contract under R.C. 3314.072, the **Sponsor** may choose to terminate this Contract prior to its expiration.

At least ninety (90) days prior to the termination of this Contract, the **Sponsor** shall notify the **School** of the proposed action in writing. The notice shall include the reasons for the proposed action in detail, the effective date of the termination, and a statement that the School may, within fourteen (14) days of receiving the notice, request, in writing, an informal hearing before the **Sponsor**. The informal hearing shall be held within fourteen (14) days of the receipt of a request for the hearing. Within fourteen (14) days of the informal hearing, the **Sponsor** shall issue a written decision either affirming or rescinding the decision to terminate this Contract.

A decision of the **Sponsor** to terminate this Contract may be appealed to the state board of education within fourteen (14) days following receipt of the **Sponsor's** written decision to terminate the contract. The state board shall conduct a hearing and issue a decision within 60 days of the notice of appeal. The decision of the state board is final.

The termination of this Contract shall be effective upon the occurrence of the later of the following events:

- (a) The date of the notice of termination; or
- (b) If an informal hearing is requested and as a result of that hearing the Sponsor affirms its decision to terminate this Contract, the effective date of the termination specified in the notice, or if that decision is appealed to the state board and the state board affirms that decision, the date established in the resolution of the state board affirming the Sponsor's decision.

If the Contract is terminated, the **School** must close permanently at the end of the current school year or on a date specified in the notification of termination and the **School** shall not enter into a contract with any other sponsor."

11.11 **Good Faith Deposit.** Immediately upon any Notice of Intent to Suspend sent to the **School** by the **Sponsor**, or, immediately upon a closure notice or suspension from any governmental or administrative agency, or a vote of closure by the School, the **School** must submit to **Sponsor**, a good faith deposit of \$15,000 to cover costs of any legal or other professional fees which may be required or desirable to facilitate matters including, but not limited to, notices to parents, transfer of files, change of locks, fees, securing assets, segregating or selling assets, returning assets, or other professional or non-professional fees or costs incurred by Sponsor which are in any way associated with termination and closure of the **School**, in case the **School** is actually suspended, closed or terminated, or in case the **School** fails to perform some or all of its responsibilities upon cessation of operations for any reason. The good faith deposit is to be returned to the **School** or sent to the appropriate creditor without interest, if all costs, expenses and

obligations paid by the **Sponsor** to comply with the **School's** responsibilities and Attachment 3.5.3., at the discretion of the Sponsor, are authorized to be and are actually covered by remaining **School** funds, if any.

- 11.12 **Failure to Open/Permanent Closure**. If the **School** fails to open for operation within eight (8) month(s) after the execution hereof, or, if the **School** permanently closes prior to the Expiration Date hereof, this Contract shall become void upon notice by Sponsor to **School**, subject only to the survival of Article X, Section 10.2 of this Contract.
- 11.13 <u>Compliance With Requests of Sponsor.</u> The School shall timely comply with all reasonable requests of the Sponsor, and allow the Sponsor to monitor the School operations. Failure to do so is grounds for suspension and termination or non renewal of this Contract. Timeliness is defined as compliance with the express provisions of this Contract, an answer in writing within five (5) business days (unless a shorter time is otherwise required pursuant to this Contract), and adequate assurances of cure or actual cure within a period of time acceptable to Sponsor.
- 11.14 <u>Headings</u>. Headings are for the convenience of the parties only. Headings have no substantive meaning.
- 11.15 <u>Assignments/No Third Party Beneficiaries</u>. This Contract and any part herefor shall not be assigned or delegated without the express written approval of **Sponsor**. This Contract shall inure to the benefit of, and shall be binding upon, the School, the Sponsor and their respective permitted successors or assigns; subject however to the following specific provision hereof: This Contract shall not inure to the benefit of anyone other than as provided for in the immediately preceding sentence and no third person or party shall have any rights hereunder, whether by agency or as a third party beneficiary or otherwise.
- 11.16 <u>Notice</u>. Any notice to one party by the other shall be in writing and effective upon receipt, and, may be satisfied by personal delivery or by any other means by which receipt can be documented, to; in the case of the **Sponsor**, the Superintendent; or, in the case of the **School Governing Authority**, the Chief Administrative Officer or a member of the **School Governing Authority**, or to the attorney for the **School Governing Authority**, at the last known business address of the **Sponsor**, and the last known business or home address of the **School** and/or its administrator or any Board member.

Should the **School** be abandoned by or not have in place, an administrator or an authorized Director of the Board, the **Sponsor** may give notice to the Ohio Department of Education.

- 11.17 <u>Severability</u>. Should any term, clause or provision of this contract be deemed invalid or unenforceable by a court of competent jurisdiction, all remaining terms, clauses or provisions shall remain valid and enforceable and in full force and effect, and the invalid or unenforceable provision shall be stricken or replaced with a provision as near as possible to the original intent.
- 11.18 <u>Changes or Modifications</u>. This Contract constitutes the entire agreement among the parties and any changes or modifications of this Contract shall be made and agreed to in

writing, authorized and executed by both parties. Notifications required by this Contract shall not be considered changes or modifications of this Contract.

11.19 <u>Attachments</u>. All <u>Attachments (1.5-11.19</u>) to this Contract are attached hereto and incorporated by reference into the Contract. Resolutions of the Sponsor Governing Board and School Governing Authority approving this Contract shall also be attached to this Contract.

Educational Service Center of Lake Erie West



(Title) with full authority to execute this Contract for and on behalf of **Sponsor** and will full authority to bind **Sponsor**. /

Date: \_\_\_\_

School Governing Authority of

Kwin a Kobie By: (Name)

Its: Board President (Title)

with full authority to execute this Contract for and on behalf of the School Governing Authority and with full authority to bind the School Governing Authority.

Date: 06/21/12

### **ATTACHMENT 1.5**

Corporate Documents, Certificate of Incorporation, Articles of Incorporation, Statutory Agent, Employer ID No., Code of Regulations (including the requirements of Article II section 2.5 of this Contract), Tax-Exempt Status Determination Letter (if any)), Organizational Chart of School and Mission Statement



DATE: 02/05/2001

DOCUMENT ID 200103303242

DESCRIPTION DOMESTIC ARTICLES/NON-PROFIT (ARN)

FILING 25.00 EXPED

10.00

PENALTY

·.00

COPY .00

CERT

.00

Receipt

This is not a bill. Please do not remit payment.

CORPORATE SERVICES 50 W. BROAD STREET COLUMBUS, OH 43215

# STATE OF OHIO

### Ohio Secretary of State, J. Kenneth Blackwell

### 1206071

It is hereby certified that the Secretary of State of Ohio has custody of the business records for

### EAST DAYTON SCHOOL OF SCIENCE AND DISCOVERY

and, that said business records show the filing and recording of:

Document(s)

DOMESTIC ARTICLES/NON-PROFIT

Document No(s): 200103303242



United States of America State of Ohio Office of the Secretary of State Witness my hand and the seal of the Secretary of State at Columbus, Ohio this 26th day of January, A.D. 2001.

Curreth Bachmell

Ohio Secretary of State

Doc ID --> 200103303242



### ARTICLES OF INCORPORATION OF THE

### EAST DAYTON SCHOOL OF SCIENCE AND DISCOVERY

The undersigned, desiring to form a nonprofit corporation pursuant to Chapter 1702 of the Ohio Revised Code, does hereby certify that:

FIRST: The name of the corporation shall be the EAST DAYTON SCHOOL OF SCIENCE AND DISCOVERY.

<u>SECOND</u>: The place in Ohio where the principal office of the corporation is to be located is Dayton, Montgomery County, Ohio.

THIRD: The corporation is organized exclusively for educational purposes within the meaning of Section 501(c)(3) of the Internal Revenue Code of 1986, as amended, or the corresponding provision of any future United States Internal Revenue Law (the "Code") to operate as a school in the State of Ohio. This corporation shall not engage in activities which are not in furtherance of the educational purposes set forth in this Article THIRD.

#### FOURTH: The following restrictions shall apply to the corporation:

(a) No part of the assets or of the net earnings of the corporation shall inure to the benefit of any member, trustee, or officer of the corporation or any private individual (except that reasonable compensation may be paid for services rendered to or for the corporation affecting one or more of its purposes). In the event of the liquidation or dissolution of the corporation, whether voluntary or involuntary, no member, trustee or officer of the corporation, or any private individual, shall be entitled to any distribution or division of the remaining assets or their proceeds.

(b) No substantial part of the activities of the corporation shall be the carrying on of propaganda or otherwise attempting to influence legislation, nor shall the corporation participate or intervene in (including the publication and distribution of statements) any political campaign on behalf of any candidate for public office.

(c) Notwithstanding any other provision in these Articles, the corporation shall not conduct or carry on any activities not permitted to be conducted or carried on by an organization exempt under Code Section 501(c)(3), or by an organization, contributions to which are deductible under Code Section 170(c)(2), or corresponding provisions of any subsequent federal tax laws. 2001 JAN 25

PH 4:

FIFTH: The names and addresses of the persons who are appointed to act in the capacity of the Initial Trustees of this corporation until the selection of their successors are as follows:

| Name               | - <u>Address</u>                               |
|--------------------|--|
| John Baker         | 1190 Beryl Trail, Dayton, Ohio 45459           |
| Greg Sample        | 300 W_First, Room 121, Dayton, Ohio 45402      |
| Bryan Bucklew      | 1 Chamber Plaza, Suite 200, Dayton, Ohio 45402 |
| Bob A. Rosenburger | Courthouse Plaza NE, Dayton, Ohio 45463        |
| Ron Adler          | 1808 Sulky Trail, Miamisburg, Ohio 45342       |

SIXTH: In the event of the dissolution of the corporation, the corporation shall, after paying or making provision for the payment of all liabilities of the corporation, dispose of all of the assets of the corporation exclusively for the purposes set forth in Article THIRD of these Articles of Incorporation.

2

IN WITNESS WHEREOF, I have hereunto subscribed my name on January 24, 2001.

M Amy J. Borm Incorporator

Page 3

### ORIGINAL APPOINTMENT OF STATUTORY AGENT

#### OF THE

### EAST DAYTON SCHOOL OF SCIENCE AND DISCOVERY

The undersigned, being the incorporator of the EAST DAYTON SCHOOL OF SCIENCE AND DISCOVERY, hereby appoints Amy J. Borman, a natural person and resident in the state in which such corporation has its principal office, upon whom any process, notice or demand required or permitted by statute to be served upon the corporation may be served. Her complete address is Amy J. Borman, Esq., Udell & Borman, Ltd., 5738 Main Street, Sylvania, Ohio 43560.

Incorporator Amy J. Borman

The undersigned hereby accepts appointment as statutory agent for the EAST DAYTON SCHOOL OF SCIENCE AND DISCOVERY.

Amy J. Borm

Dated: 1-24-01

U:\aberman\Dayton East\Articles of Ino-SA.wpd

Page 4

Doc ID -->

DATE:



C

.00

PENALTY CESCRIPTION DOMESTIC/AMENDMENT TO ARTICLES (AMD) CERT DOCUMENT ID 200212200650 PED. 05/02/2002 50.DD .00 Receipt

This is not a bill. Please do not remit payment.

**C T CORPORATION SYSTEM** JAMES TANKS 17 S HIGH ST COLUMBUS, OH 43215



Page 1

| >              | 200212200650   |                  | · · · |   |
|----------------|--|------------------|-------|---|
| Ef             | ASTMAN AND SMITH Fax: 419-247-1777 May 1 2002 14:49 P.02   |                  |       |   |
|                |  |                  |       |   |
|                |  |                  |       |   |
|                |  |                  |       |   |
| ß              | The I Kenneth Blackwell Expedite this Form; subto  | ]                | •     |   |
| E.             |  |                  |       |   |
|                |  | WHO DEES         |       |   |
| 18 Car         | Columbus, CH 432   | 218              |       |   |
|                | Requires an additional fee of M  | 30***            |       |   |
|                | v.state.ch.us/sos<br>ii: bussev@sos.state.ch.us<br>Columbus, OH 432  | 216              |       |   |
| 8-mai          |  |                  |       |   |
|                | Certificate of Amendment by  | ·                |       |   |
|                | Shareholders or Members<br>(Domestic)  |                  |       |   |
|                | Filing Fee \$50.00   |                  |       | • |
|                |  |                  |       | • |
|                |  | ,                |       |   |
| ( <u>CHI</u>   | ECK ONLY ONE (1) BOX   | <u> </u>         |       |   |
|                | Amendment Amendment Amendment  | 1                |       |   |
| . L            | (125-AMDS) (125-AMDS) (126-AMAR) (128-AMD)   |                  |       |   |
| là             | picto the general information in this section for the box checked above.   | ]                |       |   |
| Com            |  |                  |       | - |
| Nami           | e.of Corporation East Dayton School of Science and Discovery   |                  |       |   |
| Char           | ter Number 1206071   |                  |       |   |
| -              |  |                  | •     |   |
| <u>└</u> ┘.    | Please check if additional provisions attached.  |                  | • .   |   |
| The :          | above named Ohlo corporation, does hereby certify that:  | 1                |       |   |
| 10.            | A meeting of the Shareholders I members was duly called and held on June 8, 2001   |                  |       |   |
|                | (QILECTOIS) (Qap)  |                  |       |   |
| atwł           | hich meeting a quorum was present in person or by proxy, based upon the quorum present, an affirmative was cast which entitled them to exercise 100 % as the voling power of the corporation.  | 2 2              |       |   |
|                |  | SECR             |       |   |
|                | in a writing signed by all of the Ishareholders I members who would be entitled to the notice of a   | 12               |       |   |
| ہ لی ا<br>meet | The writing signed by all of the lighterationalers in memory would be and to be included in the included includ | ·, · )           |       |   |
| L              |  |                  | ,     |   |
|                |  |                  |       |   |
|                | iso applies if amended box is checked.   |                  |       |   |
| Rec            | olved, that the following amended articles of incorporations be and the same are hereby adopted to supercede   | л <sup>г</sup> . |       |   |
| and            | take the place of the existing articles of incorporation and all amendments thereto.   |                  |       |   |
| - i            |  |                  |       |   |
| J              |  |                  |       |   |

Page 1 of 3

Last Revised: Jan. 2002

• • •

. . . . ...

OHOG - 030802 C T Byren Online

541

| 20021 | 12200650 |
|-------|----------|
|       |          |

•

.

 $\mathbf{h}$ 

|        | following information must be comp<br>adment box is chocked, complete the |                                       | :d.                |
|--------|---|---------------------------------------|--------------------|
| FIRST: | The name of the corporation is:   | East Dayton School of Discove         |                    |
| Second | ; The place in the State of Ohio wh                                       | ere ils principal office is located i | is in the Cily of: |
|        | (city, village or lownship)   |                                       | county)            |
| THIRD: | The purposes of the corporation a   | re as follows:                        |                    |
|        |   |                                       |                    |
|        |   |                                       |                    |
|        |   |                                       |                    |
| 1      |   |                                       |                    |
|        | 1   |                                       |                    |

Must be authenticated by an authorized representative Authorized Représentative Any J. Boznan, Esq. L Authorized Representative

-

.

Authorized Representative

| May | 1,2002 |  |
|-----|--------|--|
| •   | Date   |  |

Date

. Date

....... .

| 541                                | Paga 2 of 3 | Last Revised: Jan. 2002 |
|------------------------------------|-------------|-------------------------|
| CH009 - 02/01/02 C T System Orlizz |             |                         |
|                                    |             |                         |
|                                    |             |                         |
|                                    |             |                         |

## \*200312102378\*

| DATE       |
|------------|
| )5/02/2003 |

()

DOCUMENT ID DESCRIPTION 200312102378 DOMESTIC/AMENDMENT TO ARTICLES (AMD) FILING 50.00 PENALTY .00

EXPED

00

)

CERT

00

COPY .00

Receipt

This is not a bill. Please do not remit payment.

EASTMAN & SMITH LTD. ATTN: JULIE A. MIERZEJEWSKI ONE SEAGATE, 24TH FLOOR TOLEDO, OH 43699



|     |     | e c                                     | Т,          |      |    |
|-----|-----|---|-------------|------|----|
|     |     | 3                                       | <u>بلار</u> |      |    |
|     | 15/ | ~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~ |             |      |    |
|     | Ęţ  |   |             |      | 5  |
| :   | V   |   |             | 1.57 | ¥. |
| • * |     | -                                       | *           | 2    |    |

Presenbed by J. Kenneth Blackwell .'

Ohio Secretary of Stam Central Ohio: (614) 466-3910 Toil Free: 1-877-SOS-FILE (1-877-767-3453)

| WWW A   | <u> Alean, Us/Sos</u> | •         |
|---------|-----------------------|-----------|
| s-mail: | busserv@sce.st        | ate.oh.ue |

| Erned | te this Form; (suesone)  |
|-------|--|
|       | The street of th |
|       |  |
| Over  | PO Box 1390  |
|       | Columbus, OH 43218   |
| HE RA | lifes an additional fee of \$100 🚥 👘   |
| ON:   | PO Bex 1028.   |
| C 49  | Columbus, OH 43216   |
|       | · · · · · · · · · · · · · · · · · · ·  |

# Certificate of Amendment by Sharehoiders or Members (Domestic) Filing Fee \$50,00

| 1) Domesilo for Profit<br>Amended<br>(122-AMAP) | LI Amendment<br>(1254AMDS)  | (2) Domestic Non-Profit<br>Amended<br>(125-AMAN)           | Amendment<br>(128-AMD)                | \         |
|---|---|--|---------------------------------------|-----------|
| Omplete the dependent inte-                     |   |  | · · · ·                               |           |
|   | mation in this section for the box she  | eked abova.  |                                       | •         |
| ame of Corporation                              | East Devion School of Discover  | V.   | ·                                     |           |
| harter Number                                   | 1208071   | •  | · · · · · · · · · · · · · · · · · · · |           |
| ame of Officer                                  | Biyan Bucklew   |  | •                                     |           |
| tle   | President   |  |                                       | ,         |
| Please check if sociational                     | provisions allected.  | · · · · · ·  | •                                     | ۰,        |
|   | rporation, does hereby certify that:  | •  |                                       | •         |
| A meeting of the                                |   | directors (non-pr  | afit umencied articles anlys          |           |
| I members was duly call                         |   | 17, 2003   |                                       | •         |
| In a writing signed by a                        | I was present in person or by proxy,<br>d them to exercise9<br>Il of theshareholders dire | based upon the quorum pre-<br>% as the voting power of the | corporation.                          |           |
|   |   | •<br>•   |                                       | · · · · · |
| · · ·   |   |  |                                       |           |
| use applies if amanded in                       | ex p checked  |  | ·                                     |           |

Last Revised: May 2002

4 Charles and the second second

|  |   | · · · · · · ·                         |
|--|---|---------------------------------------|
|  |   |                                       |
| All of the following informer  | on must be completed if an amended box is checkn<br>ked, complete the areas that apply. | id.                                   |
|  | • •   |                                       |
|  |   |                                       |
| SECOND: The place in the   | State of Ohio where its principal office is located i                                   | e in the City of:                     |
|  |   | • • •                                 |
| (oky, völage or towns  | · · · · · · · · · · · · · · · · · · ·   | uniy)                                 |
| THIRD: The purposes of   | he corporation are as follows:  |                                       |
|  |   |                                       |
|  |   | · · · ·                               |
|  |   |                                       |
|  |   |                                       |
|  |   | and a second second                   |
|  | •   | •                                     |
|  |   |                                       |
| FOURTH: The number of all  | ares which the corporation is authorized to have o                                      |                                       |
|  | (Dese not apply to box (2))   |                                       |
|  |   | · · · · · · · · · · · · · · · · · · · |
| · ·  | •   |                                       |
| · · · · · ·  |   |                                       |
| )  |   |                                       |
|  |   |                                       |
| REQUIRED<br>Must be authenticated  | Bar S. Breen  | 4-25-2003                             |
| Must be suthenticated  | Bon S. Bulen<br>Authorized Representative   | 4-25-2003<br>Date                     |
| REQUIRED<br>Must be authenticated<br>(signed) by an authorized<br>representative<br>(See Instructions) | Ban S. Banler<br>Authorized Representative  |                                       |
| Must be authenticated<br>(signed) by an authorized<br>representative                                   | Bar S. Buler<br>Authorized Representative   |                                       |
| Must be authenticated<br>(signed) by an authorized<br>representative                                   | Authorized Representative   | Date                                  |
| Must be authenticated<br>(signed) by an authorized<br>representative                                   |   |                                       |
| Must be authenticated<br>(signed) by an authorized<br>representative                                   |   | Date                                  |
| Must be authenticated<br>(signed) by an authorized<br>representative                                   |   | Date                                  |
| Must be authenticated<br>(signed) by an authorized<br>representative                                   | Authorized Representative   | Date                                  |
| Must be authenticated<br>(signed) by an authorized<br>representative                                   | Authorized Representative   | Date                                  |
| Must be authenticated<br>(signed) by an authorized<br>representative                                   | Authorized Representative   | Date                                  |
| Must be authenticated<br>(signed) by an authorized<br>representative                                   | Authorized Representative   | Date                                  |
| Must be authenticated<br>(signed) by an authorized<br>representative                                   | Authorized Representative   | Date                                  |
| Must be authenticated<br>(signed) by an authorized<br>representative                                   | Authorized Representative   | Date                                  |
| Must be authenticated<br>(signed) by an authorized<br>representative                                   | Authorized Representative   | Date                                  |
| Must be authenticated<br>(signed) by an authorized<br>representative                                   | Authorized Representative   | Date                                  |
| Must be authenticated<br>(signed) by an authorized<br>representative<br>(See instructions)             | Authorized Representative   | Date                                  |
| Must be authenticated<br>(signed) by an authorized<br>representative                                   | Authorized Representative   | Date                                  |
| Must be authenticated<br>(signed) by an authorized<br>representative<br>(See instructions)             | Authorized Representative   | Date                                  |

|         |  | -  |                              |  | TON CHAMBER OF COMMI                                    | -                |                                      | P. 02/02 F-345        |
|---------|--|--|------------------------------|--|---|------------------|--------------------------------------|-----------------------|
|         |  |  | <b>W</b> <sup>1</sup>        |  | 31-6311   | •                | ATTN: DA                             | Ni                    |
|         | m <b>SS-4</b>  | Applic   | ation for E                  | mploy  | er Identificatio  | n Numb           | per l                                | <u></u>               |
|         | v. April 2000]   | For use by employers, corporations, partnerships, trusts, estates, churches,<br>government agencies, certain individuals, and others. See instructions.) |                              |  |   | <u>(-3633)</u>   |                                      |                       |
| Inte    | Adment of the Tressury   |  |                              | D & Copy   | for your records.                                       | NI2717C(1092     | OME N                                | 0. 1545-0003          |
|         | 1 Name of applica<br>EAST DAVID  | nt (icgai name).<br>N. Schoor.   | 1508 (/IStructions)          |  |   |                  | • <u>•</u>                           |                       |
| clearly | 2 Trade name of b  | EAST DAYTON SCHOOL OF: DISCOVERY 2 Trade name of business (if dilferant from name on line 1) 3 Executor, trustee, Tears of anno                          |                              |  |   |                  |                                      |                       |
|         |  |  |                              |  |   |                  |                                      |                       |
| print   | A Mailing address  | nddross (street address) (room, apt., or state no.)  |                              | 5a Business address fil different from address on lines to the |   |                  |                                      |                       |
| PO 4    | 4b City, state, and ZIP code   |  |                              | 989 Spaulding Ave., Sg   |   |                  |                                      |                       |
| adici   | Dayton, Ohio 454(12<br>5 County and state where principal business is located  |  |                              |  | Sb City, state, and ZIP code<br>Grand Rapids, 1/1 49546 |                  |                                      |                       |
| Reaso   | A County and state   | whata principa   | t business is locate         | d  |   | <u>191 [31</u>   | 49040                                |                       |
| Ŧ       | 7 Name of principal  | officet, deneral o   | STOCK GRANDE DWA             | T. Of Ingle  | x-SSN or ITIN may be ro                                 |                  | ·····                                |                       |
|         |  |  |                              | -,   |   | quireri (sec in: | structions}                          |                       |
| Ra,     |  | aniy one box.)   | (see instructions)           |  | <b>***</b> *********************************            |                  |                                      |                       |
|         | Caution: Ir applicant  | is a limited liab  | illy company, see t          | he instruct  | ions for line Ba,                                       |                  |                                      |                       |
|         | 🔲 Sole proprietor (S   | 5N)  | 1                            | [] <b>5</b> 0  | tate (SSN of docedent)                                  | 1                | :                                    |                       |
|         |  |  | rsonal service corp.         | <u> </u>   | in administrator (SSN)                                  |                  |                                      |                       |
|         |  |  | tonal Guard                  | - 🗆 Qi   | ter corporation (spacify) >                             | -                |                                      | ·                     |
|         |  |  |                              |  |   |                  |                                      |                       |
|         | Citter nonprofit or<br>Other nonprofit or  | janization (spoc   | aly) 🕨Cummur                 | uity"  | deral government/military                               | (<br>പ്രദേഷം)    |                                      |                       |
|         | If a corporation, nam  | e the state or fr  |                              | 00000  | 1   |                  | _                                    |                       |
|         | in appreciate where incorporated   |  |                              |  |   |                  |                                      |                       |
| 9       | Reason for applying (C   | neck only one b  | INX.) (see instructions      | ) 🗌 Ba   | nkilig purpose (specify pi                              | urpase) 🕨 _      |                                      |                       |
|         | Community  |  | -                            | 말다   | inged type of organizatio                               | n ispecity ne    | :w туре) 🕨                           |                       |
|         | Hired employees #  | Chack the box  | vid see line 12,)            |  | chased going business<br>alod a must (specify type      |                  |                                      |                       |
| 0       | Dato business started  | of acounted inv  |                              | liter and  |   | _L Other (       | spacify) ►                           |                       |
|         | 2 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4  | ZUUI   |                              |  |   | g month of a     | ccounting year (see                  | instructions)         |
| 2       | First date weges or an<br>first be paid to nonres  | nuites ware pa<br>Ident alien, (mo   | it or will be paid (m        | ionith, day  | year). Note: If applicant                               | 'ls a withhok    | ting agont, enter dat                | a incoma will         |
| J       |  |  |                              |  | e: If the applicant does no<br>fructions)               |                  | ultural Agricultural                 | Household             |
|         | the second second in the second  | naonétiquel 📥  | £07                          | -) (200 875<br>17(1) 15-5 - 4                                  | ructions  | ► N/A            |                                      | L                     |
| 5       | is the principal busine  | is activity manu   | 120 Junio os                 |  |   |                  | Ves                                  | XX No                 |
| 5       | H "Yes," principal proc<br>To whom are most of<br>Public tratain   | HOL BOD INW ME   | Lervicos coldz               |  |   | · · · ·          |                                      |                       |
|         |  |  |                              |  |   | 🗌 Bus            | iness (wholesale)                    | XDE N/A               |
|         | and the second sec |  | i an antar littar            |  | t for this or any other bu                              |                  | • • • • • • Yee                      | XX No                 |
| "b.<br> | If you checked "Yes" (<br>Legal name >   | in line 172. givo  | opplicant's legal na         | me and tr  | ade name shown on prio                                  | r application.   | if different from line               | 1 or 2 above.         |
| c.      | Approximate date when fill   | n and reity and a  |                              |  | Is filed. Enter previous er                             | nployer ideni    | ification number II k<br>Yovious Ein | nown.                 |
|         |  |  | 1                            |  | dgt and bolief, it is three, correct, a                 |                  | 1                                    | _                     |
|         |  |  | - Automore Alial at the 1921 | er try knowle  | oge and bolids, it is true, comport, a                  |                  | inicial talephane matching ()        |                       |
| line :  | nd we please type or pr  | lat mtanak-t hu  | Duta_ 7- *                   | <b>4</b> –   |   |                  | 616) 222-                            | 1700<br>Ma area code) |
|         |  | rie Glooriy,) 🏲  | Bryan Buck                   | tew, I   | resident  |                  | 516 ) 8 -63                          |                       |
| ynaiu   |  | ->   | <u>. Burl</u>                | us   |   | Deta >           | 11-10-                               | 0)                    |
|         | - 10 aver   600.   | 0  | hata: Do not write           | helow this   | line. For official uso only                             |                  |                                      |                       |
| 185¢    | LANUAR I STORE   |  |                              |  | Class   |                  | asson for applying                   |                       |

### AMENDED AND RESTATED CODE OF REGULATIONS OF THE PATHWAY SCHOOL OF DISCOVERY

### ARTICLE I <u>PURPOSE</u>

Section 1. <u>Purpose</u>. The PATHWAY SCHOOL OF DISCOVERY (the "Corporation") is organized exclusively for educational purposes within the meaning of Section 501(c)(3) of the Internal Revenue Code of 1986, as amended, or the corresponding provision of any future Internal Revenue Law of the United States to operate as a school in the State of Ohio.

### ARTICLE II DIRECTORS

Section 1. <u>Number</u>. The number of Directors of the Corporation shall be at least five (5) and no more than nine (9) or such greater number as may be subsequently determined by the Directors, but in no case less than five (5).

Section 2. <u>Term.</u> Each Director will serve a three-year term, which expires on June 30<sup>th</sup> of the third year following the year of their election, and which may be renewed as many times as such Director is elected. Each Director shall hold office until that Director's term expires, or until his or her successor is elected, or until his or her earlier resignation, removal from office, or death. Provided however, of the initial Directors, two (2) served for three (3) years, two (2) served for two (2) years, and one (1) served for one (1) year, in order to create a staggered Board of Directors. The initial terms of the initial Directors were deemed to have commenced on the date the Articles of Incorporation were filed, which was on January 26, 2001.

Section 3. <u>Qualifications and Role of Directors</u>. The Directors shall be, in their capacity as Directors, the Governing Board of a public community school in Ohio upon the signing of a charter contract with a sponsor which creates the school. The Directors shall have a strong interest in the welfare of the Corporation and in education. Each Director should be willing and able to attend all meetings, both regular and special, and also be willing to accept special assignments and serve on committees.

Section 4. <u>Election of Directors</u>. Candidates for Director shall be nominated by the Board of Directors or a committee thereof.

Section 5. <u>Meetings</u>. The annual meeting of the Directors shall be held at such time and place as a majority of the Directors may determine and special meetings may be called at any time by the President or by any two (2) Directors. Provided however, once a charter is granted and a contract executed causing the Corporation to become a public school, meetings relating in any way to the business or operation of the public school must be open to the public and publicized or advertised as required by law.

Section 6. <u>Quorum and Voting</u>. The presence of a simple majority of the total number of Directors shall constitute a quorum for the transaction of business at all meetings of the Board of Directors. A vote of a simple majority of the Directors present at a meeting at which a quorum is present shall be required to effectuate action on all matters within the powers of the Board of Directors unless otherwise provided by law or herein. In addition to those Directors who are actually

present at a meeting, Directors shall for purposes of this section be deemed present and able to vote at such meeting if a conference telephone or similar communications equipment is used by means of which all persons participating in the meeting can communicate with each other at the same time. Provided however, once a charter is granted and a contract executed causing the Corporation to become a public school, the Directors must be physically present at a meeting in order to be counted as part of a quorum and to vote, for so long as required by law.

Section 7. <u>Notice and Waiver</u>. Unless waived, notice of each annual or special meeting communicating the day, hour and place, and the purpose or purposes thereof shall be given to each Director by the Secretary of the Corporation not more than sixty (60) days nor less than three (3) days before any such meeting. Notice of the time, place and purposes of any meeting may be waived in writing, either before or after the holding of such meeting, by any Director, which writing shall be filed with or entered upon the records of the meeting. The attendance of any Director at a meeting without protesting, prior to or at the commencement of the meeting, shall waive notice or lack of proper notice for that meeting. Nothing in this Section 7 shall alter, however, the duty of the Corporation to provide notice to the public of meetings, once a charter is granted and the corporation becomes a public school.

Section 8. <u>Action by Written Consent</u>. Any action required or permitted to be taken at any meeting of the Board of Directors or of any committee thereof may be taken without a meeting, if a written consent to such action is signed by all of the Board of Directors or of such committee, as the case may be, and such written consent is filed with the minutes of proceedings of the Board of Directors or committee. Such a written consent may be signed by facsimile signatures which shall be construed as originals, and/or on separate but identical documents which shall be construed as one original. Provided however, if the Corporation is operating as a public school pursuant to a fully executed charter contract, all action must be taken at open and public meetings and action by written consent shall not be allowed.

Section 9. <u>Committees of Directors</u>. The Board of Directors may create a committee or committees as the Directors may determine, the members of which committee or committees shall consist of not less than one (1) Director. A simple majority of the members of any such committee shall constitute a quorum, and the act of a simple majority of the votes cast at a meeting at which a quorum is present shall be the act of the committee. In every instance, however, the final action on all committee business shall be in the nature only of recommendations to the Board of Directors. Notwithstanding anything to the contrary in this Section 9 however, once the Corporation has signed a charter contract and becomes a public school, no committee nor any group of Directors, which consist of a majority of the Board of Directors, shall meet in a prearranged manner to discuss school business, without proper notice to the public of a regular or special meeting, and, only the votes of the Board of Directors shall be valid and binding.

Section 10. <u>Other Advisory Councils</u>. The Board of Directors may, at its discretion, also consider recommendations of associations, supporting organizations or advisory councils which are not part of the Board of Directors, such as those of parents or other pertinent groups.

Section 11. <u>Removal of Directors</u>. Any Director may be removed, with or without cause, at any time, by a majority vote of all the Directors.

Section 12. <u>Resignations and Vacancies</u>. Any Director may resign by tendering a written

resignation to the Board of Directors. The resignation shall be effective upon the receipt of the resignation by an officer of the Board or, if later, upon the date specified by the Director in his/her resignation. Vacancies in the Board of Directors shall be filled in accordance with Sections 4 and 6 of this Article II.

Section 13. <u>Powers of Directors</u>. The policies of the Corporation shall be directed by the Board of Directors in accordance with the law. Subject to the provisions of Ohio law in general, the Ohio Nonprofit Corporation Law, the Articles of Incorporation and this Code of Regulations of the Corporation, the Board of Directors shall do and perform every act and thing whatsoever which it shall deem necessary, expedient or advisable to carry out the purposes of the Corporation.

Section 14. <u>Honorary Directors</u>. Any individual, whether an emeritus Director or not, who has provided extraordinary service to the Corporation over a period of time, may be honored with the title "Honorary Director" at the discretion of the Board. Honorary Directors are elected for life, are not voting members of the Board and are permitted but not required to attend meetings.

### ARTICLE III OFFICERS

Section 1. <u>Number, Title and Election</u>. The officers of the Corporation shall consist of a President, Vice President, Secretary, and Treasurer, and may include such other officers and assistant officers as the Board of Directors shall deem advisable, each of whom shall be elected by the Board at the annual meeting of the Board. Officers shall hold office for a term of one year, or until their successors are elected and qualified, except in the event of their earlier death, resignation or removal. All officers, except the Treasurer, shall be elected from the Board of Directors of the Corporation. The Treasurer may be, but is not required to be, a member of the Board of Directors.

Section 2. <u>Vacancies</u>. A vacancy in any office because of death, resignation or removal of an officer shall be filled by the Board of Directors for the unexpired term of such office.

Section 3. <u>Resignation or Removal of Officers</u>. An officer of the Corporation may resign at any time by tendering his or her resignation in writing to the Board of Directors and such resignation may become effective immediately upon its delivery to the Board. An officer of the Corporation may be removed at any time, with or without cause, by the Board of Directors. The election or appointment of an officer for a term of office shall not be deemed to create contract rights.

Section 4. <u>President</u>. The President shall preside at all meetings of the Board and shall coordinate the activities directed by the Board of Directors and shall oversee the actions of the Corporation subject to the policies and goals established by the Board of Directors.

Section 5. <u>Vice President</u>. The Vice President shall perform the duties of the President when the President is absent, and all other duties as may be assigned by the Board of Directors.

Section 6. <u>Secretary</u>. The Secretary shall be responsible for providing notice of meetings to the Board of Directors where notice is required, and to the public for the matters concerning the public school, and shall keep or approve a record of the proceedings of the Board of Directors, and shall perform other duties as may be required by the Board of Directors.

Section 7. <u>Treasurer</u>. The Treasurer shall act as, or in conjunction with, the fiscal officer or fiscal agent of the Corporation and shall have custody of the cash, securities, and other assets of the Corporation. The Treasurer shall receive contributions, bequests, revenues, and other assets to which the Corporation is entitled and disburse funds as directed by the Board of Directors. The Treasurer shall maintain or approve appropriate books of account and supporting records and shall prepare or approve and file all returns and related reports required by federal and state statutes and regulations. However, notwithstanding the fact that the Corporation has named a person to serve as its Treasurer, the Board of Directors may contract with a third party to provide for part or all of the Treasurer's responsibilities, subject to approval by a majority of the Directors. The Board may require a bond in any amount, at its discretion or as directed by law, and the cost of the bond or bonds shall be paid for by the Corporation.

### ARTICLE IV

### INDEMNIFICATION OF DIRECTORS, OFFICERS, EMPLOYEES, AND AGENTS

Each person who at any time is or shall have been a Director, officer, employee or agent of the Corporation, or a Director member of the Governing Board of the school, and such person's heirs, executors and administrators, shall be indemnified by the Corporation, both during and after their association with the Corporation, for those acts or omissions concerning the Corporation, in accordance with and to the full extent permitted by the Nonprofit Corporation Law of the State of Ohio as in effect at the time of the adoption of these Regulations or as amended from time to time thereafter. The foregoing right of indemnification shall not be deemed exclusive of other rights to which any Director, officer, employee, agent or other person may be entitled in any capacity as a matter of law or under any regulations, agreement, vote of Directors, or otherwise. As authorized by the Board of Directors, the Corporation may purchase and maintain insurance against liability on behalf of any such person to the full extent permitted by law in effect at the time of the adoption of these Regulations or as changed from time to time.

### ARTICLE V CONTRACTS BETWEEN CORPORATION AND RELATED PERSONS

To the greatest extent allowed by Ohio law, any contract or other transaction between this Corporation and one or more of its Directors, or between this Corporation and any entity of which one or more of this Corporation's Directors are interested, whether such Director is a member of the Governing Board of the school or not, shall be valid for all purposes, notwithstanding the presence of such Director at the meeting at which the Board of Directors of the Corporation acts upon, or in reference to, such contract or transaction, and notwithstanding the participation of the Director in such action, if the fact of such interest shall be disclosed or known to the Board of Directors, and the Board of Directors nevertheless, authorize, approve or ratify such contract or transaction by a vote of a majority of the Directors present. Unless Ohio law otherwise prohibits or permits, the interested Director may be counted in determining whether a quorum is present, but may not be counted in voting upon the matter or in calculating the majority of such quorum necessary to carry such vote. This Article shall not be construed to invalidate any contract or other transaction which would otherwise be valid under the common and statutory law applicable thereto.

### ARTICLE VI BOOKS AND RECORDS

The Corporation shall keep correct and complete books, records and minutes of the Board of Directors' meetings, and, during the time when the Corporation is functioning as a public school by execution of a charter contract, such books and records shall be public records. The Secretary of the Corporation shall keep an accurate list of the names and addresses of the Board of Directors.

### ARTICLE VII **AMENDMENTS**

The Articles of Incorporation and Code of Regulations shall be adopted and amended by a majority of all of the Board of Directors.

H:\HOME\PJDOWNEY\AJB\National Heritage Academies\Pathway\Amended and Restated Code of Regulations 05-05-06.doc

These Code of Regulations were adopted on the  $\frac{9^{W}}{day}$  of  $\frac{1000}{2000}$ .

## Amended and Restated Code of Regulations

Per Amy Borman, Eastman & Smith, it is recommended that the board review and adopt the Amended and Restated Code of Regulations. The Code of Regulations were changed in response to:

- changes in law,
- specific board member requests for clarification
- name changes, if applicable

The following is an incomplete list of the changes Eastman & Smith made to the Codes of Regulations of the schools, but, please be aware that not all of these changes apply to each of the NHA schools.

Article II, Section 1 – Revised the language regarding number of minimum directors from 3 to 5.

Article II, Section 2 - Revised the language regarding the initial terms of the initial Directors because some of these terms have expired.

Article II, Section 6 - Added the following language, "unless otherwise provided by law or herein", to end of the second sentence to eliminate a potential ambiguity.

Article II, Section 8 - Revised the last sentence in the section to clarify meaning and eliminate excess verbiage.

Article II, Section 11 - Added the clause, "by a majority vote of all the Directors", to ensure that the removal of a Director is agreed upon by a majority (at least three) of the Board.

Article II, Section 12 - Revised the provision regarding the effective date of the resignation of a Director to provide that the resignation is effective immediately unless the Director expressly provides that his resignation is not effective until a later date.

Article III, Section 1 - Revised first sentence to clarify that the Corporation will have the offices of President, Vice President, Secretary, and Treasurer and indicate that the Board may create other offices. Stated the concepts contained in the last sentence of the Section in the current Code in two sentences to clarify the intent.

Article VI - Revised the last sentence to provide that it is the duty of the Secretary of the Corporation to keep an accurate list of the names and addresses of the Board of Directors because this is a function that is more properly performed by the Secretary.

Article VII - Revised this provision to clarify the intent to require that a majority of the entire Board (at least three Directors) must approve any amendment of the Articles of Incorporation or the Code of Regulations.



### Organizational Chart of the School

36

### **1.5 Mission Statement**

Working in partnership with parents and the community, the Pathway School of Discovery mission is to become one of the finest K-8 schools in the country. We offer a challenging, character-based education through a rigorous curriculum with high academic and social expectations. Our school prioritizes the academic and instructional time so each student reads, computes and writes at or above grade level. We expect our student to master basic skills and realize their full academic potential in preparation for higher education and adulthood.

### ATTACHMENT 1.7 (b)

Description of evaluation method to be used by Sponsor if School is declared in need of continuous improvement or higher, offers one or more grade levels for which an achievement test is prescribed under Ohio law, and administers reading and mathematics assessments under Ohio law, <u>if</u> Sponsor so elects under R.C. 3314.35 or .36.

If not applicable, write "N/A" here \_\_\_\_N/A\_\_\_\_\_.
# **ATTACHMENT 2.1**

School Governing Authority Board of Directors,

Names, Addresses, Telephone Nos. (work, cell and home), Electronic Mail Addresses Background Checks and Description of Selection and Removal of Board of Directors Board Meeting Schedule and Location of Meetings

# Pathway School of Discovery Roster

| Bartlett                                     | Paige                                 | Comm Preference                                    |
|--|---------------------------------------|--|
| Phone #<br>Mailing Address<br>Street Address | 937-293-9592<br>26 East Thruston Blvd | PositionSecretaryTerm3yrs, 1/09ExpirationJune 2012 |
| City<br>State<br>Zip                         | Dayton<br>OH<br>45409                 |  |
| Business Phone<br>Home Phone                 | 0<br>937-293-9592                     | Business Fax 937-293-9592<br>Home Fax              |
| Mobile Phone                                 | 937-286-9293                          | Email Addresses paigebar@msn.com                   |

| Garretson  | Susan<br>495 Burnside Drive<br>Tipp City<br>OH<br>45371 | Comm Preference                                   |  |
|--|---|---|--|
| Phone #<br>Mailing Address<br>Street Address<br>City<br>State<br>Zip |   | PositionDirectorTerm3 yr, 1/10ExpirationJune 2013 |  |
| Business Phone<br>Home Phone   | 937-506-8288  | Business Fax<br>Home Fax                          |  |
| Mobile Phone   | 937-974-1246  | Email Addresses sgarretson@moundstreet.k12.oh.us  |  |

| Martin   | Duane   | Comm Preference Cell Phone                               |
|--|---|--|
| Phone #<br>Mailing Address<br>Street Address<br>City | 937-499-9718<br>Premier Health Partners<br>40 W. 4th Street, Suite 1800<br>Dayton | PositionVice PresidentTerm3 yrs, 1/09ExpirationJune 2012 |
| State<br>Zip   | ОН́<br>45402  |  |
| Business Phone<br>Home Phone                         | 937-499-9718<br>937-264-9140  | Business Fax 937-208-9726<br>Home Fax                    |
| Mobile Phone   | 937-361-0519  | Email Addresses kdmartin@shp-dayton.org                  |

NEXT A REPORT OF COMPANY AND A COMPANY AND

| Robie  | Kevin                                | Comm Preference Cell/Email                                    |
|--|--------------------------------------|---|
| Phone #<br>Mailing Address<br>Street Address | 937-431-0556<br>264 Timberleaf Drive | PositionPresident/TreasurerTerm3 yrs, 1/11ExpirationJune 2014 |
| City<br>State<br>Zip                         | Dayton<br>OH<br>45430                |   |
| Business Phone<br>Home Phone                 | 937-610-4010<br>937-431-0556         | Business Fax 937-222-4329<br>Home Fax                         |
| Mobile Phone                                 | 937-469-0865                         | Email Addresses krobie@soinintl.com                           |

| Wick   | Kay   | Comm Preference                                   |
|--|---|---|
| Phone #<br>Mailing Address<br>Street Address<br>City<br>State<br>Zip | 861 Deer Run Road<br>Centerville<br>OH<br>45459 | PositionDirectorTerm3 yr, 1/10ExpirationJune 2013 |
| Business Phone<br>Home Phone   | 0<br>937-433-1352                               | Business Fax<br>Home Fax                          |
| Mobile Phone   | 937-416-2526                                    | Email Addresses kaycwick@aol.com                  |

ACCOUNTS OF THE SUCCESSION

# Attachment 2.1

# School Governing Authority Board of Directors

The Board of Directors Shall be selected and removed in the manner outlined in the adopted Code of Regulations found in Attachment 1.5.

### PATHWAY SCHOOL OF DISCOVERY BOARD OF DIRECTORS

### PROPOSED 2012 – 2013 BOARD CALENDAR

The Board shall meet at <u>4:00 PM</u> at Pathway School of Discovery 173 Avondale Drive Dayton, OH 45404 937-235-5498

On the following dates:

Wednesday, August 15, 2012

Wednesday, September 19, 2012

Wednesday, November 21, 2012

Wednesday, January 16, 2013

Wednesday, March 20, 2013

Wednesday, May 15, 2013

Wednesday, June 19, 2013 at North Dayton, 3901 Turner Rd.

| <u> </u>   |            |          |
|--|------------|----------|
| Posted at Pathway School of Discovery by             |            | on       |
| at   | •          |          |
| Emailed to Educational Service Center of Lake Erie V | Vest on <> |          |
|  |            |          |
| Faxed to Dayton Daily on                             | at         | <u> </u> |
|  |            |          |

3850 Broadmoor Avenue SE, Suite 201, Grand Rapids, MI 49512 173 Avondale Drive, Dayton, OH 45404

Any person with a disability who needs accommodation to participate in a meeting should contact the Principal of Pathway School of Discovery at 937-235-5498 at least 5 days in advance of the meeting to request assistance.

# **ATTACHMENT 3.1**

Transportation Plan

(Must include transportation for Career Tech Programs or other offsite plan for school sponsored curricular programs, as well as to and from school)

# Attachment 3.1

# Transportation

The school shall not provide transportation services. The school will work with the school district in which the school is located to provide transportation services to students who meet the eligibility requirements set by the school district.

# **ATTACHMENT 3.2**

Management Contract (if any)

[See also, the ESCLEW Principals for School-Management Company relationships for guidance attached hereto]

# EDUCATIONAL SERVICE CENTER OF LAKE ERIE WEST OPERATING PRINCIPLES COMMUNITY SCHOOLS AND MANAGEMENT COMPANIES

- The Educational Service Center of Lake Erie West (ESCLEW) is not in favor of any law or contractual provision that gives a management company (whether a for profit Educational Management Organization ("EMO") or a not-for-profit Charter Management Organization ("CMO") any right to replace or appoint a governing authority member of an Ohio community school.
- The ESCLEW recognizes, and the law generally requires that governing authorities appoint or elect new governing authority members. Although any stakeholder or interested party, including a management company, may submit resumes of disinterested persons, or recommendations or suggestions, the governing authority makes the sole election, vote and decision and the management company does not participate in that appointment or voting process.
- The ESCLEW, while preserving the concept of competition and the varied structures of management contracts among EMO's and CMO's, will strongly encourage a governing authority to assess the type of contract contemplated, its pros and cons, and its provisions to assure quality administrative management of the school.
- The ESCLEW will use its influence to improve the relationships between EMO's, CMO's and School Governing Authorities based on transparency, trust, quality, and best practices between boards and their administrations.
- The ESCLEW will monitor expectations of a management company by (a) requiring attachment of any management agreement to and within the charter contract and (b) approving any subsequently new or modified contracts, and (c) monitoring through its oversight of the school, how the management company carries out the provisions of the charter contract between the ESCLEW and the school.
- The ESCLEW will strongly encourage school governing authorities to annually evaluate any management company's performance under the school-management company contract, and provide evidence to the sponsor that the evaluation occurred.
- The ESCLEW recognizes that by law, governing authority members may be paid a limited stipend pursuant to the applicable laws. Statutorily, the stipend may be required to be paid for by a management company, not because the board member is an employee of the management company, but rather that the management company is required to pay the stipend on behalf of the governing authority.
- The ESCLEW will request that management contracts (a) continue through a school year at the discretion of the school, so long as the management company is being paid for services rendered, and (b) provide transition services or cooperation services if terminated.
- The ESCLEW will remain neutral in a management company-school contract dispute, but shall testify, if required, as to the facts it knows first hand, documents signed by it, and matters related to its oversight as a sponsor. The ESCLEW will monitor disputes and take all reasonable steps necessary to avoid disruption to services for public school children. The well-being of public school students and public school education will be primary in these situations.
- The ESCLEW believes that management companies in primary charge of educational services must present clear and concise educational performance data to the governing authorities of the schools managed in order to show growth, progress or lack thereof in all areas measured by the State of Ohio and otherwise required by the ESCLEW.

- The ESCLEW believes that purchases made with public funds belongs to the school. Purchases paid for by the fee for service to a management company are fees for services rendered unless otherwise required by law (such as for equipment or real property under federal grants).
- The ESCLEW recognizes that Community School Governing Authorities control all revenues, through their treasurers, fiscal agents, grant monitoring and contracts. Fees paid out to management companies are for administration and consulting services rendered and are not considered different than fees paid directly to other contractors or administrators.
- The ESCLEW accepts that management contracts with start-up service fees are appropriate so long as the fees are documented, transparent and approved legitimate start-up services provided to the governing authority. Management contracts with termination fees are approved if the fees are based on past services, reimbursements, closure services, or fees approved by the State auditor or previously approved by the Internal Revenue Service in the school's Application for Tax Exempt Status.
- The ESCLEW expects that loans to a community school from a management company be documented clearly in writing and for a stated term, and that interest, if any, should be at or below market rate.
- The ESCLEW believes that managed schools sponsored by the ESCLEW should make provisions with the EMO or CMO for the governing authority to have its own attorney and accountants, in the case of negotiations or controversy between the parties. If a school has not made allowances for this in its management agreement, the ESCLEW feels that the management company should pay for, budget, or set aside provisions for the school's independent professional advisors as needed. The ESCLEW will require that revised or updated contracts address this issue.

The ESCLEW does not disapprove of a management company paying for the school's audits, accountants or attorneys from its fee, so long as the school's independent advisors are assured independence and payment, and confidentiality is preserved.

• The ESCLEW does not act as an auditor in checking all expenses of the school under vendor contracts, and does not assess the value of the services provided under third-party contracts. In addition, the ESCLEW expects the schools it sponsors to exert influence in their management contracts to require the management company to (a) comply with the guidance and laws concerning reporting expenditures to the auditor (if required by law), and (b) to guard against duplication of services between the management company, the school and other non-management company vendors and (c) to otherwise follow all laws, rules and regulations of the auditors and other applicable governmental agencies. The ESCLEW does not audit an independent, private management company, but expects that company to be able to substantiate to the School, that its staffing is sufficient to provide the services contracted for. If an EMO or a CMO is not in compliance with all laws, rules, or regulations, the ESCLEW may use all available remedies it may have at its disposal to require compliance.

#### MANAGEMENT AGREEMENT

This Management Agreement is made and entered into as of the <u>S</u> day of <u>March</u>, 2002, by and between National Heritage Academies, Inc. a Michigan corporation ("NHA"), and East Dayton School of Discovery, an Ohio non profit corporation and Ohio public community school (the "School").

#### RECITALS

The School is a charter school, organized as a public community school under the revised Ohio Code Section 3314 (the "Code"). The School has submitted an application (the "Application") for, and has been or will be granted, a contract (the "Contract") by the Ohio Board of Education (the "Sponsor") to organize and operate a public community school, with Sponsor as the authorizing body.

The School and NHA desire to create an enduring educational alliance, whereby the School and NHA will work together to promote educational excellence and innovation, based on NHA's school design, comprehensive educational program and management principles.

In order to facilitate the organization and opening of the School, and to implement an innovative educational program at the School, the parties desire to establish this arrangement for the management and operation of the School.

Therefore, for good and valuable consideration, the receipt of which is hereby acknowledged, it is mutually agreed as follows:

#### ARTICLE I

#### **CONTRACTING RELATIONSHIP**

A. Authority. The School represents that it is authorized by law to contract with a private entity and for that entity to provide educational management services. The School further represents that it has been or will be granted the Contract by Sponsor to organize and operate a public community school. The School is therefore authorized by the Code and Sponsor to supervise and control such school, and is invested with all powers necessary or desirable for carrying out the educational program contemplated in this Agreement.

**B.** Contract. The School hereby contracts with NHA, to the extent permitted by law, for the provision of all labor, materials, equipment, facilities and supervision necessary for the provision of educational services to students, and the management, operation and maintenance of the School in accordance with the educational goals, instructional programs, curriculum, methods of pupil assessment, admission and other policies and criteria, school calendar and school day schedule, age and grade range of pupils to be enrolled, and method to be used to monitor compliance with performance of targeted educational outcomes, all as adopted by the School's Board of Directors (the "Board") and included in the Contract between the

School and Sponsor. The School certifies that upon its execution, this Agreement will be a valid and binding obligation of the School enforceable in accordance with its terms and to the best of its knowledge, except as the enforceability thereof may be limited by applicable reorganization, insolvency, liquidation, adjustment of debt, moratorium or other similar laws affecting the enforcement of the rights of creditors generally as such laws may be applied in the event of the reorganization, insolvency, liquidation, adjustment of debt or other similar proceeding or a moratorium applicable to the School and by general principles of equity.

b

C. Designation of Agents. The Board designates the employees of NHA as agents of the School having a legitimate educational interest such that they are entitled to access to educational records under 20 U.S.C. §1232g, the Family Rights and Privacy Act ("FERPA").

**D.** Status of the Parties. NHA is a for-profit Michigan corporation, and is not a division or a part of the School. The School is an Ohio non profit corporation and part of the State of Ohio's program of public education as authorized by the Code, and is not a division or part of NHA. The parties to this Agreement intend that the relationship created by this Agreement is that of an independent contractor and not employer - employee. Except as expressly provided in this Agreement, no agent or employee of NHA shall be deemed to be the agent or employee of the School. NHA shall be solely responsible for its acts and the acts of its agents, employees and subcontractors. The relationship between NHA and the School is based solely on the terms of this Agreement, and the terms of any other written agreements between NHA and the School.

#### **ARTICLE II**

#### TERM

A. Term. This Agreement shall be effective as of  $\underline{July 1, 2002}$  and unless otherwise renewed or terminated pursuant to this Agreement shall continue until termination or expiration of the Contract, inclusive of any contract renewal periods and subject to the terms and obligations herein.

#### ARTICLE III

#### **FUNCTIONS OF NHA**

A. **Responsibility.** NHA shall be responsible and accountable to the Board for the administration, operation and performance of the School in accordance with the Contract. NHA's responsibility is expressly limited by: (i) the School's budget which is to be submitted and approved by the Board as provided in this Agreement, and (ii) the availability of base state funding to pay for said services. Neither NHA nor the School shall be required to expend School funds on services in excess of the amount set forth in the School budget.

**B.** Educational Program. NHA agrees to implement the educational goals and programs as set forth in the Contract (the "Educational Program"). In the event NHA determines that it is necessary to modify the Educational Program, NHA shall inform the Board of the

proposed changes and obtain Board approval, and if required under the Contract, approval of Sponsor. The parties hereto acknowledge that an essential principle of the Educational Program is its flexibility, adaptability and capacity to change in the interest of continuous improvement and efficiency, and that the School and NHA are interested in results and not in inflexible prescriptions. Not less than annually, and otherwise as requested, NHA will provide the Board with updated reports on progress towards implementing each of the School's educational goals in the Educational Program.

4

(

C. Specific Functions. NHA shall be responsible for the management, operation, administration, accounting and Educational Program at the School. Such functions include, but are not limited to:

1. Implementation and administration of the Educational Program, including the selection and acquisition of instructional materials, equipment and supplies, and the administration of any and all extra-curricular and co-curricular activities and programs approved by the Board.

2. Management of all personnel functions, including professional development for the School Administrator and all instructional personnel and the personnel functions outlined in Article VI.

3. Securing a facility to be leased or otherwise provided to the Board, operation of the facility, and the installation of technology integral to the school design.

4. All aspects of the business administration of the School.

5. All aspects of the accounting operation, including budgets, general ledger management and financial reporting, including but not limited to audits (annual or special).

6. Transportation and food service, if any is provided, for the School.

7. Any other function necessary or expedient for the administration of the School, including but not limited to the annual reports, ADM or similar reporting, EMIS and other reporting required by Ohio law.

**D. Purchases.** Purchases made by NHA on behalf of the School with the School's funds, such as non-proprietary instructional and/or curriculum materials, books and supplies, and equipment will be the property of the School (exclusive of capital items leased or purchased by NHA). NHA's educational materials and teaching techniques used by or at the School shall be subject to disclosure to the extent required under the Ohio Revised Code and the Freedom of Information Act. However, NHA shall own all proprietary rights to, and the School's proprietary interest shall not include, curriculum or educational materials that were previously developed or copyrighted or similarly protected by NHA, or curriculum or educational materials that are developed by NHA with funds from the School that are not otherwise dedicated for the specific purpose of developing School curriculum or educational materials.

E. Subcontracts. NHA reserves the right to subcontract any and all aspects of all services it agrees to provide to the School, including, but not limited to transportation and/or

3

food service. However, NHA shall not subcontract the management, oversight or operation of the teaching and Instructional Program, except as specifically permitted in this Agreement or with approval of the Board.

**F.** Place of Performance. NHA reserves the right to perform functions other than instruction, such as purchasing, professional development, and administrative functions, off-site, unless prohibited by state or local law.

**G.** Student Recruitment. NHA and the Board shall be jointly responsible for the recruitment of students subject to agreement on general recruitment and admission policies. Students shall be selected in accordance with the procedures set forth in the Contract and in compliance with the Ohio Revised Code and other applicable law.

**H. Due Process Hearings.** NHA shall provide student due process hearings in conformity with the requirements of state and federal law regarding discipline, special education, confidentiality and access to records, to an extent consistent with the School's own obligations.

I. Legal Requirements. NHA shall provide educational programs that meet federal, state, and local requirements, and the requirements imposed under the Code and the Contract, unless such requirements are or have been waived, but the School shall interpret state and local regulations as liberally as allowed by law, in order to give NHA flexibility and freedom to implement its educational and management programs.

J. Rules and Procedures. NHA shall recommend reasonable rules, regulations and procedures applicable to the School and is authorized and directed to enforce such rules, regulations and procedures adopted by the School.

)

**K.** School Year and School Day. The school year and the school day shall be as provided in the Contract consistent with the Ohio Revised Code.

L. Pupil Performance Standards and Evaluation. NHA shall implement pupil performance evaluations that permit evaluation of the education progress of each School student. NHA shall be responsible and accountable to the Board for the performance of students who attend the School. NHA will utilize assessment strategies required by the terms of the Contract. The Board and NHA will cooperate in good faith to identify measures of and goals for School students and school performance, including but not limited to parent satisfaction.

M. Services to Disabled Students and Special Education. NHA shall provide special education services to students who attend the School in conformity with the requirements of state and federal law. NHA may subcontract as necessary and appropriate for the provision of services to students whose special needs cannot be met within the School's program, subject to approval of the School Board. Such services shall be provided in a manner that complies with local, state and federal laws and applicable regulations and policies.

N. Contract between the School and Sponsor. NHA will not act in a manner which will cause the School to be in breach of its Contract with the Sponsor or its Policies or with applicable law.

4

**O.** Unusual Events. NHA agrees to timely notify the Board and/or school administrator of any anticipated or known: (i) material health or safety issues, (ii) labor, employee or funding problems, or (iii) problems of any other type that could adversely affect the School in complying with its responsibilities hereunder.

(

)

**P.** Student and Financial Records. All student and financial information related to the School shall be available for inspection at the School upon reasonable request consistent with applicable federal and state laws.

**Q.** School Records/Proprietary. The financial, educational and student records pertaining to the School are School property, and such records are subject to the provisions of the Ohio Public Records Law to the extent required by applicable law. All School records shall be physically or electronically available, upon request, at the School. Except as prohibited under the Contract and applicable law, the Sponsor and the public shall have access to the School's records.

#### ARTICLE IV

### **OBLIGATIONS OF THE BOARD**

A. Good Faith Obligation. The Board shall be responsible for its fiscal and academic policy. The Board shall exercise good faith in considering the recommendations of NHA, including but not limited to, NHA's recommendations concerning policies, rules, regulations and budgets. The Board's failure to adopt NHA's reasonable recommendations with respect to policies, rules and regulations to enable NHA to implement the school design as set forth in the Contract may, at NHA's option, be deemed a material event which shall entitle NHA, at its option, to unilaterally terminate this Agreement pursuant to Article VII.A.1.

**B.** Assistance to NHA. The School shall cooperate with NHA in furnishing all information and submitting all forms and reports required in association with this Agreement, including timely notice of all Board meetings. The School shall timely furnish NHA all documents and records necessary and in the Board's possession for NHA to properly perform its responsibilities under this Agreement.

C. Unusual Events. The School agrees to timely notify NHA of any anticipated or known: (i) material health or safety issues, (ii) labor, employee or funding problems, or (iii) problems of any other type that could adversely affect NHA in complying with its responsibilities hereunder.

**D. NHA Office Space.** For the term of this Agreement, suitable office space shall upon request be provided at the School without cost for NHA personnel and subcontractors. The office space shall be used by NHA only for NHA activities related to the School. The School shall also provide NHA, upon NHA's request, with an additional room to be used for activities related to the School.

E. Retained Authority. The Board shall retain the authority to make reasonable regulations relative to anything necessary for the proper establishment, maintenance,

management, and carrying on of the School, including regulations relative to the conduct of pupils while in attendance at the School or en route to and from the School.

.

**F.** Building Facility. NHA shall lease or otherwise cause a facility to be made available to the Board for school classroom facilities. The facility shall comply with, or otherwise be approved, with respect to state and federal (ie: ADA) regulations governing the use of the facility as an elementary/middle school, as applicable.

### ARTICLE V FINANCIAL ARRANGEMENTS

A. Revenues. Except as hereinafter provided, all monies received by the School Board shall be deposited in the School's depository account within three (3) business days with a financial institution acceptable to the Board, provided, however, that upon the receipt of a notice from NHA, the School agrees to pay all such sums owing under this Agreement directly to the account or party specified in such notice. Except as specifically excluded by the terms of this Agreement, the term "Revenues" shall include all funds received by or on behalf of the School, including but not limited to:

- 1) Funding for public school students enrolled in the School.
- 2) Special education funding provided by Federal or State Governments to the School that is directly allocable to special education students in the School.
- 3) Gifted and Talented funding provided by Federal and State Governments that is directly allocable to Gifted and Talented students in the School.
- 4) At-Risk Funding provided by Federal and State Governments to the School that is directly allocable to At-Risk students in the School.
- 5) Funding provided by Federal and State Governments to the School that is directly allocable to students in the School with limited English proficiency.
- 6) Federal and State grant sources, including Title I, which is directly allocable to the School.
- All other grants and donations received by the School (except to the extent NHA is not required or involved in soliciting, administering, or managing such grants and/or donations).
- 8) Fees charged to students for extra services as and to the extent permitted by law.
- (All of the above are hereinafter collectively referred to as the "Revenues").

In performing its obligations under this Agreement, NHA shall comply with the terms of the Budget approved and authorized by the Board, and NHA thereby guarantees fiscal solvency. The expenditure of Revenues received from governmental entities shall be consistent with all applicable regulations and policies, and in the case of private donations, the directives of the donor where applicable.

#### B. Budget

Ĺ

( )

1. <u>Projected Budget</u>. NHA shall provide the Board with an annual projected Budget (the "Budget"). For the School's first academic year, the Budget shall be submitted prior to the beginning of the academic year. Thereafter, the Budget shall be submitted to the Board prior to May 1st of the next academic year.

2. <u>Budget Detail</u>. The Budget shall contain reasonable detail as requested by the Board. The Budget shall include all projected expenses and costs reasonably associated with operating the School and the NHA school program including, but not limited to, the projected cost of: all services and education programs provided to the School, leasehold and other lease or purchase costs incurred for the facility, maintenance and repairs to School facilities and capital improvements except as otherwise agreed upon, supplies and furnishings necessary to operate the School, all taxes of any kind that are assessed or imposed, insurance premiums, utilities, professional fees, and other costs and expenses connected to operating the School.

3. <u>Approval</u>. The Budget shall be prepared by NHA and submitted to the Board for approval. The Budget may be amended from time to time as deemed necessary by NHA and the Board.

4. <u>Expenditures</u>. NHA shall comply with the provisions of the Budget or amended Budget and shall not deviate materially therefrom without Board approval.

5. <u>Board Designated Funds.</u> Notwithstanding any other provision of this Agreement, during the term of this Agreement, there shall be reserved in the School's account an amount up to 2%, not to exceed \$35,000. The designated funds shall be used for the annual audit by the Auditor of the State of Ohio. No special or independent audits shall be paid by Board designated funds, however, the designated funds may otherwise be used during the year by the Board at its discretion. During each year of the Agreement, the designated funds shall accumulate pro rata as Revenues are received during the year.

C. Fee. NHA shall receive all Revenues as its gross revenue, and it shall pay all operating costs of the School identified in the Budget approved by the Board. The difference, if any, between the amount of the School's Revenues and the amount actually expended by NHA in operation and/or management of the School during its fiscal year shall be referred to herein as the "Fee". Notwithstanding any provision to the contrary contained herein, the School acknowledges and agrees that, with the exception of moneys which, pursuant to State or federal law or in the case of private donations the directions of the donor, are restricted as to use, upon the receipt of Revenues due or payable to the School either by the Sponsor or a governmental unit, such moneys immediately become compensation of NHA for its services provided pursuant to this Agreement. Further, the School understands and acknowledges that the School has no further right to or claim upon the Revenues as outlined in Article VII of this agreement.

**D.** Availability of Funds. NHA shall only be required to perform its responsibilities upon this Agreement to the extent that there are sufficient Revenues on a per

7

annum basis to make payments in accordance with the terms of the Budget provided such budget has been reasonably updated on a regular basis to reflect needed changes.

E. Other Public School Academies. The School acknowledges that NHA may enter into similar management agreements with other public school academies. NHA shall maintain separate accounts for expenses incurred by and behalf of the School and other NHA managed schools, and shall only charge the School for expenses incurred by or on behalf of the School. If NHA incurs authorized reimbursable expenses on behalf of the School and other NHA managed schools which are incapable of precise allocation, then NHA shall allocate such expenses among all such academies, including the School, on a prorated basis based upon the number of students enrolled at such academies, or upon such other equitable basis as is acceptable to the parties.

F. Financial Reporting. NHA shall provide the Board with:

()

1) The projected annual Budget as required by the terms of this Agreement.

2) Detailed statements of all Revenues received, and detailed statements of all direct expenditures for services and or expenses rendered or incurred to or on behalf of the School, whether incurred on-site or off-site, upon request.

3) Reports on School operations, finances, and student performance shall be provided upon request (at regularly scheduled Board meetings), but not less frequently than four (4) times per year.

4) Other information on a periodic basis to enable the Board to (i) monitor NHA's performance and the efficiency of its operation of the School, and (ii) furnish reports and information which the School is required to provide pursuant to its Contract and/or the Code.

G. Access to Records. NHA shall keep accurate financial records pertaining to its operation of the School, together with all School financial records prepared by or in possession of NHA, and shall retain all of the said records for a period of five (5) years from the close of the fiscal year to which such books, accounts, and records relate. NHA and the School shall maintain the proper confidentiality of personnel, students, and other records as required by law.

H. Review of Operational Budget. The School Board shall be responsible for reviewing and approving the annual Budget of the School.

I. Independent Audit. If required by statute or to obtain financing, the School Board and NHA shall select and retain an independent auditor to conduct an audit of the School. Subject to applicable law, all finance and other records of NHA related to the School will be made available to the School's independent auditor. The annual audit to be performed by the Auditor of the State will be paid by the School using Board designated funds. Any independent or special audit shall be paid for through Revenues by way of the NHA fee.

J. Start-up Financing/Operating Losses. NHA will provide funds for start-up costs and/or operating losses for the School, including funds for the development of a curriculum, technology system and school operations plan; recruiting, selecting and pre-service

training of staff members; and cleaning, fixing and equipping of the School building as required by this Agreement. NHA advances shall be included in the Budget and shall be in amounts acceptable to NHA. NHA shall be reimbursed from the Revenues as and when funds are available and upon proper documentation. Nothing herein shall be deemed to exclude the School from receiving state or federal start-up funds, grants or sub-grants.

**K.** Marketing. Marketing and development costs paid by or charged to the School shall be limited to those costs specific to the School program, and shall not include costs for the marketing and development of NHA.

L. Other Financing. The Board may apply to NHA for financing from time to time. The School shall reimburse NHA for any such financing from its Revenues.

#### ARTICLE VI

#### PERSONNEL & TRAINING

A. Personnel Responsibility. NHA shall select and hire qualified personnel to perform services at the School. Personnel shall be employees of NHA, unless otherwise agreed by NHA and the Board. Each party shall be responsible for all compensation for their respective employees. The compensation of all employees will be paid in accordance with the Budget referenced in Article V as amended from time to time in a manner mutually acceptable to the Board and NHA. NHA shall have the responsibility and authority to determine staffing levels, and to select, evaluate, assign, discipline, transfer and terminate personnel consistent with state and federal law.

School Administrator. The accountability of NHA to the School is an essential В. foundation of this Agreement. Since the responsibility of the school administrator is critical to the School's success, NHA shall have the authority, consistent with state law, to select and supervise each school administrator and to hold the school administrator accountable for the success of the School. NHA shall consult with the Board with respect to the hiring of the School Administrator, and NHA shall remove the School Administrator from the School if the Board is reasonably dissatisfied with his or her performance. The duties of the school administrator, and the terms of the school administrator's employment contract, shall be determined by NHA subject to approval by the Board. The school administrator shall be accountable for the performance of the School, and shall work with NHA in conjunction with the operation and management of the School. Since the selection and performance of the school administrator is essential to the success of the School, the Board's failure to adopt/ratify NHA's recommendations with respect to the school administrator shall, at NHA's option, constitute a material event which shall entitle NHA, at its option, to unilaterally terminate this Agreement pursuant to Article VII A.1. below.

C. Teachers. NHA shall determine the number of teachers, and the applicable grade levels and subjects, required for the operation of the School. NHA shall provide the School with such teachers, qualified in the grade levels and subjects required, as are required by the School. The curriculum taught by such teachers shall be consistent with the Educational Program. Such teachers may, in the discretion of NHA, work at the School on a full or part time

9

basis. If assigned to the School on a part time basis, such teachers may also work at other schools managed or operated by NHA. Each teacher assigned to the School shall hold a valid teaching certificate issued by the state board of education under the Code and undergo a criminal background check as if such teacher was employed by the School.

**D.** Support Staff. NHA shall determine the number and the functions of support staff required for the operation of the School. NHA shall provide the School with qualified staff to efficiently operate the school in accordance with the Contract. The support staff may, in the discretion of NHA, work at the School on a full or part time basis. If assigned to the School on a part time basis, the support staff may also work at other schools managed or operated by NHA.

**E. Training.** NHA shall provide training in its methods, curriculum, program, and technology to all teaching personnel on a regular basis. Non-instructional personnel shall receive such training as NHA determines reasonable and necessary under the circumstances.

**F.** Limitations on Discretion. All decisions made by NHA, and any discretion exercised by NHA, in its determination of staffing levels and its selection, evaluation, assignment, discipline, and transfer of personnel, shall be consistent with the Budget, State and federal law, and consistent with the parameters adopted and included within the Educational Program.

G. Background Checks. NHA shall comply with Ohio law regarding background checks and certification or licensure, as applicable, for all persons working in the School.

#### ARTICLE VII

#### TERMINATION OF AGREEMENT

#### A. Termination.

1. By NHA. NHA may, at its option, terminate this Agreement prior to the end of the terms specified in Article II in the event the Board fails to remedy a material event within 30 days after notice from NHA. A material event includes, but is not limited to, NHA's failure to receive for any reason compensation or reimbursement as required by the terms of this Agreement, or the School's loss or suspension of its Contract.

2. By School. The School may terminate this Agreement prior to the end of the terms specified in Article II in the event that NHA shall fail to remedy a material breach within 30 days after notice from the Board. Material breach includes, but is not limited to: (i) failure to account for its expenditures or to pay School operating costs (provided funds are available to do so), (ii) failure to follow policies, procedures, rules, regulations or curriculum duly adopted by the Board which are not in violation of the Contract, this Agreement, or law, or (iii) receipt by the Board of unsatisfactory reports from NHA or from an educational consultant retained by the Board about matters concerning NHA's performance or the performance of the staff which are not adequately corrected or explained. 3. By Either Party. Either party may terminate this Agreement for any reason upon giving not less than 90 days notice to the other party.

#### **B.** Termination/Expiration.

ſ

()

1. Effective Date of Termination. In the event this Agreement is terminated by either party prior to the end of the term specified in Article II, absent a material event or unusual and compelling circumstances, the termination will not become effective until the end of the current academic year.

2. Removal of personal property. Upon termination or expiration of this Agreement, NHA shall have the right to remove equipment and other assets owned or leased by NHA. Equipment and other assets owned by the School or leased by the School from third parties shall remain the property of the School.

3. Future Advances/Out-of-Pocket Expenses. Upon termination or expiration of this Agreement, for any reason, all future advances or out-of-pocket expenses paid by NHA shall be immediately repaid by the School unless otherwise agreed in writing by NHA.

#### ARTICLE VIII

#### INDEMNIFICATION

A. Indemnification. Each party to this Agreement does hereby indemnify and hold harmless the other, and Sponsor, and their respective boards of directors, partners, officers, employees, agents, representatives, and attorneys from and against any and all claims, actions, damages, expenses, losses or awards which arise out of (i) the negligence of the other party, (ii) any action taken or not taken by the other party, or (iii) any noncompliance or breach by the other party of any of the terms, conditions, warranties, representations, or undertakings contained in or made pursuant to this Agreement. As used in this subsection, "party" shall include the party's trustees, directors, officers, employees, agents, representatives and attorneys. Such indemnification may be achieved by the joint purchase of general liability and property insurance policies, by directors and officers' liability policies or by such other means as the parties may mutually agree.

#### ARTICLE IX

#### INSURANCE

A. Insurance Coverage. Each party shall maintain general liability insurance in the amount of One Million Dollars (\$1,000,000.00) per occurrence (or such greater amount if required by the terms of the Contract or applicable law), with the other party listed as an additional insured. In addition, the School shall maintain an umbrella liability policy of two million dollars (\$2,000,000.00) (or such greater amount if required by the terms of the Contract or applicable law), with NHA listed as an additional insured. The building and related capital facilities remain the responsibility of the Board and the Board shall cover its property with insurance. Each party shall, upon request, present evidence to the other that it maintains the requisite insurance in compliance with the provisions of this paragraph. Each party shall comply with any information or reporting requirements required by the other party's insurer(s), to the extent reasonably practicable.

(

}

()

**B.** Workers' Compensation Insurance. Each party shall maintain workers' compensation insurance as required by law, covering their respective employees.

#### ARTICLE X WARRANTIES AND REPRESENTATIONS

A. School Warranties and Representations. The School represents that it has the authority under law to execute, deliver and perform this Agreement and to incur the obligations provided for under this Agreement. The Board warrants that its actions have been duly and validly authorized, and that it will adopt any and all resolutions or expenditure approvals required for execution of this Agreement.

**B.** NHA Warranties and Representations. NHA warrants and represents that it is a corporation in good standing and is authorized to conduct business in the State of Michigan. NHA will comply with all registration and licensing requirements relating to conducting business under this Agreement and in the State of Ohio. The School agrees to assist NHA in applying for such licenses and permits and in obtaining such approvals and consents.

C. Mutual Warranties. The School and NHA mutually warrant to the other that there are no pending actions, claims, suits or proceedings, to its knowledge, threatened or reasonably anticipated against or affecting it, which if adversely determined, would have a material adverse effect on its ability to perform its obligations under this Agreement.

#### ARTICLE XI

#### MISCELLANEOUS

A. Sole Agreement. This Agreement supersedes and replaces any and all prior agreements and understandings between the School and NHA.

( )

)

**B.** Force Majeure. Notwithstanding any other sections of this Agreement, neither party shall be liable for any delay in performance or inability to perform due to acts of God or due to war, riot, embargo, fire, explosion, sabotage, flood, accident, labor strike, or or other similar occurrences; provided either party may terminate this Agreement in accordance with the termination provisions contained in this Agreement if sufficient grounds exist as provided in the Article governing termination.

C. State Governing Law/Waiver of Jury Trial. The rights of all parties hereto shall be subject to the jurisdiction of and be construed according to the laws of the State of Ohio. NHA and the School hereby waive the right to a jury trial in any action, proceeding or counterclaim brought by either NHA or the School against the other.

**D.** Agreement in Entirety. This Agreement (including attachments) constitutes the entire agreement of the parties.

E. Official Notices. All notices and other communications required by the terms of this Agreement shall be in writing and sent to the parties hereto at the facsimile number or address set forth below. Notice may be given by (i) by facsimile with written evidence of confirmed receipt by the receiving party of the entire notice, (ii) certified or registered mail, postage prepaid, return receipt requested, or (iii) personal delivery. Notice shall be deemed to

have been given on the date of transmittal or personal delivery if given by facsimile or personal delivery, or upon the date of postmark if sent by certified or registered mail. Notices to the School shall be sent to the current address of the then current Board President, with a copy to the then current Board attorney. The address of the parties hereto for the purposes aforesaid, inclusive of the address of the initial Board President and Board, are as follows:

The School:

۰.

( )

( )

East Dayton School of Discovery

173 Avondale Road Dayton, Ott 45404 Phone: Facsimile:

WITH COPY TO:

Eastman & Smith Ltd. Attn: Amy Borman P.O. Box 10032 Toledo, OH 43699-0032 Phone: (419) 247-1716 Facsimile: (419) 247-1777

NHA:

National Heritage Academies, Inc. Attn: Peter R. Ruppert 989 Spaulding Avenue, S.E. Grand Rapids, Michigan 49546 Telephone: (616) 222-1700 Facsimile: (616) 222-1701

WITH A COPY TO:

McShane & Bowie Attn: John R. Grant 1100 Campau Square Plaza 99 Monroe Ave., N.W. Grand Rapids, Michigan 49501

**F.** Assignment. NHA may assign this Agreement with the consent of the School Board.

G. Amendment. This Agreement shall not be altered, amended, modified or supplemented except by memorandum approved by the Board and signed by both the President of the School and authorized officer of NHA.

**H.** Waiver. No waiver of any provision of this Agreement shall be deemed or shall constitute a waiver of any other provision. Nor shall such waiver constitute a continuing waiver unless otherwise expressly stated.

I. Cost and Expenses. If any party commences an action against another party as a result of a breach or alleged breach of this Agreement, the prevailing party shall be entitled to have and recover from the losing party reasonable attorneys' fees and costs of suit.

J. Severability. Should any term or provision of this Agreement be deemed unenforceable or invalid in any way, such term shall be stricken and the remainder of this Agreement shall continue in full force and effect, or, such term or provision shall be replaced, if mutually agreed upon, with a new provision that replicates the parties' original intent as nearly as possible.

**K.** Compliance with Law. The parties to this Agreement agree to comply with all applicable laws and regulations.

IN WITNESS WHEREOF, the undersigned have executed this Agreement as of the date and year first above written.

Date: March 8 .2002

Date: March 8

(

NATIONAL HERITAGE ACADEMIES, INC., a Michigan Corporation

Bv: Peter Rupperd

Its: President

EAST DAYTON SCHOOL OF DISCOVERY By: By: By: Bulen

Its: PRESIDENT

H:\HOME\AJBorman\East Dayton School of Discovery\Mgt agreement 12 27 01 version.wpd U:\Universal\BoardRelations\Ohio Schools\Management Agreement 2 11 02 kmh LANSING 26755-5 292734-2

.2002

# ATTACHMENT 3.5.3

# Attachment 3.5.3

| School | Name: |  |
|--------|-------|--|
|        |       |  |

Sponsor Name\_\_\_\_\_

| I. Initial Notificat | ions, Student Records and School Records   |
|----------------------|--|
| Completion Date      | Mandatory Task   |
|                      | Notify ODE that the school is closing and send the board resolution or an official sponsor notice from the superintendent or CEO within 24 hours of the action which includes the date of closing of the school. A school is officially closed when instruction is no longer taking place and the governing authority or sponsor issues an official notice which states the date of the school's closure.                              |
|                      | Notify parents that the school is closing through a formal letter from the school GA and the sponsor superintendent or CEO within 24 hours of the action. The letter to include but not limited to: the reason for the closing of the school, options for enrolling in another community school, traditional schools or nonpublic schools and contact information.   |
|                      | Notify the Ohio State Teachers Retirement System and School Employees Retirement<br>System.  |
|                      | The sponsor shall take control of and secure all school records, property and assets immediately when the school closes: Reference the following DAS site for records retention schedules: http://apps.das.ohio.gov/rims/General/General.asp   |
|                      | 1. Student records shall be put into order and transcript materials produced immediately;  |
|                      | <ol><li>A final FTE review shall be requested while student records are on site at the closed school<br/>and original student records shall be retained for the final state audit;</li></ol>   |
|                      | 3. Copies of student records shall be provided by the school's Chief Administrative Officer to<br>all resident districts within seven business days of closure of the school as defined in section<br>3314.44 of the revised code; original records shall be sent by the sponsor to resident districts<br>upon completion of the final state audit. In the interim, sponsors retain original records until<br>completion of the audit. |
|                      | 4. Special education records shall be provided directly to receiving school special education<br>administrators for all students with disabilities, particularly for students with physical needs or<br>low incidence disabilities.  |
|                      | In concert with the governing authority, notify the school's staff of the decision to close the school:  |
|                      | 1. Provide a clear written timeline of the closing process;  |
|                      | 2. Ensure that STRS and SERS contributions are current;  |
|                      | 3. Clarify COBRA benefits and when medical benefits end;   |
|                      | <ol> <li>Remind the faculty of their obligation to teach up to the date of closing or otherwise<br/>determine that the school is properly staffed up to the time of closing;</li> </ol>  |
|                      | 5. Ensure that each faculty's LPDC information is current and available to the teachers;   |
|                      | 6. Provide sponsor contact person information to all staff.  |

| <b>Completion Date</b> | Mandatory Task  |
|------------------------|---|
|                        | If the governing authority does not retain a treasurer to oversee the remaining financial activity<br>(note that this Contract requires retention of the School Treasurer for oversight of all closure<br>and post closure activity and responsibility) the sponsor, as may be provided for in the<br>community school contract with the governing authority, shall act as or appoint a receiver to<br>oversee the closing of the financial records in the absence of a fiscal officer. |
|                        | Keep State and Federal assets separated for purposes of disposition. Federal dollars<br>cannot be used to pay state liabilities. Account for all school property throughout the closing<br>process by distinguishing state from federal dollars:  |

| Completion Data | II. Disposition of Assets   |
|-----------------|---|
| Completion Date | Mandatory Task  |
|                 | 1. Review the financial records of the school;  |
| ·····           | a. Establish the fair market (initial and amortized) value via fixed assets policy, for all fixed assets;   |
|                 | b. Establish check off list of purchasers with proper USAS codes, state codes, the price of eac item and identify the source of funds;  |
|                 | c. Identify staff who will have legal authority for payment processes (e.g. checks, cash, credit cards, etc.);  |
|                 | d. Establish disposition plan for any remaining items;  |
| ·····           | e. Identify any State Facilities Commission guarantees.   |
|                 | Prepare documentation for disposition of the school's fixed assets:   |
|                 | 1. Consistent with section 3314.051, offer real property acquired from a public school district to that school district's board first at fair market value. If the district board does not accept the offer within 60 days, dispose of the property in another lawful manner below.   |
|                 | 2. For Federal Title and other consolidated and competitive funds, follow EDGAR liquidation procedures in 34 CFR 80.32 including disposition for items valued at \$5,000 or greater;  |
| · · ·           | 3. Public Charter School Program. PCSP assets must first be offered to other community schools with requisite board resolutions consistent with the purpose of the PCSP. If there are no takers, then an auction sale must be held to dispose of the assets along with the state funded assets:   |
|                 | a. Notify Office of Community Schools, then public media (print media, radio) of the date and location of any property disposition auction;   |
|                 | b. Follow EDGAR liquidation procedures in 34 CFR 80.32 for items valued at \$5,000 or greater   |
|                 | c. Provide board resolutions and minutes of any assets transferred at no cost to another school;  |
|                 | e. Provide OCS with a written report of the property, and if available, a bill of sale;   |
|                 | f. After the above steps have been taken, any remaining assets may be offered to any public<br>school district with documented board resolutions by the community school and the accepting<br>district.   |
|                 | 4. National School Lunch Program. Cafeteria equipment purchased with funds from the National School Lunch Program can only be liquidated through written guidance issued by the Office for Child Nutrition (OCN). Contact OCN prior to proceeding with any liquidation of Equipment.  |
|                 | 5. Technology. Return to eTech (formerly Ohio SchoolNet) hardware and software acquired with eTech grants.  |
|                 | Utilize only state dollars, auction proceeds, foundation dollars and any other non-federal dollars to pay the following in order: [See Note Below]  |
|                 | <ol> <li>The school must and the sponsor may consult with its legal counsel prior implementing this<br/>section to determine if there are any outstanding federal or state claims that may need to be<br/>paid first including but not limited to including federal and state taxes, Medicare, Workers<br/>Compensation, and city wage taxes. If applicable, provide documentation of any Workers<br/>Compensation claims.</li> <li>STRS/SERS/retirement systems and other adjustments</li> </ol> |
|                 | 3. Teachers and staff;  |
|                 | 4. Private creditors or those entities that have secured a judgment against the school, including audit preparation and audit costs (prepared financials); [See Note 2 Below].  |
|                 | 5. Any remaining funds shall be forwarded to ODE for redistribution to resident school districts consistent with ORC 3314.074 (A)   |
|                 | 6. If the assets of the school are insufficient to pay all persons or entities to which compensation is owed, the prioritization of the distribution of the assets to individual persons or entities within each class of payees may be determined by decree of a court in accordance section 3314.074 and Chapter 1702 of the Ohio Revised Code.   |

|                 | II. Disposition of Assets   |
|-----------------|---|
| Completion Date | Mandatory Task  |
|                 | NOTE 1: Federal dollars can be used to pay the teacher costs and STRS/SERS/retirement costs for any employees who were paid from federal funds when the school was open. NOTE 2: Audit preparation costs can be paid from federal funds under certain conditions. Contact the Office of Grants Management for specific guidance before applying any costs against federal funds to support audit costs. |

|                 | III. Preparation of Itemized Financials   |  |
|-----------------|---|--|
| Completion Date | Mandatory Task  |  |
|                 | Review and prepare the following itemized financials:   |  |
|                 | <ol> <li>Year-end financial statements, notes to the financial statements and if applicable schedule<br/>of federal awards;</li> </ol>  |  |
|                 | <ol> <li>A cash analysis (taking the previous month's recap and reconciliation of bank accounts to<br/>books) for determination of the cash balance as of the closing date);</li> </ol> |  |
|                 | 3. Compile bank statements for the year;  |  |
|                 | 4. List of investments in paper hard copy format  |  |
|                 | 5. List of all payables and indicate when a check to pay the liability clears the bank;   |  |
| · · · · ·       | 6. List all unused checks (collect and void all unused checks);   |  |
|                 | 7. List of any petty cash;  |  |
|                 | 8. List of bank accounts, closing the accounts once all transactions are cleared;   |  |
|                 | 9. List of all payroll reports including taxes, retirement or adjustments on employee contract.   |  |
|                 | 10 List of all accounts receivable.   |  |
| utert 18-0.11   | 11. List of assets and their disposition.   |  |
| <b></b>         | 12. Final FTE audit information.  |  |
|                 | 13. Arrange for and establish a date for the Auditor of State to perform a financial closeout audit.  |  |

|                 | IV. Final Payments and Adjustments  |
|-----------------|---|
| Completion Date | Mandatory Task  |
|                 | Give the sponsor rights to continually monitor the condition of the closed school and to receive<br>or transmit funds on behalf of the school as directed by an appropriate agency. Receipt of<br>funds can happen more than a year after a school's closure, and the sponsor is hereby<br>authorized to be the recipient of such funds and make all necessary adjustments. |
|                 | <ol> <li>The sponsor may receive any funds or adjustments and credited to the account of the<br/>closed school.</li> </ol>  |
|                 | 2. The Sponsor may determine if any portion of any funds or adjustments can be applied to satisfy any remaining debts and may pay such debts with school funds.   |
|                 | 3. The Sponsor may send all or the remaining portion to the Ohio Department of Education,<br>Office of Policy and Payments, for final disposition.  |

The Treasurer, the School Governing Authority, and the Chief Administrator identified below have completed the above school closing requirements.

| School Name:                |       |
|-----------------------------|-------|
| School Treasurer:           | Date: |
| School Governing Authority: | Date: |
| Chief Administrator:        | Date: |

# Attachment 4.1

# **Conflict of Interest Policy**

See Article V in the Code of Regulations for a Conflict of Interest Policy.

# **ATTACHMENT 6.3** Educational Program

- A. Curriculum and Evidence/Research of Viability of Curriculum
- B. Classroom based and non-classroom based learning opportunities include learning opportunities off site, by internet, by independent study, on contingency days, by field trip, on suspension or expulsion, etc. "Learning Opportunities" is currently defined as follows:

... classroom-based or non-classroom-based supervised instructional and educational activities which are defined in the community school's contract and are:

- (1) provided by or supervised by a licensed teacher;
- (2) goal oriented; and
- (3) certified by a licensed teacher as meeting the criteria established fee completing the learning opportunity" (emphasis added). See OAC 3301-102-02
- C. Focus, Mission, Philosophy, Goals and Objectives
- D. Instructional Delivery Methods
- E. School Calendar (including adequate contingency days)
- F. Alignment with Ohio Academic Standards
- G. Any Credit Flexibility Program

# ATTACHMENT 6.3 Educational Program

# A. Curriculum and Evidence/Research of Viability of Curriculum

# Description of Curriculum

Through Pathway School of Discovery's partnership with NHA, the school has a curriculum in place that includes well-defined learning standards, aligned with our mission and with the Ohio Academic Content Standards. To develop its curriculum, NHA studied information from various standards and assessment experts across the nation, including groups like National Assessment of Educational Progress (NAEP), ACT®, Achieve, Inc., the Thomas B. Fordham Institute, the National Council of Teachers of English, The National Mathematics Advisory Panel, the American Association for the Advancement of Science, and the National Council for Social Studies. NHA comprehensively researched what students need to know to be successfully prepared for rigorous high school and college educational programming, and then used a backward design approach to develop a curriculum that defines the expertise needed by students to be college-ready. Since college readiness goes beyond academic preparedness, the NHA Curriculum includes learning goals in core academic and co-curricular areas, and includes character development as well.<sup>1</sup> During its curriculum development process, NHA also aligned its curriculum with the Ohio Academic Content Standards, making sure that its learning objectives were consistent with those identified by the Ohio Department of Education. The NHA Curriculum has also been aligned with the standards identified through the Common Core State Standards Initiative.

# Objectives, Skills, and Content of Curriculum

### English Language Arts

Literacy, including reading and writing, is a critical component of college readiness. "Low literacy levels often prevent high school students from mastering other subjects" and struggling readers are often excluded from academically challenging courses.<sup>2</sup> Developing reading proficiency and strong literacy skills in elementary and middle grades is essential to ensuring that students are on a college-readiness trajectory. The ELA curriculum is designed to produce highly literate students who are proficient readers and strong writers. The curriculum enables students to read, comprehend, write, and respond thoughtfully to what they encounter in the classroom and the world around them through its reading, writing, speaking, listening, and viewing components.

<sup>&</sup>lt;sup>1</sup> See Conley, D. (2008, October). What Makes a Student College Ready? *Educational Leadership*, 66 (2). Retrieved from <u>http://www.ascd.org/publications/educational\_leadership/oct08/vol66/num02/toc.aspx</u>.

<sup>&</sup>lt;sup>2</sup> ACT, Inc., Reading Between the Lines: What the ACT Reveals About College Readiness in Reading (Iowa City, IA, 2006).

In early grades, the curriculum emphasizes the five components of reading instruction as outlined by the National Reading Panel, including phonemic awareness, phonics, fluency, vocabulary, and text comprehension. The curriculum also helps students develop the intellectual processes needed to remember, understand, analyze, evaluate, apply, and extend the ideas they encounter while reading. Student learning is enhanced through reading a variety of fiction and nonfiction texts, including a wide range of expository/informational and literary genres. Through writing, students learn the technical skills and processes necessary to produce different types of texts, with appropriate conventions, but also learn to write creatively and critically. And through speaking, listening, and viewing, students become collaborative and reflective in their learning processes. Students who master the ELA curriculum are prepared to read for deep meaning and understanding, write and speak effectively to communicate ideas and information while using appropriate language conventions, listen actively and critically as they encounter new information and ideas, and generate new ideas based on what they encounter both inside and outside the classroom.

Beyond this, the curriculum in grades K-2 focuses on and supports the process of learning to read. Decoding, word recognition, and building vocabulary are important as students begin to develop understanding and fluency. In grades 3-5, the curriculum supports a transition from learning how to read to reading in order to learn; learning shifts to understanding that texts have specific purposes and reading with those purposes in mind. Students are introduced to informational text in addition to a variety of literary texts. Reliance on the basal reader decreases and additional resources like novels, newspapers, magazines, and web-based resources are used to support reading instruction. In grades 6-8, the curriculum focuses on extending reading and comprehension skills, developing deep evaluation and analysis skills and the ability to make connections within and between texts.

The language arts curriculum supports learning in language conventions, mechanics, spelling, and writing. In the earliest grades, students learn how to write words and sentences using appropriate mechanics and grammar and begin to use the writing process to convey information and narrative through written text. In middle grades, the curriculum develops in students a deeper understanding of writing through a focus on prewriting strategies, organizational formats, drafting, revising, proofreading, and publishing. The curriculum ensures that students learn to write for different purposes; writing includes narratives, stories, poems, interpretive responses, essays, and descriptive pieces. In later grades, the curriculum continues to extend students' writing skills through narrative, expository, persuasive, and technical writing, and technology enhances students' ability to write, revise, edit, and publish their work. Grammar, punctuation, spelling, and writing conventions are taught as part of language arts for students in all grades.

### Mathematics

If students are to be well-equipped for college and beyond, students must be prepared through their K-8 education to take Algebra II and other advanced mathematics courses in high school.

"A strong grounding in high school mathematics through Algebra II or higher correlates powerfully with access to college, graduation from college, and earning in the top quartile of income from employment."<sup>3</sup> The development of a deep understanding of mathematical concepts makes such success possible. The school's mathematics curriculum is designed to ensure that all students complete Algebra I by eighth grade, thus placing students on a college- and career-readiness trajectory. Students who successfully complete Algebra I prior to entering high school are much more likely to complete Algebra II and other more advanced mathematics prior to graduating high school, which means they will be "more than twice as likely to graduate from college" than students who don't complete it.<sup>4</sup>

The National Council for Teachers of Mathematics has recognized the importance of the study of algebra in developing mathematical fluency and has also noted the importance of the study of other mathematics components such as number sense and number operations, measurement, geometry, data analysis and probability, and problem solving. The curriculum focuses on each of these components beginning in kindergarten. Number Sense is developed through a variety of concrete models, allowing students to use the area of the brain used for the comprehension of mathematical knowledge. Students are prepared to be fluent in computation using formal algorithms and also learn essential measurement and data analysis skills. Students also learn to make connections and apply mathematical knowledge through problem solving and inquiry.

ł

In grades K-2, algebraic awareness, number sense, and computational fluency are the main focus of students' learning. Students develop the skills necessary to progress into higher level mathematics; through open-ended problem solving, they increase their critical thinking skills and ability to see connections across mathematics as well as other subjects. In grades 3-5, learning shifts from computation to fractional awareness. The part-to-whole relationships that are developed in these early grades lead to a deeper understanding of fractions, percents, and decimals and computation. Algebraic skills move from pattern recognition and development to the creation of equations and the use of variables. As students move into grades 6-8, the focus shifts to the study of algebra and functions. Number sense remains a critical focus area through the study of integers, rational and irrational numbers, exponents, and absolute values. Conceptual ideas are integrated through lab activities that provide exploratory opportunities for students to explicitly connect abstract ideas to concrete examples.

The mathematics curriculum also ensures that students become effective as mathematical communicators by engaging them in thinking, reading, and writing about mathematics to help them understand the foundational concepts necessary for success in more complex mathematical coursework.

<sup>&</sup>lt;sup>3</sup> National Mathematics Advisory Panel, Foundations for Success: The Final Report of the National Mathematics Advisory Panel (Washington, D.C.: U.S. Department of Education, 2008).

<sup>&</sup>lt;sup>4</sup> Ibid.

### Science

As the Association for the Advancement of Science and the National Council on Science explain, developing college-ready and scientifically literate students involves teaching a mixture of content knowledge, the practices and skills of scientists, and information on the nature of science. The curriculum, which includes study in Life Science, Physical Science and Earth and Space Science, is designed to develop content knowledge about the results of scientific discoveries regarding the natural world. The teaching of practices and skills of scientists requires that students participate in the scientific process of inquiry and discovery through conducting investigations, using instruments, and applying mathematical skills that model the process used by scientists to learn about the universe. The curriculum provides students with the opportunity to participate in the scientific process, in addition to read, write, discuss and experiment with science through high level thinking and problem solving. It also ensures that students use the scientific processes and skills through lab work and investigations.

The content of the science curriculum in grades K-2 focuses scientific learning primarily through the study of events and phenomena in nature that can be observed with the five senses. Investigations at this level are modeled, simple, and structured, allowing students to write journals on their discoveries, create simple pictographs of data, and draw conclusions from observations under the direction of the teacher. The incorporation of non-fiction readers on each topic provides students the opportunity to develop grade level appropriate science vocabulary.

Building on these foundational skills, including the integration of skills from English language arts and Math, the curriculum for grades 3-5 directs students to begin exploring cause and effect connections between events (e.g., the sun rises every day and it gets warmer during the day, ergo, the sun provides heat to earth). Investigations for students in grades 3-5 are still often modeled, but they are more complex and involve more detailed measurements, use a variety of tools such as balances and microscopes, and require students to control for multiple variables. Students engage in the practices of scientists as they chart data, complete lab reports, and create graphs in order to communicate results of investigations orally and in writing while defending their conclusions.

The curriculum for students in grades 6-8 includes further exploration of cause and effect relationships while also connecting knowledge of concepts to real world examples and solidifying congruence between students' understanding of phenomena to that of the scientific community. Investigations are more student-directed, from the guiding question through the scientific procedures to the organizing, analyzing and reporting of data. Within each unit of instruction in grades 6-8, students employ the skills of scientists as they make connections with respect to the content of the unit using the methods of science. The scientific method is consistently integrated into content units throughout the year, and students conduct science investigations through both closed lab investigations and through open investigations in response to the questions being raised during content studies.

## Social Studies

The Social Studies curriculum ensures that students are not only prepared for high school and college, but also prepared for the rest of their lives as global citizens. According to ACT, developing students' understanding in the disciplines of history, culture, economics, and civics ensures that they are ready for college. In addition to supporting learning in these areas, the Social Studies curriculum also allows students to develop and utilize critical thinking skills by making connections, inferences, and arguments around the content and skills they are learning in these disciplines. This focus on content, skills, and critical thinking produces students who are knowledgeable in Social Studies and who are prepared to participate in society as informed citizens.

The curriculum emphasizes the Social Studies disciplines that best prepare students to be intelligent members of society. Students who master the Social Studies curriculum understand how geography and economics interact in a global society. Through exposure to primary and secondary sources, students develop knowledge of our shared national and world history and are able to make connections between the past and present and between cultures and government systems that are familiar and those that are not.

Specifically, the Social Studies curriculum in grades K-2 focuses on students being able to describe basic historical events, people, and conflicts and to develop an understanding of their own culture and local community. In these early grades, students begin to become exposed to the history and cultures of places around the world to expand their knowledge beyond what is in close proximity to them. By grades 3-5, the curriculum helps students describe historical happenings in more detail while allowing them to comment on how these happenings had an impact on other historical occurrences in American and world cultures. By grades 6-8, students are regularly practicing how to make connections between historical events in the United States and the world and current events, as well as developing their knowledge of the relationship between geography, history, economics, and culture.

The Social Studies curriculum also supports learning geography, economics, and civics. In the primary grades, students become familiar with basic geographical representations, economic concepts and resources, and functions of government by studying local examples of their community and school. In the intermediate grades, the curriculum gives students opportunities to extend this basic knowledge to gather more complex information, describe concepts in more detail, and deepen their understanding of the relationship between geography, economics, and civics. Third through fifth grade students extend their learning by studying these relationships through national examples. In the middle school grades, the curriculum helps the students use their knowledge of geography, economics, and civics to make generalizations about countries around the world, to compare these countries to the United States, and to describe how these disciplines work together in a global society.

# Character Development

Because teaching virtues is integral to the development of children and to preparedness for high school and college, the curriculum includes explicit, integrated character development. This component of the curriculum, which we call Moral Focus, identifies the skills, behaviors, character traits, and virtues that students need to apply their academic learning as citizens and to be successful learners in high school and college. A comprehensive character development approach, drawing from both the *Smart & Good Schools Initiative*<sup>5</sup> and the Greek Cardinal Virtues, is focused through the study of Moral Character, Performance Character, and Social Character.

Moral Character guides students towards becoming their best self. Through the curriculum, students learn that character is defined by the virtues that they incorporate as a consistent part of their behavior. They also learn that they, as individuals, bear the responsibility for developing these virtues. This aspect of the curriculum consists of a school wide focus around nine monthly virtues, which were derived from the Greek Cardinal Virtues. Students develop these virtues through explicit teaching and integrated practice until they become settled habits. Each monthly virtue is comprised of three focus virtues that build from kindergarten to eighth grade. This allows students to acquire a deep understanding of each virtue. Each monthly focus virtue within the curriculum is clearly defined within the curriculum and includes expected student behavior as it relates to that virtue.

Performance Character helps students grow in their ability to do their best work. Through the study of performance character, students learn that effort creates ability and that achievement must be earned. Students learn to strive for excellence in all that they do, work hard to overcome obstacles, find joy in a job well done, and develop the work and study habits needed to succeed in school and life. Students develop the skills to take ownership of their learning, and they acquire the tools needed to be able to define, prioritize, and complete tasks independently and utilize their time efficiently. In addition to these academic strategies, students learn to set goals for improvement and to monitor their progress in meeting those goals.

Social Character focuses on the skills students need to interact best with others. The social character aspect of the character development curriculum enhances students' ability to listen and be understood by others. Students learn to identify and manage their emotions in order to regulate their behavior and make wise choices. They consider possible outcomes, alternatives, and consequences to their choices with the understanding that they alone are responsible for their actions. Students also acquire the tools needed to have effective and lasting interpersonal relationships with others.

<sup>&</sup>lt;sup>5</sup> See Matthew Davidson and Thomas Lickona, *Smart & Good High Schools: Integrating Excellence and Ethics for Success in School, Work, and Beyond* (Cortland, NY: Center for the 4<sup>th</sup> and 5<sup>th</sup> Rs, 2005). Respect and Responsibility / Washington D.C.: Character Education Partnership.
Students learn to internalize these concepts through the intentional and consistent focus on the application of character development concepts throughout all aspects of school life. The character development curriculum is not another subject to be taught during the school day. It defines the character traits and behaviors that need to be consistently and intentionally modeled for and exhibited by students at all times throughout the day; complete integration is thus essential for students to learn the importance of developing and maintaining a strong personal character and the qualities necessary for success as both a student and a citizen. Through consistent modeling and integration of the virtues and behaviors identified in the character development curriculum, staff and students create a learning environment built on a foundation of respect and care where everyone works hard to achieve academic goals and improve school culture.

### Art

Study in Art allows students to explore the many different aspects of this subject while acquiring an understanding of the significant role art has played and plays in the expression of ideas throughout history and the current age. Students in lower grades learn to recognize and describe art forms from a variety of historical eras and places, and they also learn that art takes many forms and has many purposes. Students study and reproduce styles and techniques used by artists and discuss the qualities of art by examining that of artists and critiquing their own work. Students in upper elementary and middle school focus on comparing and contrasting artwork from different time periods, cultures, artists and mediums. Historical and contemporary art are studied as students create a variety of original works using appropriate styles and mediums to express and communicate themes, tone, mood, and images.

### Music

In music, students study the basics of music theory, music appreciation, and great composers and their works. Students first learn to identify basic music forms and patterns, to describe the elements of music using appropriate music vocabulary, and to sing and play basic classroom instruments. By listening to a variety of music, students also compare and contrast music from different composers, historical periods, cultures, styles, and genres. Students continue to refine their musical knowledge and skills through singing, playing, improvising, and composing. They learn to identify and use key signature, meter signature, notation, bass and treble clef, tempo, and dynamic markings and they also learn to perform and respond to the cues of the conductor and make necessary adjustments. Students also continue to listen to and evaluate many types of music from historical eras, genres, composers, and cultures and describe how music communicates mood, tone, and images.

## Physical Education

Physical education includes a sequence of developmental experiences whereby children learn through movement. Students initially learn and practice basic non-locomotor skills and locomotor skills, learn to manipulate objects through throwing, catching, striking, pushing, pulling, and climbing, and take part in a variety of individual and group activities and games in order to increase their body awareness, practice new skills, and learn to move safely with respect to other people. Students also begin to learn about the physical and mental benefits of a healthy lifestyle that is characterized by physical activity. Students then use locomotor skills in more complex ways and the concepts of fairness, positive attitude, teamwork, and sportsmanship are learned. Students learn to recognize the correlation between practice and mastery of skill and complete various types of drills and routines in order to increase their own abilities. Students ultimately refine all the simple and complex skills necessary for physical activity of various types and focus on healthy lifestyles through nutrition and fitness.

## Library and Technology

The Library and Educational Technology Program is designed to expose students to a wide variety of classic and contemporary literature, to instill a life-long love of reading, and to develop their information literacy skills. The library and educational technology collections contain specific materials that support the curriculum and provide students with quality literature and a variety of informational and technological resources. The Library and Educational Technology Program is built on the expectations for students derived from the *Big6* model for teaching information skills created by Mike Eisenberg and Bob Berkowitz (1998) and Information Literacy Standards developed by the American Association of School Librarians (AASL) and the Association for Educational Communications and Technology (AECT). Technology has a supporting role in the context of the core academic curriculum. The school's Library staff works with classroom teachers to help plan and integrate lessons and provide resources, and students use technology to research, compose, and present information related to topics studied in other content area classes.

## Evidence/Research of Viability of Curriculum

Schools utilizing the NHA Curriculum have a track-record of academic success with students, regardless of socio-economic or other factors. This track-record has been achieved through topquartile academic growth rates that lead ultimately to higher proficiency rates. Evidence that demonstrates the effectiveness of NHA's educational approach is notable.

Following are a number of high-level statistics for NHA partner schools:

- State test results for the 2010-11 school year revealed that NHA partner schools have a higher rate of proficiency than their respective local school district 74% of the time.
- NHA partner schools' fall-to-spring growth on nationally-normed assessments places it at the 75th percentile nationally over the course of the last five years. This academic growth accomplishment is attained regardless of the student demographic and without consideration of the demographic of the national sample used.

- NHA partner school's rate of growth for students below grade level for the 2010-11 school year was 139% as compared to the national average of 100%.
- With over 45,000 students and over 60% qualifying for free and reduced-price lunch, NHA partner schools collectively would rank as one of the 100 largest urban school districts in the country.

NHA partner schools have established a commendable academic record serving students who are the most likely to be in academic risk, thereby establishing the viability of the curriculum.

B. Classroom based and non-classroom based learning opportunities – include learning opportunities off site, by internet, by independent study, on contingency days, by field trip, on suspension or expulsion, etc. "Learning Opportunities" is currently defined as follows:

... classroom-based or non-classroom-based supervised instructional and educational activities which are defined in the community school's contract and are:

- (1) provided by or supervised by a licensed teacher;
- (2) goal oriented; and
- (3) certified by a licensed teacher as meeting the criteria established fee completing the learning opportunity" (emphasis added). See OAC 3301-102-02

Description of classroom-based learning opportunities are described in sections A, C, and D.

## C. Focus, Mission, Philosophy, Goals and Objectives

## <u>Focus</u>

A number of core beliefs stand together to provide the focus for Pathway School of Discovery's educational program. These core beliefs are the foundation upon which the school's educational program has been built and serve as guide-posts for the school as it works to prepare students for high school, college, and life success.

## K-8 School Design

If students are to be adequately prepared for rigorous high school programming that leads to college readiness, then their elementary and middle school educational opportunities and experiences are critically important. Our K-8 school design is central to ensuring a significant early investment in the lives of the students we serve.

Research shows that learning during the formative years of students' lives is foundational to their future academic success.<sup>6</sup> A student's reading skill at the end of third grade, for example, is a reasonably accurate predictor of whether or not that student will graduate from high school.<sup>7</sup> Early performance is important in all academic areas, and high-quality schooling across core academic content areas in the primary years is absolutely essential for later success in school and life. Studies by ACT note specifically that "the level of academic achievement that students attain by eighth grade has a larger impact on their college and career readiness . . . than anything that happens academically in high school."<sup>8</sup> The academic preparedness of students upon entry to high school greatly affects their subsequent attainment: 82% of students who are placed in their high school's degree, compared to only 9% of those who are placed in their high school's least intense curriculum.<sup>9</sup> Our focus on providing exceptional learning opportunities for students in grades K-8 ensures that Pathway School of Discovery students receive the high-quality education they deserve and that they are well prepared for rigorous high school study, leading to college readiness and life success.

The K-8 school design is not just beneficial to students academically, but prepares them to achieve success socially and emotionally as well.<sup>10</sup> Some suggest that the K-8 model is a more effective approach than the standard elementary, middle, and high school models that many schools utilize. The transition to middle school can be highly challenging for many students at a time in their lives when they may not yet be prepared to cope with the increased responsibility and social pressures of middle school. This can result in regression on academic, emotional, and social fronts as students seek to find their footing in a new and unknown environment. A consistent K-8 schooling experience, with less transition and greater stability, serves to ensure that students develop appropriately during their middle school, Pathway School of Discovery not only provides students with the foundation they need to be successful academically, but also provides its students with a greater sense of stability, security, and community as they traverse their important adolescent years.

<sup>&</sup>lt;sup>6</sup> George Kuh, Jillian Kinzie, Jennifer Buckley, Brian Bridges, & John Hayek, *What Matters to Student Success: A Review of the Literature*. (National Postsecondary Education Cooperative, 2006).

<sup>&</sup>lt;sup>7</sup> National Research Council, *Preventing Reading Difficulties in Young Children*, eds. Catherine E. Snow, Susan Burns, and Peg Griffin, Committee on the Prevention of Reading Difficulties in Young Children (Washington, DC: National Academy Press, 1998).

<sup>&</sup>lt;sup>8</sup> ACT, The Forgotten Middle: Ensuring that All Students Are on Target for College and Career Readiness before High School (Iowa City, IA: ACT, 2008).

<sup>&</sup>lt;sup>9</sup> Clifford Adelman, *The Toolbox Revisited: Paths to Degree Completion from High School Through College* (Washington DC: U.S. Department of Education, 2006).

<sup>&</sup>lt;sup>10</sup> Pricilla Pardini, "Revival of the K-8 School: Criticism of Middle Schools Fuels Renewed Interest in a School Configuration of Yesteryear," *School Administrator*, March 1, 2002; Anne Marie Chaker, "Middle School Goes out of Fashion: Amid Evidence Kids Struggle to Move to Junior High, Districts Shift to K-8 Model," *The Wall Street Journal*, April 6, 2005; Marc S. Tucker and Judy B. Codding, *Standards for our Schools: How to Set Them, Measure Them, and Reach Them* (San Francisco, CA: Jossey Bass, 1998).

### Character Development

Teaching virtues is not optional in a school; it is inevitable. In effect, a school teaches virtues in everything it does-from the way students are treated by teachers and staff, to the way the principal treats teachers, to how students treat and are allowed to treat their peers. The virtues that students encounter within the four walls of the school have a significant effect on their development. Because of this, the teaching of virtues at Pathway School of Discovery is an explicit and integrated component of the school's educational program.<sup>11</sup>

A school's culture and student behavior directly correlate with student achievement. "Most [children] have to pay attention, study the material, and do their homework carefully if they are to achieve in high school. These behaviors don't come naturally. They have to be learned and practiced and gradually integrated into a person's character."<sup>12</sup> Research shows that the teaching of virtues is helpful not only in terms of promoting students' academic success, but also in terms of promoting a school climate and culture that are highly conducive to teaching and learning. Teaching virtues at Pathway School of Discovery helps the school ensure that each student is cared for as a family cares for its children and that each student is prepared for rigorous high school and college educational experiences.

## *Learning is an Adult Responsibility*

Student learning at Pathway School of Discovery is, first and foremost, an adult responsibility. Nothing has a greater impact on student learning than great teaching in every classroom. If a student does not master the material that is taught in a classroom, the student has not failed; rather, the school has failed the student. In cases where students do not learn as expected, instruction must be adjusted in order to ensure that appropriate learning takes place for all students. Research has shown that the quality of instruction strongly predicts the level of learning in a classroom—more strongly than does a student's race or income.<sup>13</sup>

A typical student who scores at the 50<sup>th</sup> percentile in mathematics is likely to continue scoring at the 50<sup>th</sup> percentile two years later, if he or she has an average teacher and attends an average school. However, that same student-with an effective teacher in an effective school-would rise to the 96<sup>th</sup> percentile within that same two-year period. Conversely, an average student would decline to the 3<sup>rd</sup> percentile over that time period with an ineffective teacher.<sup>14</sup> A vast

<sup>13</sup> Linda Darling-Hammond, "Teacher Quality and Student Achievement: A Review of State Policy Evidence," Center for the Study of Teaching and Policy, University of Washington, 1999, Available from http://www.politicalscience.uncc.edu/godwink/PPOL8687/WK11March%2029%20Teachers/Darling-

<sup>&</sup>lt;sup>11</sup> See Section A for details pertaining to the school's character development program, including its Moral Focus curriculum.

<sup>&</sup>lt;sup>12</sup> Kevin Ryan, "Character Education: Our High Schools' Missing Link," Education Week, January 29, 2003.

Hammond%20Review%20essay%20on%20teacher%20quality%20and%20outcomes.pdf. <sup>14</sup> Kirsten Miller, "School, Teacher, and Leadership Impacts on Student Achievement," *Policy Brief*, Mid-continent Research for Education and Learning, November 2003 available from

body of research supports the finding that students' learning is influenced by the content knowledge, pedagogical knowledge, and classroom management skills of their teachers. As such, the school works to manage instruction to ensure that students receive consistent, high-quality instruction during their time at Pathway School of Discovery.

We believe that having one principal manage nearly every employee in the school does not set our school up for success. "One of the common misconceptions about leadership at the school level is that it should reside with a single individual—namely the principal."<sup>15</sup> In order to create a different kind of school culture, both a principal and other school leaders are needed. A "strong leadership team," including "the principal and other administrators operating as key players and working with a dedicated group of classroom teachers" is essential to the success of the school.<sup>16</sup> In order to effectively manage instruction, we have adopted a distributed leadership model through which the principal is supported by three deans. This model allows teachers to receive frequent classroom observation and feedback from school leaders.

Quality instruction is the result of clear expectations for teachers, frequent observation, consistent feedback and coaching, and an intentional focus on developing professional practices through ongoing learning and coaching and the collaborative work of a professional learning community. Pathway School of Discovery has implemented a system to hire the most qualified teachers, provide high-quality professional development opportunities, hold teachers accountable for academic results, and support them through weekly classroom observations and feedback. In this way, the school effectively ensures appropriate learning for all students.

## College Readiness

If we are to accomplish our mission, students must leave Pathway School of Discovery after 8<sup>th</sup> Grade well-prepared for success in high school, ready to enroll in rigorous high school programming that continues to prepare them for college and career placement. Students must also understand, however, that success in high school, college, and life is not only about academic aptitude; other factors affect success as well. Indeed, both academic and psychosocial factors have a bearing on how students succeed in school and in life.

The school's educational program includes a focus on developing college readiness in all students. Pathway School of Discovery promotes college readiness through a focus on both academic and psychosocial factors that influence school and life success: students develop key academic skills and strategies and master content knowledge necessary to succeed in high school and beyond; they develop behavioral, attitudinal, and perceptual attributes correlated with

http://www.mcrel.org/PDF/PolicyBriefs/5032PI\_PBSchoolTeacherLeaderBrief.pdf; Robert Marzano, What Works in Schools: Translating Research into Action (Alexandria, VA: ASCD, 2003).

<sup>&</sup>lt;sup>15</sup> Robert Marzano, *What Works in Schools: Translating Research into Action* (Alexandria, VA: ASCD, 2003). <sup>16</sup> *Ibid.* 

college and life success; and they develop contextual knowledge about high-school and college placement.<sup>17</sup>

These core beliefs, taken together, articulate the focus of the school and are the foundation of the school. Through an intentional focus on implementing an educational program designed with these beliefs in view, we will continue meeting the needs and expectations of the families and students who choose Pathway School of Discovery.

## <u>Mission</u>

Working in partnership with parents and community, the Pathway School of Discovery mission is to become one of the finest K-8 schools in the country. We offer a challenging, characterbased education through a rigorous curriculum with high academic and social expectations. Our school prioritizes the academic program and instructional time so that each student reads, computes, and writes at or above grade level. We expect our students to master basic skills and realize their full academic potential in preparation for higher education and adulthood.

## Educational Philosophy

"Schools can have a tremendous impact on student achievement if they follow the direction provided by the research," writes nationally-recognized education researcher Robert Marzano.<sup>18</sup> Marzano identifies a number of school-level and teacher-level factors—akin to those identified through Effective Schools research—that have a high-probability of increasing student learning. Pathway School of Discovery's educational philosophy is grounded in those methodologies that research has shown to be effective in increasing student achievement.

As a result of its partnership with NHA, Pathway School of Discovery benefits from a systemic implementation of school- and teacher-level factors that have been shown to produce positive results in terms of student learning. School-level factors include a guaranteed and viable curriculum, challenging goals and effective feedback for students, parent and community involvement, safe and orderly environment, and collegiality and professionalism. Teacher-level factors include instructional strategies, classroom management, and classroom curricular design. By implementing an educational program designed to include these strategies, which have the greatest positive impact on student learning, Pathway School of Discovery ensures strong academic performance from its students.

Pathway School of Discovery believes that great schools are synonymous with great instruction, and great instruction is a function of the right people, doing the right things the right way. We are intentional in our efforts to attract, develop, and retain exceptional talent; talented people

<sup>&</sup>lt;sup>17</sup> Adapted from David Conley, *Toward a More Comprehensive Definition of College Readiness* (Eugene, WA: EPIC, 2007).

<sup>&</sup>lt;sup>18</sup> Marzano, R. (2003). What works in schools: Translating research into action. Alexandria, VA: ASCD.

operate, in turn, within a system that is designed to promote student success at the highest possible level.

With the knowledge that the greatest impact on student learning is the quality of instruction, each part of the educational program is designed to create the norms and conditions for exemplary instruction for each student. In order to know if we are fulfilling our responsibility of student learning, we measure results at the student, classroom, teacher, and school level. These results inform our decisions from accountability to intervention. With data to inform decisions, relationships with students and amongst adults enable us to act on evidence. An intentional school climate and culture is established where all in the building behave with care for each student, as a family does for its children, with the goal of self-managed classrooms.

Time is a critical resource for learning. Given this, classroom and school schedules reflect the principle that we need to calendarize priorities: math and English language arts are scheduled in the morning; science and social studies are a focused priority; joint planning is scheduled to organize the implementation of the curriculum; teachers collaboratively create, share, and refine lessons and units of instruction, and examine student work as evidence of student learning together in learning communities. A distributed and shared leadership structure allows school leaders to manage instruction by observing and meeting one-on-one with the teachers they manage on a weekly, scheduled basis, in order to provide feedback and coaching.

The curriculum is deliberate in its college-preparatory design by defining essential learning goals for all students. Since college and life success require more than just aptitude, we teach virtues as an integrated but explicit part of the curriculum. The formative assessment process, which is a planned process through which teachers and students use frequent, varying, and ongoing assessment-based evidence to adjust what they are doing, is utilized to implement and assess the intended curriculum. Additionally, the school utilizes common curricular materials that effectively support instruction and are aligned with the curriculum.

A shared expectation across the school is that classrooms exemplify engagement, clarity of instructional intent, and rigor. This is accomplished through effective instruction, classroom management, and classroom curriculum design techniques. Further, teachers, staff, and school leaders work to engage parents by initiating a positive relationship between the school and home. In short, the school is designed to be a place where best practices become common practice.

### Goals and Objectives

Pathway School of Discovery's academic goals can be found in Attachment 6.4. The educational goals and objectives are further detailed in the attached curriculum CD.

## D. Instructional Delivery Methods

While the curriculum identifies and communicates what is essential for students to know and be able to do, classroom instruction drives student engagement with the curriculum. As such, instructional methods and strategies employed by teachers at Pathway School of Discovery are designed to promote engagement, to include clarity of instructional intent, and to be appropriately rigorous. Instruction takes various forms, including focused lessons, guided instruction, collaborative learning, and independent work.<sup>19</sup> The best approach is contextually-driven and contingent upon students' progress toward meeting instructional goals.

Focused lessons, when employed, may include direct instruction, modeling, demonstration, development of metacognitive awareness, and/or teacher think-alouds, among other methods, and they may be conducted in whole-group or small-group settings as student need dictates. Guided instruction may be done with large groups or with small, needs-based groups during workshop time or during content-area learning blocks. Guided instruction, when provided in small-group settings, allows teachers to differentiate instruction in purposeful, meaningful ways in order to meet student needs. Collaborative learning may be used in order to provide opportunities for students to learn from peers, to discuss or debate ideas and information, or participate in collaborative, inquiry-based learning of curricular content. Finally, independent work is sometimes used to provide students with time to practice applying skills or knowledge that has been previously learned, to extend individual learning through the application of knowledge and skills to novel situations, or to promote the individual development of higher order thinking skills.

Beyond this, pedagogy at Pathway School of Discovery includes effective, research-based instructional strategies embedded within daily lesson plans and implemented during instruction. Strategies that have high probabilities of effectiveness are utilized based on the expected student learning outcomes of the lesson and are employed by teachers in accordance with student need. Such instructional strategies include the following, which have been identified as highly effective by researchers at Mid-continent Research for Education and Learning (McREL): (1) Identifying Similarities and Differences; (2) Summarizing and Note Taking; (3) Reinforcing Effort and Providing Recognition; (4) Homework and Practice; (5) Nonlinguistic Representations; (6) Cooperative Learning; (7) Setting Objectives and Providing Feedback; (8) Generating and Testing Hypotheses; and (9) Cues, Questions, and Advanced Organizers.<sup>20</sup> Teachers at Pathway School of Discovery integrate these methods into their instruction as appropriate, based both on student need and what research identifies as the most effective approach for content delivery.

The instructional approach employed by teachers at Pathway School of Discovery ensures that

<sup>&</sup>lt;sup>19</sup> Douglas Fisher and Nancy Frey, Better Learning through Structured Teaching (Alexandria, VA: ASCD, 2008).

<sup>&</sup>lt;sup>20</sup> See Robert Marzano, Debra Pickering, and Jane Pollock, *Classroom Instruction that Works: Research-Based* 

Strategies for Increasing Student Achievement (Upper Saddle River, NJ: Pearson Education, Inc., 2001).

students master the essential learning goals articulated by the curriculum and develop college readiness in accordance with the school's educational design and the aligned standards adopted by the Ohio Department of Education. As teachers plan and implement instruction using these instructional methods and strategies, students gain the knowledge, proficiency, and skills needed to perform at high levels.

Because effective instruction is the result of intentional planning, instruction at Pathway School of Discovery is driven by a unique and collaborative planning approach that includes the implementation of a formative assessment process. This approach to planning is characterized by three sequential but interrelated steps: year-long planning, unit planning, and daily planning.

## Year-long Planning

Teachers collaborate within grade level team members in order to develop an instructional pacing framework each school year. Through this process of year-long planning, teachers sequence all Educational Objectives and assessment opportunities within each content area for a specific grade level and are organized by quarter or month. Year-long plans are adjusted frequently over the course of the year, as student need dictates, and serve as the outline by which teachers develop more detailed unit plans, organize daily instruction, and develop year-long plans for subsequent school years.

Teachers and their grade level teams use multiple sources of data, such as formative assessment data, state OAA assessment data, Northwest Evaluation Association (NWEA) Measures of Academic Progress (MAP) assessment data, and other assessment data to identify instructional areas for focus and improvement.

## Unit Planning

During unit-level planning, teachers collaborate to develop units of instruction for each Educational Objective and to determine opportunities to formatively assess student learning during the instructional process. As they develop unit plans, teachers develop end-of-instruction assessments and plan for opportunities to make adjustments to instruction based on student learning progress.

Teachers use a formative assessment process to implement the curriculum. Although various definitions and practices exist with respect to formative assessment, we have adopted W. James Popham's definition. He says, "Formative assessment is a planned process in which assessment-elicited evidence of students' status is used by teachers to adjust their ongoing instructional procedures or by students to adjust their current learning tactics." Popham identifies a number of characteristics of formative assessment, noting that it:

- Is a process, not any particular test.
- Is used not only by teachers, but also by students.

- Takes place during the instructional process.
- Provides assessment-based feedback to teachers and students.
- Helps teachers and students make adjustments that will improve students' achievement of intended educational objectives.<sup>21</sup>

The components of the formative assessment process are: 1) Identifying Objectives and Determining End-of-Instruction Assessments; 2) Developing Building Blocks; 3) Analyzing Evidence; and 4) Responding to Evidence. These are detailed below.

## Identifying Objectives and Determining End-of-Instruction Assessments

During the formative assessment process, teachers first collaborate to identify objectives for instruction based on the year-long plan. Then teachers develop end-of-instruction assessments to determine students' status at the end of each unit. Assessment methods are selected based on the learning goal and sound assessment design. This "assessment-influenced" approach is important, for through it teachers "exemplify the curricular aim or aims being sought and ultimately decide what mastery of those aims looks like—what its mastery calls for a student to be able to do."<sup>22</sup>

## Developing Building Blocks

After the Educational Objectives are reviewed and appropriate end-of-instruction assessments are determined, grade-level teams identify building blocks of instruction that scaffold student learning toward mastery of each Educational Objective. These building blocks are used to construct a variety of formative assessments that are woven into daily instruction and serve as check points during the learning process. Grade level teams utilize the building block assessments across their classrooms. The school's electronic gradebook, provided within AtSchool®, allow teachers to record student scores on these assessments. Both teachers and students track their progress over time.

## Analyzing Evidence

Whether gathered through building block assessments or end-of-instruction assessments, teachers analyze evidence at the student and classroom level. The specificity of these assessments enables teachers to determine exactly what knowledge or skills need further development for particular students. As teachers meet to examine student work, they gauge the effectiveness of their instruction and determine students' degree of mastery of the Educational Objectives. They analyze evidence to discover student misconceptions and to identify knowledge or skill gaps that may inhibit success. Grade-level teams also analyze end-of-instruction assessments to ensure that the curriculum is coherent across classrooms within the school.

<sup>&</sup>lt;sup>21</sup> Popham, W. James. (2008). Transformative assessment. Alexandria, VA: ASCD.

<sup>&</sup>lt;sup>22</sup> Popham, W. James. (2009). Instruction that measures up. Alexandria, VA: ASCD.

### Responding to Evidence

As units are prepared, teachers identify opportunities for planned instructional adjustments in case students do not learn as expected. Teachers may then make these adjustments during the course of unit instruction, as prompted by evidence from the building block assessments or summative assessments. If learning progresses more quickly than expected, then instruction moves at an accelerated rate through the unit plan. However, if learning progresses more slowly than expected, then more time is spent delivering instruction within the unit. Because students track their own progress, they know precisely what they need to focus on to improve their learning. Teachers consider end-of-instruction assessment evidence as they develop subsequent units of instruction.

## Daily Planning

Daily planning allows teachers to connect Educational Objectives with instructional resources, effective teaching strategies, and instructional methods that best support student learning. Daily planning includes developing strategies to check for understanding on an ongoing basis and determining if learning activities are provided intellectual engagement for each student. Teachers work backward from the unit plan to structure instruction in the way that best leads to mastery of Educational Objectives. In addition, the teacher's daily instructional objectives are presented in student-friendly language through "I Can" statements. These "I Can" statements are incorporated into lessons so that students know precisely what their learning goal is and can articulate it in their own terms.

## E. School Calendar (including adequate contingency days)

On Following Page



## **ATTACHMENT 6.4**

Performance Standards, Testing and Assessments, Goals, Initial and Yearly Performance Benchmarking

(Also, address attendance rates, teacher turnover, student re-enrollment, parent satisfaction, success of mission driven programs (such as Career Technical, or otherwise) special education program and compliance, assessments (internal and state required), performance index scores, Adequate Yearly Progress-, graduation rates and graduation tests, and any internal goals and missions affecting educational performance and assessments.)

## **ATTACHMENT 6.4, continued**

Sponsor's Monitoring of Assessments and Goals shall include but not be limited to the following:

- 1. If the School drops one ranking level in Ohio Report Card standing, the Sponsor may, at its discretion, require an explanation, analyses or corrective action plans concerning a specific area or areas causing the drop in ranking.
- 2. If the School drops two or more ranking levels in Ohio Report Card standing, or, drops to the lowest report card ranking, the Sponsor may, at its discretion, require a targeted, detailed corrective action plan, require outside consultants or counseling, require meetings with the administration or Board of Directors to identify issues, or any other means of inducing academic improvement.

The above monitoring is not a pre-condition to any disciplinary process allowed by law, is not a mandatory Sponsor responsibility, and is in addition to any and all disciplinary procedures allowed by law.

#### ATTACHMENT 6.4

Performance Standards, Testing and Assessments, Goals, Initial and Yearly Performance Benchmarking

- 1. Each year, the School shall pursue achieving 75% proficiency or above on Ohio's Annual Achievement Assessment on tested subjects in grades 3-8 in accordance with Ohio law.
- 2. Each year, the School shall obtain a designation of "continuous improvement" by obtaining a performance index score of at least 80.0 or meeting AYP.
- Each year, the percent of students enrolled in the School performing at or above proficient on the Ohio Achievement Assessment (OAA) in each tested grade will, in 50% or more of the grades, exceed the average performance of students tested in the same grades and subjects in the Local School District.
- 4. Each year, the School will be in the 50<sup>th</sup> percentile or higher of public schools nationally as measured by the percentage of students that meet or exceed beginning of year to end of year MAP growth on the NWEA mathematics, reading and language usage.
- 5. Each year, the School shall pursue meeting or exceeding the 93% attendance rate in accordance with Ohio law.
- 6. Over the term of the current contract, average yearly attrition will not exceed 25%.
- 7. Each year, the School shall pursue meeting or exceeding a net promoter score of 60% on parent satisfaction surveys. Net promoter score is based on the likelihood of a parent to recommend the school to others.
- 8. The School will retain 70% of teachers year over year excluding those teachers who leave for medical, maternity, relocation, retirement or performance reasons.

## **ATTACHMENT 6.5**

Commitment to Racial and Ethnic Balance (Plan to Achieve and Continue)

#### Attachment 6.5

#### I. Commitment to Racial and Ethnic Balance

The School will serve general population students and will not discriminate in its enrollment or admission procedures.

Our admission policy promotes the inclusion of all students, regardless of race, ethnicity, gender, learning ability or physical disability. It is our goal that the school's student population demographically reflects the community's population. We will market to the community surrounding the school to the fullest extent possible to support this goal.

#### II. Plan to Achieve or Continue

The school will use a variety of methods to promote a racial and ethnic balance that reflects the population of the surrounding area. Our marketing plan will include:

- 1. Demographics of the area
- 2. Population counts in the target location
- 3. List of all radio stations, including those serving ethnic groups
- 4. List of all newspapers, including community papers
- 5. List of community organizations (churches, boy's and girl's clubs, local businesses, child care, libraries and organizations serving ethnic groups)

The school's marketing team will develop the following relationships:

- A. Contact real estate offices to set up presentations and provide them with brochures
- B. Contact child care centers and pre-schools, develop relationships with director and promote Kindergarten field trips and Kindergarten workshops
- C. Distribute flyers to locations where families or parents congregate
- D. Contact "welcome wagon" organizations
- E. Contact recreation leagues for children (i.e. soccer clubs, gymnastics, etc.)
- F. Contact organization serving ethnic populations
- G. Investigate advertising opportunities
- H. Contact all parents who have indicated an interest in the school
- I. Involvement in and presentations to service clubs

## **ATTACHMENT 6.7**

Suspension, Expulsion, Permanent Exclusion Policy Due Process Procedures Policy for Discipline, Suspension, Expulsion of Disabled Students

# Student Responsibility: Student Code of Conduct

Our goal is to offer a quality education program. To fulfill this goal, students need a positive, safe, and orderly school environment in which learning can take place without disruption. Students who do not observe the rules of good conduct in the classroom or on the playground decrease both the learning and safety of others and their own opportunities to learn. Therefore, our staff takes a very proactive role in enforcing the Student Code of Conduct outlined below.

Students are expected to demonstrate respect and courtesy by obeying staff members, being kind to others, and being considerate of others' and the school's property.

#### Acts of Misconduct

The acts of misconduct listed in this Student Code of Conduct are not to be construed as an allinclusive list or as a limitation upon the authority of school officials to deal appropriately with other types of conduct which interfere with the good order of the school, the proper functioning of the educational process, or the health and safety of students.

The following list includes examples of prohibited student conduct. Parents or students who are unsure of what conduct is prohibited by each act should consult with the principal. This list is not all-inclusive. Additionally, a student who engages in an act of misconduct that violates the law may be referred to law enforcement.

Students are expected to follow the Code of Conduct when

- on school property;
- in a motor vehicle being used for a school related purpose;
- at a school-related activity, function or event;
- en route to or from school;
- at any time or place when the student's behavior has a direct and immediate effect on maintaining order, safety, health and discipline in the school

Acts of misconduct include, but are not limited to, the following:

- Failing to cooperate with or persistently disobeying staff members or volunteers
- Improperly communicating with staff members, volunteers, or students
- Use of profane and/or inappropriate language
- Behaving or dressing indecently (including public displays of affection having sexual connotations)
- Disrupting school or violating building rules and regulations
- Fighting with, assaulting, verbally assaulting, harassing, or cyber-bullying others
- Violating the Technology Use and Internet Safety Policy
- Violations of rules or policies as set forth in the Parent and Student Handbook
- Committing arson
- Committing criminal sexual conduct
- Acts committed while enrolled at another school\*
- Performing coercion, extortion, or blackmail
- Trespassing or loitering
- Visiting school property or attending school activities while suspended or expelled

- Making false alarms
- Making false allegations against staff, volunteers, or students
- · Falsifying records, being dishonest, cheating, and plagiarizing
- Misusing copyrighted material
- Being excessively tardy or absent
- Smoking or using tobacco products
- Damaging or stealing property or possessing stolen property
- Possessing electronic communication devices, laser pointers, fireworks, explosives, chemical substances, alcohol, drugs, narcotic drugs, counterfeit substances, weapons, dangerous instruments, look-a-like weapons, and personal protection devices (e.g., tasers, mace, and pepper-spray)
- Committing other criminal acts not included here
- Harassment, intimidation, or bullying including behavior or gestures which symbolize gang membership or causing and/or participating in activities which intimidate or affect the attendance of another student.
- Gang activity: means any conduct engaged in by a student on behalf of a gang, to perpetuate the existence of a gang, to affect the common purpose or design of a gang, or to represent gang affiliation, loyalty or membership (also see the school's Dress Code). These activities include recruiting, threatening or intimidating other students or staff in furtherance of a gang's purpose or activities. Gang means any organization, association, or group of three or more persons who one of its activities being the commission of criminal acts and which has an identifying name or sign or symbol.

\*If your child has been suspended or expelled from another Ohio school, your child will be offered an opportunity for an informal hearing before the Principal to determine whether the school will honor the suspension or expulsion as initially imposed by the other Ohio school. If the school chooses to honor the disciplinary removal imposed by the other school then your child will be temporarily denied admittance to the school until after the suspension or expulsion expires.

If your child has been expelled or otherwise removed for disciplinary purposes from a public school in another state, your child will be offered an opportunity for an informal hearing before the Principal. Based on the outcome of the hearing, your child may be denied admittance to the school for the shorter of the period of the removal or the period of removal which would have been applied had the student committed the same offense in the school.

## Disciplinary Procedures

A student may be disciplined at any level depending upon the frequency and/or severity of the act of misconduct. A Behavior Referral Report will be completed for each violation of the Code of Conduct and copies are made for the parent and kept in the student's file. Discipline records will be included in any student file properly requested by the parent to be transferred to a subsequent school. Corporal punishment is prohibited as a means of discipline.

Level 1 <u>EARLY INTERVENTION</u>: The behavior may be a violation of the code of conduct or a disruption of the orderly operation of the classroom or a school activity. The inappropriate behavior with the student is addressed.

Level 2 <u>PARENT NOTIFICATION</u>: The parent is notified by mail or a note home of the misbehavior.

Level 3 <u>PARENT CONTACT</u>: A conference with the parent is held in order to discuss the incident and appropriate disciplinary action.

Level 4 <u>BEHAVIOR CORRECTIVE ACTION PLAN (B-CAP)</u>: When the behavior(s) has reached a level of disruption of the academic process a conference with the parents, teacher and/or principal will be held in order to implement a B-CAP. A B-CAP will take into consideration the cause of the inappropriate behavior, positive interventions that might be utilized to diminish the inappropriate behavior and necessary consequences that will take place if the behavior continues. The B-CAP shall be signed by all parties and copies are made for the parent and kept in the student's file. If the disciplinary action is related to unexcused tardies or absences an Attendance Corrective Action Plan (A-CAP) will be implemented.

Level 5 <u>SUSPENSION OF 10 SCHOOL DAYS OR LESS OR OTHER DISCIPLINARY INTERVENTIONS</u>: When the act of misconduct disrupts the school environment, the resulting consequence issued by the school administration may include suspension of up to 10 school days, or other disciplinary interventions such as restitution, counseling and exclusion from school activities. The school administration may convene a meeting with the student, parents and others to develop or update an existing Behavior Contract that outlines the expected behavior and disciplinary action. The Behavior Contract shall be signed by all parties and copies are made for the parent and kept in the student's file. The student will be granted a right to due process as described in the Due Process Procedures of this Code of Conduct.

#### Level 6 EXPULSION:

FOR 20 SCHOOL DAYS OR LESS: When the student consistently violates the acts of misconducts by serving multiple out of school suspensions or does not follow the behavior corrective action plan, or if the student poses a continuing danger to persons or property, or an ongoing threat of disrupting the academic process, the principal, may expel the student for twenty (20) school days or less, pursuant to the Due Process Procedures outlined below.

FOR 20 SCHOOL DAYS OR MORE: When the act of misconduct is considered a crime under state law or is so extreme that it threatens the safety of others or poses a continuing damage to persons or property or an ongoing threat of disrupting the academic process, the principal may expel the student from school for twenty (20) school days or more, not to exceed eighty (80) school days, pursuant to the Due Process Procedures outlined below.

Students who are served under the Individuals with Disabilities in Education Act (Special Education) or under Section 504 of the Rehabilitation Act of 1973 are entitled to certain additional rights in the area of discipline based upon their qualification for services under federal laws.

## Harassment, Intimidation, and Bullying

The school prohibits any and all acts of harassment, intimidation, and bullying (including acts committed through the use of a cellular telephone, computer, page, personal communication device, or other electronic communication device) of any student on school property, on a school bus, at school sponsored events, or transportation to and from school. Harassment, intimidation, and bullying is classified as either (1) violence within a dating relationship or (2) any intentional written, verbal, electronic, or physical act or threat that a student has exhibited toward another particular student or staff member more than once and the behavior, in addition to being sufficiently severe, persistent, or pervasive that it creates an intimidating,

threatening, or abusive educational environment for other students or staff member, has one or more of the following effects:

- Physically or mentally harming a student or staff member
- Damaging student or staff member property
- Placing a student or staff member in reasonable fear of harm
- Placing a student or a staff member in reasonable fear of damage to his/her property

All school employees, volunteers, and students are required to immediately report prohibited incidents of which they are aware to the principal. A report of harassment, intimidation, or bullying should be done in written form, providing as much information as possible. The report may be done anonymously. If the principal is not available, school employees, volunteers, and students should inform the grade level dean or teacher so they can ensure the principal is made aware of the issue. The principal is then responsible for determining whether an alleged incident constitutes a violation. In so doing, the principal will conduct a prompt, thorough, and complete investigation of the reported incident and prepare a report documenting the prohibited incident. Parents of any student involved in a prohibited incident will be notified and subject to federal and state law requirements, will be granted access to any written reports pertaining to the incident. Disciplinary procedures under this Student Code of Conduct, including but not limited to suspension, will be followed for any act of harassment, intimidation, or bullying. Those procedures will include protecting any victim from additional harassment, intimidation, or bullying, and from retaliation following a report. Any student who is found to have falsely accused another of bullying or harassment is subject to remedial action in accordance with the Student Code of Conduct. The school will follow any applicable federal or state reporting requirements and report incidents to the school board as appropriate.

#### Discipline of Off Campus Behavior

The school reserves the right to discipline students' off campus behavior, including behavior during transportation to and from school or school trips, which substantially disrupts the school's educational process or mission or threatens the safety or well-being of a student or staff member. Some factors which may be considered in determining whether the behavior warrants discipline include, but are not limited to, the following:

- Whether the behavior created material and substantial disruption to the educational process or school's mission due to the stress on the individual
- Whether a nexus to on-campus activities exists
- Whether the behavior creates a substantial interference with a student's or staff member's security or right to educate and receive education
- Whether the behavior invades the privacy of others
- Whether any threat is deemed to be a true threat by the administration or Board, using factors and guidelines set out by the courts or by common sense, reasonable person standards

#### Public School Bus Discipline

If the principal, Dayton Public Schools, or Trotwood Public Schools determines that a student's behavior on a public school bus violates school rules, he/she may suspend the student from school bus riding privileges for the length of time deemed appropriate for the violation and remediation of the behavior. Dayton Public Schools and Trotwood Public Schools reserve the right to deny bus transportation to any student who is unable to comply with their rules.

In partnership with Dayton Public Schools and Trotwood Public Schools, the school will honor their bus referral process in addition to applying the school's Student Code of Conduct.

Level 1: Upon receipt of the first bus referral, the principal or principal's designee will address the student's behavior with both the student and parent. A letter will be sent to the parent regarding the incident and future consequences.

Level 2: Upon receipt of the second bus referral, the principal or principal's designee will address the student's behavior with both the student and parent. The student will be suspended from transportation services for three days. During these three days, parents will have to provide alternative transportation to get their child to and from school.

Level 3: Upon receipt of the third bus referral, the principal or principal's designee will address the student's behavior with both the student and parent. The student will be suspended from transportation services for five days. During these five days, parents will have to provide alternative transportation to get their child to and from school.

Level 4: Upon receipt of the fourth bus referral, the principal or principal's designee will address the student's behavior with both the student and parent. The student will be suspended from transportation services for ten days. During these ten days, parents will have to provide alternative transportation to get their child to and from school.

**Level 5:** Upon receipt of the fifth bus referral, the principal or principal's designee will recommend to Dayton Public School's or Trotwood Public School's Transportation Department that the student be denied transportation for a period of time.

**Level 6:** Upon receipt of the sixth bus referral, the principal or principal's designee will recommend to Dayton Public School's or Trotwood Public School's Transportation Department that the student be denied transportation for the remainder of the school year.

#### Expulsions

A student shall be expelled for one (1) year for bringing a firearm, gun, or other destructive device on school property, in a school vehicle, or to any school sponsored activity and will be referred to the criminal justice system.

A firearm is defined as any weapon, including a starter gun, which will or is designed to or may readily be converted to expel a projectile by the action of an explosive, the frame or receiver of any such weapon, any firearm muffler or silencer, or any destructive device. A destructive device is a firearm for purposes of this policy, and includes but is not limited to, any explosive, incendiary, or poison gas, bomb, grenade, rocket having a propellant charge of more than four (4) ounces, missile having an explosive or incendiary charge of more than one quarter ounce, mine, or other similar device.

A student may be expelled for one (1) year for bringing a knife on school property, in a school vehicle, or to any school-sponsored activity, or for making a bomb threat to a school building or any premises at which a school activity is occurring at the time of the threat.

A knife is defined as any cutting instrument consisting of at least one sharp blade.

A student may be expelled for a one (1) year for possessing a firearm or knife at school or during a school event which firearm or knife was initially brought onto school property by another person.

A student may be expelled for one (1) year for making a bomb threat to a school building or to any premises which a school activity is occurring at the time of the threat.

A student may be expelled for up to eighty (80) days for serious misconduct or rules violations or for other just cause.

A student may be expelled for one (1) year for committing an act that is a criminal offense when committed by an adult that results in serious physical harm to persons or serious physical harm to property.

The principal may modify a one (1) year expulsion pursuant to state law requirements (See definition of Expulsion).

#### Restrictions upon Student during Suspension or Expulsion

During the period of suspension, removal, or expulsion the student may not attend or participate in any school functions without permission from the principal and only, if accompanied by a parent who accepts responsibility for the student's actions and/or behavior at the facility.

The Board also authorizes the principal to suspend a student from any or all co-curricular or extra-curricular activities for misconduct or rules violations. The length of suspension shall be determined by the principal commensurate with the seriousness of the student's misconduct or rules violations in accordance with the discipline code.

If the principal determines that a student's behavior on a school vehicle violates school rules, he/she may suspend the student from school bus riding privileges for the length of time deemed appropriate for the violation and remediation of the behavior.

The Board authorizes the principal to provide for options to suspension or expulsion of a student from school, which shall include a program whereby a student performs community service either in lieu of or as a part of a suspension or an expulsion.

The Board designates the principal or designee as its representative at all hearings regarding the decision of a suspension or expulsion, and the appeal of a suspension. The Board will hear the appeal of an expulsion.

#### Due Process Procedures

The following due process procedures only govern the suspension or the expulsion of a student from the school's regular educational program. If a student charged with violation of this Student Code of Conduct has been returned to the regular school program pending a decision, then such action of reinstatement shall not limit or prejudice the school's right to suspend or expel the student following that decision.

#### A. Suspensions

The Principal or designee may suspend a student.

1. <u>Prior to</u> the imposition of the suspension, a written Notice of Intended Suspension from School will be given to the student, or if the student poses an immediate danger or

threat to the educational environment, the principal or designee shall issue a *Notice of Emergency Removal and Intent to Suspend from School*, which contains the following:

- a. The reasons for the intended suspension; and
- b. Notification of an informal hearing.
- 2. The student must be allowed an informal hearing as soon as practicable before the Principal or designee to challenge the reasons for the intended suspension or otherwise explain his actions. In the event of an emergency removal, an informal hearing must be held as soon as practicable after the removal, but within three (3) school days after the initial removal is ordered. The student is not entitled to call witnesses at this informal hearing.
- 3. Within one (1) school day after the suspension is imposed, the Principal or designee shall provide written Notice of Suspension from School and Notice of Rights Re: Suspensions from School notification to the parent of the student and the treasurer of the Board of the suspension. The notice must contain the following:
  - a. The reasons for the suspension;
  - b. Notification of the right to appeal to Board's designee;
  - c. The right to representation at all appeals;
  - d. The right to a hearing before the Board's designee.

Suspensions imposed during the last ten days of the school year may be carried over into the following school year. However, only the Principal may impose a carryover suspension.

B. Expulsion

Only the Principal may expel a student.

- 1. <u>Prior to the imposition of the expulsion a Notice of Intended Expulsion from School shall</u> be provide to the student and the student's parent, or if the student poses an immediate danger or threat to the educational environment, the principal or designee shall issue a *Notice of Emergency Removal and Intent to Expel from School*. The notice must include the following:
  - a. The reasons for the intended expulsion; and
  - b. The time and place for a hearing, which must be not less than three (3) nor more than five (5) school days after giving the notice, unless the period is extended by the Principal at the request of the student, the parent or representative. In the event of an emergency removal, an informal hearing must be held as soon as practicable after the removal, but within three (3) school days after the initial removal is ordered. The parent must be sent written notice of any extension, and the subsequent notice should contain the same information required in the original notice. The student shall have the right to challenge the reasons for the expulsion or otherwise explain his/her actions.

- 2. Within one (1) school day after the expulsion is imposed, the Principal shall provide a written *Notice of Expulsions from School* and *Notice of Rights Re: Expulsion from School* to the parent of the student and the treasurer of the Board of the expulsion. The notice must include the following:
  - a. The length of expulsion (20 school days or less/20 school days or more)
  - b. The reasons for the expulsion;
  - c. Notification of the right to appeal to the Board;
  - d. The right to representation at all appeals;
  - e. The right to an appeal hearing before the Board;
  - f. The right to request that the hearing be held in executive session upon appeal;
  - g. When the Principal expels a student for more than twenty days or for any period of time extending into the next semester or school year, the School shall provide, along with the above Notices, a *Notice of Assistance Programs Re: Expulsion from School (for more than 20 school days)*, which contains information about services or programs offered by public and private agencies that work toward improving those aspects of the student's attitudes and behaviors that contributed to the incident giving rise to the expulsion. The information must include names, addresses, and phone numbers or the appropriate public and private agencies.

During the period of expulsion, the School may, but is not required to, continue educational services in an alternative setting.

The Principal is required to follow through on expellable offenses even if the student in question withdraws from the School prior to the hearing or the Principal's decision.

The Principal may apply any remaining part or all of the period of expulsion into the following year.

## C. Appeals before the Board of Directors

An expulsion may be appealed to the Board of Directors at the request of the student's parent. At the appeal, the student may be represented by counsel, present witnesses, cross-examine witnesses, give testimony under oath and to have a verbatim record of the appeal proceedings provided by the Board of Directors.

The student has the right to request a closed hearing. The Board, after hearing all testimony shall deliberate in closed session and issue its formal action in open session.

All decisions made by the Board of Directors in regard to expulsions are final.

## Definition of Terms

Behavior Referral Report is a specific written record of the student's act of misconduct. It documents the date, nature of the offense, and the disciplinary action taken by the teacher and/or school administration.

<u>Attendance Corrective Action Plan (A-CAP)</u> is a written document that identifies the causes for excessive tardies and/or absences and what steps will be taken to minimize the number of occurrences. It is signed by the parents, student, and staff member.

<u>Behavior Corrective Action Plan (B-CAP)</u> is a written document that identifies the cause of repeated misconduct and what steps will be taken in order to help a student overcome inappropriate behavior. It is signed by the parents, student, and staff member.

<u>Behavior Contract</u> is a written document that outlines the expected behavior and disciplinary action for a student. It is signed by the parents, student, and staff member.

<u>In-School Suspension Plan</u> refers to a decision to temporarily remove a student from the mainstream classroom and to place the student in a segregated environment within the school building that allows the student to continue to progress in their coursework but denies them access to regular classroom instruction and social interaction with their peers. The In-House Suspension room must be supervised by an adult employee of the building at all times.

<u>Suspension</u> refers to a decision to temporarily remove a student's right to attend school or any school-related activity not to exceed ten (10) school days. A meeting with the parents may be convened to discuss the incident that led to the suspension and to develop a Behavior Contract for the future.

<u>Emergency Removals</u> In the event that, in the opinion of the principal, a student's presence at the school creates a health risk, presents a danger to other persons or property, or seriously disrupts the functions of the school, the student may be removed from the premises without formal suspension or expulsion procedures, with notice and procedures to follow the removal in accordance with R.C. \$3313.66.

Expulsion is defined as the involuntary removal of a student for more than ten (10) school days. Expulsion may be imposed for up to eighty (80) school days, or the number of school days remaining in the semester, whatever is greater. However, if the offense involves a firearm or knife or is an offense that would be a crime if committed by an adult and results in serious physical harm to persons or property, the student may be expelled for up to a year.

The principal may modify a one (1) year expulsion if the student establishes in a clear and convincing manner at least one (1) of the following:

- 1. A recommendation from the group of persons knowledgeable of the student's educational needs in accordance with IDEA.
- 2. The weapon was not knowingly possessed by the student.
- 3. The student did not know or have reason to know that the object or instrument possessed by the student constituted a weapon.
- 4. The student may be eligible for participation in an alternative program.

## ATTACHMENT 6.12 and 6.13

Admissions Procedures Open Enrollment Policy, if any At-Risk Definitions, if applicable Attendance Policy (including covenant to maintain written log of daily attendance verified by the applicable teacher(s) and signed by the Principal, and kept in chronological order) Automatic Withdrawal Rule Truancy Policy and Procedure

## PATHWAY SCHOOL OF DISCOVERY Admission and Enrollment Policy

Admission to the school shall be open to all age-appropriate children for grade levels offered in accordance with the school's charter contract without charge for tuition and without discrimination on the basis of intellectual or athletic abilities, measures of achievement or aptitude, disability, status as a handicapped person, homeless status, English proficiency, religion, creed, race, sex, color, national origin or any other basis that would be illegal for an existing school district. Admission shall comply with all applicable federal and state laws. Admission shall be limited to those students who are residents of the state, except a foreign exchange student.

The school will remove barriers to the enrollment and retention in school of children and youth experiencing homelessness by developing and implementing practices and procedures consistent with the McKinney-Vento Homeless Education Assistance Act and applicable state law. The school will ensure that all identified homeless children and unaccompanied youth receive a free and appropriate education and are given meaningful opportunities to succeed in the school.

It is the policy of the Board that its educational service provider develop and implement practices and procedures that control the admission and enrollment of students, including public notice and random selection to be used when the number of applicants exceed the number of available spaces for grades offered. Detailed application, random selection and admission practices and procedures shall be available to parents and the general public at the school office. The Board will annually approve offered seats and maximum class size of the school.

References:

US Constitution, Fourteenth Amendment Title IX of Education Amendments Act (20 USC 1681 et. seq.) The Civil Rights Act of 1964 The McKinney-Vento Homeless Education Assistance Act (42 USC §11434a[2]) Rehabilitation Act of 1973 (29 USC 791 et. seq.) Equal Educational Opportunity Act of 1974 (20 USC 1703 et. seq.) The Americans with Disabilities Act of 1990 (42 USC 12101 et. seq.) Ohio Revised Code §3314 et seq. National Heritage Academies Admissions and Enrollment Practices & Procedures National Heritage Academies Homeless Child Practices & Procedures

Effective Date: January 5, 2011

#### NHA PRACTICES AND PROCEDURES: ADMISSIONS AND ENROLLMENT (OH)

The school will comply with all applicable federal and state laws related to admissions and enrollment.

#### **Non-Discrimination**

The school will not discriminate on the basis of intellectual or athletic abilities, measures of achievement or aptitude, disability, status as a handicapped person, homeless status, English proficiency, religion, creed, race, sex, color, national origin or any other basis that would be illegal for an existing public school.

#### **Open Enrollment Period and Notice**

The "**Open Enrollment Period**" for the first year of operation will be determined prior to June 30 by the NHA Admissions Department and included in the notice of Open Enrollment. In all subsequent years, the Open Enrollment Period is from the first day of school of the current school year until 5:00 p.m. on the last day of business in February of the current school year. Notice of the Open Enrollment Period and application process will be designed to inform the persons most likely to be interested in the school.

National Heritage Academies (NHA) and/or the school will provide notice of open enrollment on its website and by (a) printing a legal notice of the enrollment period in a local newspaper of general circulation; (b) mailing a written notice of the Open Enrollment Period and an application to all families who inquire about school enrollment; and (c) posting a written notice of the Open Enrollment Period at the school. In addition, notice may also be provided by airing a public service announcement on local television.

As part of the enrollment process, the school staff will communicate to meet with families, parents/guardians and students prior to the first day of school.

#### **Application Procedures**

Interested parties may obtain applications at:

- The school's website
- The offices of the school
- The service center of NHA at 3850 Broadmoor SE, Suite 201, Grand Rapids, MI 49512 or by calling 866-NHA-ENROLL from 8:00 a.m. to 5:00 p.m. EST.

Applications will be mailed, emailed or faxed to anyone requesting an application by telephone.

Applications for the current school year will be accepted until the end of the current school year and available seats will be filled. Applications for the subsequent school year are received during and after the Open Enrollment Period. If applications received during the Open Enrollment Period exceed offered seats in any grade level ("over-subscribed grades"), a random selection process will take place for all affected grade levels. If applications received are fewer than offered seats in each and every grade level ("under-subscribed grades"), all eligible applicants will be accepted and a random selection process will not be conducted.

All applications received after the Open Enrollment Period <u>will not</u> be eligible to participate in the random selection process, and will be added to the end of the accepted list if offered seats are still available after the random selection process, or to the resulting waiting list created at the time of the random selection process.

Accepted applicants must confirm their intent to attend the school within four weeks of acceptance by returning certain initial forms, including an Admissions Form and an Official Release of Records Form. The school will send letters to parents reminding them of this obligation in order to enroll their child. The school will send all applicants a postcard to inform parents/guardians that if the student does not attend the first day of school or call in to request an excused absence by the date and time indicated, the student will

#### NHA PRACTICES AND PROCEDURES: ADMISSIONS AND ENROLLMENT (OH)

forfeit his/her registered status in the school and will not be enrolled. The school may attempt to call all applicants who have not responded to inquire whether the applicant is still planning to attend.

Once students are enrolled and remain enrolled, they will remain eligible to be re-enrolled at the school for successive years without having to re-enter the random selection process. However, they will be requested to complete a re-enrollment form by the end of the Open Enrollment Period showing intent to re-enroll for the subsequent school year. All applicants on a waiting list must re-submit an application for the following school year during the next Open Enrollment Period.

#### **Random Selection Process**

The random selection process shall be open to the public, and the school will notify all applicants of the time and place. A neutral third party person will be present during the random selection process. This person will not be related to any student, staff member, anyone applying to the school, or an NHA employee. Names will be randomly selected until all offered seats have been filled. Any remaining names will be randomly selected to establish waiting list priority used to fill available offered seats prior to and during the school year for which the student applied. After all eligible names have been randomly selected, the school will add the names of applicants who submitted applications after the Open Enrollment Period in the order in which they were received. The random selection process is open to the public and will be video recorded. In the event of any discrepancy, the video recording will be the official record of placement of students.

#### **Class Size and Offered Seats**

Class size and offered seats will be recommended by NHA and submitted to the school board of directors for approval. In order to make provision for student attrition (reenrolling students who indicate that they are coming back but do not return on the first day of school) and erosion (new students who have been accepted for offered seats but are absent without excuse on the first day of school), the school may over-subscribe grades. The number of students to be over-subscribed will be determined based on historical and forecasted attrition and erosion. In addition, the number of classrooms may fluctuate in the event the number of students enrolled warrants the increase or decrease in number of classrooms. In no event will over-subscription, or fluctuations in the number of classrooms result in a violation of any provision or limit contained within the school's charter contract or applicable law.

#### **Enrollment Preferences**

Enrollment preference is first given to currently enrolled students. Next preference is given to the following ordered categories of applicants:

- · Siblings of students who are attending the school the previous year
- Applicants residing in the school district where the school is located
- All remaining applicants

If permitted by law, other enrollment preferences may be granted. If a student is selected for a grade level that still has offered seats available and the student has a sibling applying for a grade that no longer has offered seats available, the student will be accepted for his/her grade level and the student's sibling will be placed on the waiting list for his/her grade level with sibling preference. Therefore, while sibling preference applies, siblings are not guaranteed a seat.

#### Standby Opportunity Plan

The Standby Opportunity Plan (SOP) is a procedure by which the school may decide to revise its waiting list on the first day of school. If the school follows this procedure, the school will send all applicants on the waiting list a registration card prior to the first day of school. To be included in the SOP, the applicant must return the card to the school by 1:00 p.m. on the first day of school, providing phone numbers where the

#### NHA PRACTICES AND PROCEDURES: ADMISSIONS AND ENROLLMENT (OH)

applicant can be reached the first day of school between the times listed on the card. In the event of an offered seat becoming available, the school will attempt to reach the parent participating in the SOP and offer the seat. If the school cannot reach the parent at the phone numbers and during the times provided on the card, the school will contact the next person on the waiting list who is participating in the SOP. If a student participates in the SOP and a seat is not available for them, they may receive a higher waiting list priority than those students who did not participate.

#### Procedural Steps

#### Step 1: Setup

A list with the name of each student who submitted an application during the Open Enrollment Period will be created. The list will include, but not be limited to, the student's name, birth date, grade level to which the student is applying, street address, and names and grade levels of any siblings who are also applying for admission to the school.

#### Step 2: Admission of Applicants Applying for Under and Over-Subscribed Grades

A neutral third-party person (as previously described) will perform a random selection of the names of each applicant. Any under-subscribed grades will be considered before the over-subscribed grades in descending order. After all under-subscribed grades have been identified, the order of the over-subscribed grades will be randomly selected. Once the grade order has been established, randomly selected students will be placed in available seats or on the waiting list in the applying grade if an offered seat is not available. If the selected student is accepted and has siblings who are also applying for admission, the siblings will be accepted if there are offered seats available or placed on the waiting list with sibling preference if offered seats are not available. If the selected student is placed on the waiting list and has siblings who are also applying, the siblings' names will not be selected at this time or granted sibling preference, but will wait until their grade level is selected.

#### Step 3: Waiting List Priority

Students will continue to be randomly selected until all names are selected. After a grade level's seats are full, all remaining names will be placed on the waiting list in the order in which they are selected. Applications received after the Open Enrollment Period will be added to the end of the waiting list for the appropriate grade in the order in which they were received.

When a seat becomes available in a particular grade due to attrition, erosion, or other event, if that particular grade has a waiting list, that available seat will be filled by the first student on the waiting list for that particular grade. If a waiting list does not exist for that particular grade, but exists for another grade, the school may (subject to applicable enrollment limits and board approved offered seats) fill the available seat using the first student on the waiting list in a different grade, the grade deemed most beneficial to student and school considering class size, teacher capacity, and other school operational factors.

#### Appeals

Any parent or guardian who wishes to contest or appeal any aspect of the random selection process may do so in writing to the school's board sent to the school's address. Following receipt of the parent's/guardian's written appeal, a school board designee will contact the parent/guardian to discuss the nature of the concern or objection. Final decisions will be made by the school board or its designee.

#### NATIONAL HERITAGE ACADEMIES HOMELESS CHILD PRACTICES & PROCEDURES

#### 1. PURPOSE

To remove barriers to the enrollment and retention in school of children and youth experiencing homelessness in compliance with the McKinney-Vento Homeless Education Assistance Act (42 USC §11434a[2]) ("McKinney-Vento Act").

#### 2. PRACTICE

The school will adhere to the provisions of the McKinney-Vento Act and applicable state requirements to ensure that all identified Homeless Children and Unaccompanied Youth receive a free and appropriate education and meaningful opportunities to succeed in the school.

#### Definitions

"Homeless Child" - a child who does not have a fixed, regular, and adequate nighttime residence or whose primary nighttime location is in a public or private shelter designated to provide temporary living accommodations, or a place not designed for, or ordinarily used as regular sleeping accommodations for human beings. This definition includes a child who is:

- sharing the housing of other persons due to loss of housing, economic hardship or similar reason (sometimes referred to as double-up);
- living in motels, hotels, trailer parks, or camping grounds due to the lack of alternative adequate accommodations;
- living in a car, park, public space, abandoned building, substandard housing, bus or train stations or similar settings;
- abandoned in hospitals;
- awaiting foster care placement (eligibility for services for children already placed in foster care should be discussed between the Liaison and the social service provider);
- a migratory child who qualifies as homeless because he or she is living in circumstances described above; or
- an Unaccompanied Youth.

"School of Origin" - the school the child attended when permanently housed or the school in which the child was last enrolled.

"Unaccompanied Youth" - a youth not in the physical custody of a parent or guardian.

#### 3. APPLICABILITY

This document applies to all National Heritage Academies (NHA) schools.

#### 4. **RESPONSIBILITY**

- **4.1** The school principal will designate a local homeless Liaison ("Liaison").
- **4.2** The Liaison will serve as one of the primary contacts between homeless families and school staff, district personnel, shelter workers, state coordinators for Homeless Children, public and private service providers in the community, housing and placement agencies, and other service providers.
- **4.3** The Liaison will:

- Identify and enroll Homeless Children;
- Ensure that Homeless Children and their families receive eligible educational services;
- Make referrals to health, mental health, dental, and other services;
- Inform parents/guardians of educational and related opportunities available to their children;
- Provide parents/guardians with meaningful opportunities to participate in their child's education;
- Inform parents/ guardians and Homeless Children of all transportation services, and assist them in accessing these services;
- Clearly communicate all required information in a form, manner, and language that is understandable;
- Ensure proper mediation of enrollment disputes according to the McKinney-Vento Act and complaint procedures;
- Disseminate public notice of the educational rights of Homeless Children;
- Conduct annual training for school personnel on possible indicators of homelessness, sensitivity in identifying Homeless Children, and procedures for reporting to the Liaison; and.
- Record *AtSchool* information (See Exhibit 6.1);

### 5. **PROCEDURES**

#### 5.1 Identification

The burden is on the school to show that the child is not a Homeless Child. The Liaison, in collaboration with school personnel and community organizations, will identify Homeless Children, both in and out of school. Community organizations may include family and youth shelters, soup kitchens, motels, campgrounds, drop-in centers, welfare departments and other social service agencies, street outreach teams, faith-based organizations, truancy and attendance officers, local homeless coalitions, and legal services.

The Liaison must use the Student Residency Questionnaire ("SRQ") (Exhibit 6.2; for Louisiana, see Exhibit 6.3) upon enrollment and clearly describe current living arrangements of the child to determine whether the child meets the definition of a Homeless Child. Upon the receipt of an SRQ indicating potential homelessness, the Liaison will implement these practices and procedures document and ensure adherence with federal, state and NHA requirements.

NOTE: For New York schools only: Upon determination of appropriate school selection, the parent (or Liaison if no parent is available) will complete the New York STAC-202 form (Exhibit 6.4), following the instructions contained therein. Upon receipt of the STAC-202 form, the Liaison will forward the form to the New York Commissioner, Intervention Services at the NHA Service Center, and keep a copy in the school's records.

#### 5.2 School Selection

Homeless Children have the right to remain at their School of Origin or to attend any school that houses students who live in the attendance area in which they are actually living.

If the parent/guardian agrees, Homeless Children will remain at their School of Origin to the extent feasible. Homeless Children may remain at their School of Origin the entire time they are in transition and until the end of any academic year in which they become permanently housed. The same applies if they lose their housing between academic years.

*Feasibility* is a child-centered determination, based on the needs and interests of the particular child and the parent/guardian or child's wishes. Services that are required to be provided,

including transportation and services under federal and other programs, shall not be considered in determining feasibility. Feasibility considerations may include:

- Safety of the child;
- Continuity of instruction;
- Likely area of family or child's future housing;
- Time remaining in the academic year;
- Anticipated length of stay in temporary living situation;
- School placement of siblings; or
- Whether the child has special needs that would render the commute harmful.

#### 5.3 Enrollment

The school selected for enrollment must immediately enroll any Homeless Child. Unaccompanied Youth may either enroll themselves or be enrolled by a parent, non-parent caretaker, older sibling, or the Liaison. Enrollment may not be denied or delayed due to the lack of any document normally required for enrollment, any unpaid school fees, lack of uniforms or clothing that conforms to the school's dress code or any factor related to the child's living situation.

The school will coordinate the transfer of school records with other schools and contact the child's previous school to obtain school records. Initial placement of a child whose records are not immediately available will be made based on the child's age and information gathered from the child, parent, and previous schools or teachers. If no immunization records are available, the school office will refer students to the Liaison to assist with obtaining these records from state registries and/or community based clinics.

The school will excuse any tardiness or absence related to a Homeless Child's living situation when applying any school policy regarding tardiness or absences.

#### 5.4 Services

The school must provide Homeless Children services comparable to services offered to other students in the school, including:

- Transportation;
- Title I;
- Educational services for which the student meets eligibility criteria, including special education and related services and programs for English language learners;
- School nutrition programs (the school will provide free meals to the Homeless Child as all Homeless Children are automatically eligible for free meals);
- Vocational and technical education programs;
- Gifted and talented programs; and
- Before- and after-school programs

#### Transportation

At a parent/guardian's request, the school will provide transportation to and from the School of Origin. The Liaison will coordinate these arrangements, which may include arrangements with the social service district. The school must provide the transportation for the entire time the child has a right to attend that school, as defined above, including during pending disputes. The length of the commute will only be considered in determining the *feasibility* of placement in the School of Origin based on potential harm to the child. Prior to selection of a school, the Liaison will inform the parent/guardian or Unaccompanied Youth of this right to transportation. Transportation disputes will not result in a Homeless Child missing school. If such a dispute arises, the school will arrange transportation and immediately bring the matter to the attention
of the state authorities following the appropriate complaint procedures as detailed in this document.

#### <u>Title I</u>

Homeless Children are automatically eligible for Title I services. The school will reserve the necessary funds to provide services comparable to those provided to Title I students attending non-participating schools, including education related support services and removing barriers that prevent attendance. The Liaison and the Title I director at the NHA Service Center will develop the formula (based upon the per-pupil Title I expenditures) to use for determining the necessary funds to reserve.

The Title I director and the Liaison will ensure coordination between the Title I plan and the McKinney-Vento Act, including the academic assessment, reporting and accountability systems required by federal law and the U.S. Department of Education.

#### Educational Services

The school shall give evaluations of Homeless Children suspected of having a disability priority and coordinate the evaluation with the student's prior and subsequent schools, as necessary, to ensure timely completion of a full evaluation. When necessary, the school will expeditiously designate a surrogate parent for Homeless Children suspected of having a disability. The school will immediately implement the child's Individualized Education Program (IEP), if available, and promptly conduct any necessary IEP meetings or re-evaluations. If complete records are not available, IEP teams will use good judgment in choosing the best course of action, balancing procedural requirements and the provision of services in an attempt to avoid any disruption in services.

#### 5.5 Complaint and Dispute Resolution

The following steps and procedures should take place in resolving disputes regarding enrollment, school placement, or services.

#### <u>Complaint</u>

- A complaint is an oral or written and signed statement alleging the violation of a federal or state law, rule, or, regulation. The complaint must allege a violation that occurred not more than one (1) year prior to the date that the complaint is received, unless a longer period is reasonable because the violation is considered systemic or ongoing.
- Parents, teachers, administrators, or other concerned individuals or organizations may file a complaint ("Complainant"). Faculty and staff with knowledge of a complaint must refer the Complainant to the Liaison.
- The Complainant may submit the complaint in writing to the Liaison, using the attached Complaint Resolution Initiation Form (Exhibit 6.5). The Complainant may choose to initiate the complaint orally.

#### Role of School

- Immediately enroll the child in the school preferred by the person(s) bringing the complaint.
- Provide all educational services for which the child is eligible, pending resolution of the dispute.

#### Role of Liaison

- After receipt of the complaint, the Liaison must provide a written explanation of the school placement decision and/or provided services to the Complainant and discuss the complaint with the Complainant. The Liaison must provide a written proposed resolution or a plan of action to the Complainant within five (5) days of receipt of the complaint, or within seven (7) business days of notification of dispute (Georgia only).
- If the Liaison does not resolve the dispute, the Complainant may forward it to the school principal. The Liaison must provide a written resolution to the parties within five (5) days of the discussion with the principal, or within ten (10) business days of the second dispute (Georgia only). The Complainant has a right to obtain assistance from advocates or attorneys in addressing a complaint.
- The Liaison will carry out the dispute resolution in an expeditious manner and will provide the Complainant these written procedures, including the appeal procedures outlined below.
- In the event the school is unable to resolve the complaint, the Complainant may pursue the applicable appeal procedure(s).

#### Appeal Procedures – Colorado

If the dispute is not resolved at the school level, the Liaison will assist the Complainant in contacting a mediator from the Colorado Mediator Resource Network (services at no charge to the Complainant) or may direct the Complainant to the Colorado Department of Education (CDE).

Address the complaint to the following address:

• State Coordinator for the Education of Homeless Children and Youth, Colorado Department of Education, State Office Building, 201 East Colfax Avenue, Denver, Colorado 80203-1799.

Use the Colorado Dispute Report Form (Exhibit 6.6) during the appeal process.

The complaint should include:

- the name, address, and telephone number of the person filing the appeal;
- the relationship or connection of the person to the child in question;
- the name and age of the child involved;
- the name of the school and school personnel involved in the complaint;
- the federal requirement alleged to have been violated;
- a description of the situation that prompted the complaint;
- a description of the attempts that were made to solve the issue; and
- the relief the person is seeking.

If the State Coordinator is unable to resolve the complaint within 15 business days, the Complainant may file a written complaint to the State Coordinator who, with a team, will review the complaint with the mandates of the Title X law. Within 15 days of receipt of the complaint, the Coordinator will issue a written decision to the parties via mail.

#### <u> Appeal Procedures – Georgia</u>

If the dispute is not resolved at the school level, the Complainant may direct the complaint to the Georgia Department of Education state homeless coordinator. The Liaison may assist the Complainant in contacting the Department.

Address the complaint to the following address:

• Program Consultant, Homeless Education Program, Georgia Department of Education, 1866 Twin Towers East, 205 Jesse Hill, Jr. Drive, SE, Atlanta, Georgia 30334.

#### Appeal Procedures – Indiana

If the dispute is not resolved at the school level, the Complainant may direct the complaint to the Indiana Department of Education (IDOE). Complaints made under this process must be in writing and signed by the Complainant.

If the complaint involves enrollment or school placement of homeless children, address the complaint to the following address:

• State Board Liaison; Indiana Department of Education; Room 299, State House; Indianapolis, Indiana 46204.

If the complaint involves comparable services offered to homeless children, address the complaint to the following address:

• Director of the Division of Educational Options; Indiana Department of Education; Room 299, State House; Indianapolis, Indiana 46204.

The complaint should include:

- the name, address, and telephone number of the person filing the appeal;
- the relationship or connection of the person to the child in question;
- the name and age of the child involved;
- the name of the school and school personnel involved in the complaint;
- the federal requirement alleged to have been violated;
- a description of the situation that prompted the complaint;
- a description of the attempts that were made to solve the issue; and
- the relief the person is seeking.

The IDOE will issue a letter of acknowledgement to the Complainant and the Liaison containing the IDOE's commitment to issue a resolution in the form of a *Letter of Findings*.

An IDOE complaint investigator will review all relevant information and issue the *Letter of Findings* to the Complainant.

If the complaint involves enrollment or school placement, a hearing shall be held with each interested party given at least 10 days notice of the hearing. Any interested party may appear at the hearing in person or by counsel, present evidence, cross-examine witnesses, and present in writing or rally summary statements of position. The Indiana State Board of Education or a hearing examiner may conduct the hearing at any place in Indiana.

The determination of the IDOE or Indiana State Board of Education as a result of the proceedings described above is final and binding on the parties to the proceedings.

#### <u> Appeal Procedures – Louisiana</u>

If the dispute is not resolved at the school level, the Liaison will assist the Complainant in contacting the local (EBR) Liaison for assistance with the appeal process. In the event the dispute is not resolved at the (EBR) Liaison level, the Complainant may contact the State Coordinator to hear an appeal of the Local (EBR) Liaison's decision.

Address the complaint to the following address:

• State Coordinator – Homeless Education, Louisiana Department of Education, Office of School & Community Support, P.O. Box 94064, Baton Rouge, Louisiana 70804.

The DOE will acknowledge receipt of the complaint in writing to the Complainant and provide written resolution of the complaint within 60 days of the date the DOE receives the complaint. The decision will include the Complainants right to request the Secretary of the U.S. Department of Education to review the final decision of the DOE, at the Secretary's discretion.

#### Appeal Procedures – Michigan

If the dispute is not resolved at the school level, the Complainant may direct the complaint to the Michigan Department of Education. Complaints made under this process must be in writing and signed by the Complainant.

Address the complaint to the following address:

• State Homeless Coordinator, Office of Field Services, Michigan Department of Education, P.O. Box 30008, Lansing, MI 48909.

The complaint should include:

- the name, address, and telephone number of the person filing the appeal;
- the relationship or connection of the person to the child in question;
- the name and age of the child involved;
- the name of the school and school personnel involved in the complaint;
- the federal requirement alleged to have been violated;
- a description of the situation that prompted the complaint;
- a description of the attempts that were made to solve the issue; and
- the relief the person is seeking.

The State Homeless Coordinator will gather needed information from statements of the parties involved and will forward the information to the Director of the Office of Field Services along with a recommendation of resolution or for further investigation.

Within 30 days after receiving a complaint, the Director of the Office of Field Services will recommend a resolution and will inform interested parties in writing of the decision.

If a Complainant or one of the parties involved in the complaint disagrees with the decision, that party may, within 10 working days, appeal to the Deputy Superintendent. This appeal must be in writing and state why the party disagrees with the decision of the Director of the Office of Field Services.

Within 30 days after receiving an appeal, the Deputy Superintendent will render a final administrative decision and notify the Complainant and the school in writing.

If the party disagrees with the decision of the Deputy Superintendent, the party may request a review of the decision by the United States Secretary of Education in accordance with 34 CFR Part 299.11.

#### Appeal Procedures – North Carolina

If the dispute is not resolved at the school level, the Complainant may direct the complaint, orally or written, to the North Carolina Department of Public Instruction.

Address the complaint to the following address:

• State Homeless Coordinator, Homeless Education & Rural Education Achievement Program Consultant, Compensatory Education, NC Department of Public Instruction, 6351 Mail Service Center, Raleigh, NC 27699-6351.

The complaint should include:

- the name, address, and telephone number of the person filing the appeal;
- the relationship or connection of the person to the child in question;
- the name and age of the child involved;
- the name of the school and school personnel involved in the complaint;
- the federal requirement alleged to have been violated;
- a description of the situation that prompted the complaint;
- a description of the attempts that were made to solve the issue; and
- the relief the person is seeking.

The Liaison will provide the State Homeless Coordinator with any information that the State Homeless Coordinator requests regarding the issues presented in the appeal.

The State Homeless Coordinator will provide the school and the Complainant the opportunity to respond to any decision made and to provide any additional evidence the Complainant deems relevant.

Within 10 schools days following receipt of the appeal, the State Coordinator shall issue a final written decision to the school and the Complainant .

#### Appeal Procedures - New York

If the dispute is not resolved at the school level, the Liaison will assist the Complainant in contacting the Commissioner, completing the New York Form Petition (Exhibit 6.7), and providing copies of the form and supporting documentation at no cost to the Complainant.

The Liaison will provide the Complainant an acknowledgement of receipt of the Form Petition and will transmit the Form Petition and related documents on behalf of the Complainant to the Office of Counsel, New York State Education Department, State Education Building, Albany, New York 12234 within five days of receipt.

If a stay request is made on the Form Petition and the Commissioner grants a stay order, the Homeless Child may continue attending the school until the Commissioner issues an appeal decision. If the Commissioner denies the stay request, the Homeless Child can be asked by the principal to leave the school immediately.

If the Commissioner sustains the appeal of the Complainant, the Homeless Child can continue attending the school. However, if the Commissioner dismisses the appeal, the Homeless Child can be asked by the principal to leave the school immediately.

#### <u>Appeal Procedures – Ohio</u>

If the dispute is not resolved at the school level, the Complainant may direct the complaint to the Ohio Department of Education. Complaints made under this process must be in writing and signed by the Complainant.

Address the complaint to the following address:

 Homeless Education Coordinator, Ohio Department of Education, 25 S. Front Street, Mail Stop 404, Columbus, Ohio 43215. The complaint should include:

- the name, address, and telephone number of the person filing the appeal;
- the relationship or connection of the person to the child in question;
- the name and age of the child involved;
- the name of the school and school personnel involved in the complaint;
- the federal requirement alleged to have been violated;
- a description of the situation that prompted the complaint;
- a description of the attempts that were made to solve the issue; and
- the relief the person is seeking.

The Homeless Education Coordinator will recommend a decision to the Complainant and the Liaison. If unresolved, the Complainant may file a final appeal to the State Superintendent of Public Instruction for review and disposition.

#### Appeal Procedures - Wisconsin

If the dispute is not resolved at the school level, the Complainant may direct the complaint to the State Superintendent of Publid Instruction, in accordance with the Wisconsin Department of Public Instruction Complaint Resolution and Appeals Process, Chapter PI 1.

Address the complaint to the following address:

• State Superintendent of Public Instruction, P.O. Box 7841, Madison, WI, 53707-7841.

The complaint should include:

- the name, address, and telephone number of the person filing the appeal;
- the relationship or connection of the person to the child in question;
- the name and age of the child involved;
- the name of the school and school personnel involved in the complaint;
- the federal requirement alleged to have been violated;
- a description of the situation that prompted the complaint;
- a description of the attempts that were made to solve the issue; and
- the relief the person is seeking.

#### 6. EXHIBITS

- 6.1 AtSchool Homeless Procedures
- 6.2 Student Residency Questionnaire
- 6.3 Louisiana Residency Questionnaire
- 6.4 New York STAC-202 Form
- 6.5 Complaint Resolution Initiation Form
- 6.6 Colorado Dispute Report Form
- 6.7 New York Form Petition

Last revised: June 30, 2011

# Parental Partnership: Attendance

Regular attendance establishes good work habits and self-discipline. Our school records attendance every morning, afternoon, and in each middle school class period.

#### Arrival

Students should arrive at school between 7:35 a.m. and 8:00 a.m. After exiting their vehicle, students should proceed directly into the building. They may not run, shout, or otherwise act in a disorderly manner. All students wearing hats should remove them upon entering the building. Prompt arrival is essential for an orderly opening ceremony.

#### Dismissal

School dismisses at 3:15 p.m. In the interest of both the students' safety and the teachers' preparation time, parents should pick up students immediately after dismissal and no later than 3:30 p.m. Parents are not permitted to pick up students from their classrooms. Students who ride the bus will be escorted to their buses by a teacher.

#### Tardiness and Early Dismissal

Students are considered tardy if they arrive to their homerooms after the start of either class or morning ceremony, if held prior to class. Early dismissals from school will also be considered tardies. Parents who dismiss their child early from school, for any reason, must sign the child out.

The tardy is considered unexcused unless a note is provided by the parent the day of the tardy and the tardy is excused under the reasons below. The note must contain the following information:

- Student's name
- Teacher's name
- Date of tardy
- Reason for tardy
- Phone number
- Signature of parent

If a tardy is due to a doctor or dental appointment, a note from a physician is also required. Tardies are excused for only the following reasons:

- Personal illness or injury
- Funerals
- Doctor or dental appointments (which could not be arranged outside school hours)
- Religious observances
- Late arrivals due to weather conditions (which do not require parent note)
- Authorized tardy (approved by the principal)

Abuse of excused tardies will be addressed by the principal and may result in unexcused tardies with disciplinary action taken under the Student Code of Conduct in the sole discretion of the principal.

#### Absences

All absences are to be verified by a call from the parent to the school office no later than the start of the school day. Voicemail is available before and after school hours.

If an absence is not verified by a call from the parent, the absence is considered unexcused. An absence is excused if a note is provided by the parent within two days following the absence and meets the guidelines below for an excused absence. The note must contain the following information:

- Student's name
- Teacher's name
- Date of absence
- Reason for absence
- Phone number
- Signature of parent

If an absence is due to illness and lasts five days or more, a note from a physician is also required. Absences are excused only for the following reasons:

- Personal illness or injury
- Funerals
- Doctor or dental appointments (which could not be arranged outside school hours)
- Religious observances
- Out-of-state travel (up to a maximum of four days) to participate in a school approved enrichment or extracurricular activity
- Authorized absence (approved by the principal)

Abuse of excused absences will be investigated by the principal and may result in a referral to the Prosecutor's Office and/or Social Services or other disciplinary action. A letter of potential withdrawal and referral to Juvenile Court with possible referral to Social Services may be sent after 5 consecutive days, 7 days in a month, or 12 days in a year of unexcused absences.

Unexcused tardies and absences are handled in the following manner:

| Range of<br>Unexcused<br>Tardies | Range of<br>Unexcused<br>Absences | Progressive Consequences   |
|----------------------------------|-----------------------------------|--|
| 1 - 3                            | 1 - 2                             | Early Intervention   |
| 4 - 6                            | 3 - 4                             | Parent Notification (Email or note)  |
| 7 - 9                            | 5 - 6                             | Parent Contact (Phone or meeting)  |
| 10 - 12                          | 7 - 9                             | Level 4 infraction*<br>Completion of Attendance Corrective Action Plan (A-CAP) |
| 13 - 16                          | 10 - 14                           | 1 <sup>st</sup> Level 5 infraction*<br>Appropriate Consequence                 |
| 17 - 20                          | 15 - 18                           | 2 <sup>nd</sup> level 5 infraction*<br>Appropriate Consequence                 |
| 21 - 24                          | 19 - 22                           | Level 6 infraction*<br>Possible Expulsion                                      |

\*Potential referral to the Prosecutor's Office, Juvenile Court, and/or Social Services with certified return receipt to parent. Factor considered in grade promotion.

Note: Any student who accumulates consecutive days equal to 105 hours of unexcused absences will be automatically withdrawn.

#### Truancy

A student is habitually truant if unexcused absences occur five (5) or more consecutive school days, seven (7) or more school days in one (1) month, or twelve (12) or more school days in one (1) school year. If the student is habitually truant, the Board authorizes the principal to inform the student and his/her parents of the truancy record and the principal's intent to notify the Juvenile Court and possible referral to Social Services.

A student is chronically truant if unexcused absences occur seven (7) or more consecutive school days, ten (10) or more school days in one (1) month, or fifteen (15) or more school days in one (1) school year.

The principal may act as the school's attendance officer or delegate that duty. The school's attendance officer shall investigate possible school attendance violations, and he/she is authorized under Ohio law to serve warrants, to enter places where children of compulsory school age are employed, and to take such other actions as may be necessary to enforce the compulsory education laws.

The principal is authorized to refer the truancy to mediation or to establish a parent education program for parents of students who are habitually truant. Any parent assigned to the program who does not complete the program is to be reported to law enforcement authorities for neglect of parent education.

#### Family Vacations

Parents planning to take their child on a trip must notify the principal or secretary at least two weeks before departure. The student's absence will be an excused absence if approved by the principal. Lengthy trips are discouraged and may be cause for withdrawal. It becomes the student's responsibility, with the help of the parents, to make arrangements with each teacher for missed assignments and to complete assigned work upon his/her return to school. The student is to complete all assignments within the same number of days as the length of the vacation. Incomplete assignments will not receive credit. Students who miss an announced test during their absence will take the test when all missed work is completed and within the same number of days missed due to the vacation.

#### Family Death or Terminal Illness

If there is a death, terminal illness, or similar traumatic situation in a student's family that may affect his/her attendance, emotional well-being, and/or level of concentration, the office staff members should be notified.

#### Illness during the School Day

If a student becomes ill during the school day, appropriate arrangements will be made for the student's care while waiting for parent pick-up. Students with vomiting or fever must be symptom-free for 24 hours before returning to school.

#### Voluntary Withdrawal

To withdraw a student, a parent must complete a Student Withdrawal form available in the office. The principal is required to sign.

## ATTACHMENT 7.2(h) <u>REPORTING REQUIREMENTS</u> (not exhaustive)

- Current roster of staff by position
- Criminal background checks (FBI and BCI&I) -employees and Board
- Certificates (copy) for all teachers and aides new/renewal Information on "highly qualified" status of each
- Current roster of students identify regular and special education, §504 plans or English language learners
- Current roster of Board members including names, addresses (home or work), electronic mail addresses, and home and work telephone numbers
- Current roster of all staff with contact information and salaries or compensation
- Schedule of regular Board meetings dates, times and places
- Notice of all regular, special or emergency Board meetings
- Copies of all approved and signed Board minutes (after every meeting)
- School calendar
- Lease/rental agreement/deed
- Building inspection certificates fire and health department inspections/building permits/Certificate of Occupancy
- Current School environment report (Health Dept.) Any asbestos management plans.
- Board approved budget by June 30<sup>th</sup>
- Inventory per Board Fixed Assets Policy
- Articles of Incorporation, Code of Regulations, Employer ID No., IRS Determination Letter (if any), supplemental bylaws or policies
- Safety plan new/revised
- Approved and updated technology plan
- Annual report
- Complete set of staff, student, safety, Board and other school policies
- Financials (at least bi-monthly)
- Structural chart
- Timely submission of all required or requested information into AOIS document management system

## **ATTACHMENT 8.2**

Staff Benefits Dismissal and Disciplinary Procedures for Staff Disposition of Employees if Contract is Terminated



Employee Handbook 2011-2012

**Revised July 2011** 

# **Table of Contents**

| WELCOME TO NATIONAL HERITAGE ACADEMIES                               | 2    |
|--|------|
| NATIONAL HERITAGE ACADEMIES OVERVIEW                                 | 3    |
| SPECIAL NOTICE TO STAFF IN NEW YORK SCHOOLS                          | 4    |
| SPECIAL NOTICE TO TEACHERS IN NORTH CAROLINA SCHOOLS                 |      |
| NHA CONTINUED SUCCESS FACTORS  |      |
| REWARDS AND RECOGNITION  |      |
| Pay Systems  | 7    |
| SALARY PRO-RATION  | 7    |
| PERFORMANCE APPRAISALS AND PAY INCREASES                             | 7    |
| Bonuses  | 7    |
| Direct Deposit   | 8    |
| DEDUCTIONS FROM WAGES OF EXEMPT EMPLOYEES                            |      |
| OVERTIME   |      |
| <br>Рауснескя  | 8    |
| TIMESHEETS   |      |
| VACATION   |      |
| HOLIDAYS   | 10   |
| TRAVEL EXPENSE REIMBURSEMENT   |      |
| Benefit Elections  |      |
| GROUP HEALTH AND DENTAL INSURANCE                                    |      |
| GROUP VISION INSURANCE   |      |
| FLEXIBLE SPENDING ACCOUNT  |      |
| BENEFITS CONTINUATION (COBRA)  |      |
| LIFE INSURANCE/ACCIDENTAL DEALTH AND DISMEMBERMENT                   |      |
| Short-Term Disability  |      |
| Long-Term Disability   |      |
| 401(K) SAVINGS PLAN  |      |
| SERS/STRS  |      |
| <u>561895110</u>   |      |
| TUITION REIMBURSEMENT  |      |
| EMPLOYEE ASSISTANCE PLAN   |      |
| Nursing Mothers  |      |
| Service Recognition  |      |
| SERVICE RECOGNITION<br>EXCELLENCE IN TEACHING AWARD                  | . 16 |
| CHARTER SCHOOL / TEACHER APPRECIATION WEEK & AMERICAN EDUCATION WEEK |      |
| DISCOUNT FOR EDUCATORS   | . 16 |
| ADOPTION ASSISTANCE  |      |
| ADOPTION ASSISTANCE  |      |

| LEAVES OF ABSENCE                               | . 17 |
|---|------|
| Bereavement Leave                               | . 17 |
| FAMILY AND MEDICAL LEAVE (FMLA)                 | . 17 |
| UNPAID PERSONAL LEAVE (NOT COVERED BY FMLA)     | . 20 |
| MILITARY LEAVE                                  |      |
| JURY AND WITNESS DUTY                           | . 21 |
| POLICIES AND PROCEDURES                         | . 22 |
| EQUAL EMPLOYMENT OPPORTUNITY                    | . 22 |
| CAREER OPPORTUNITY PROGRAM                      |      |
| Personnel Records                               |      |
| CRIMINAL CONVICTIONS                            | . 23 |
| HIRING OF RELATIVES                             | . 23 |
| Personal Appearance                             | . 23 |
| ATTENDANCE AND PUNCTUALITY                      | . 23 |
| RELIGIOUS ACCOMMODATION POLICY                  | . 24 |
| PERSONAL DATA CHANGES                           |      |
| PRE-EMPLOYMENT PERSONNEL PAPERWORK              | . 24 |
| REFERENCE AND BACKGROUND CHECKS                 | . 24 |
| LICENSES AND CERTIFICATIONS                     | . 25 |
| TOBACCO-FREE ENVIRONMENT                        | . 25 |
| WEATHER EMERGENCY                               | . 25 |
| INVESTIGATIONS                                  |      |
| Personal Property                               | . 25 |
| Sexual and Other Unlawful Harassment            | . 25 |
| Drug-Free Environment                           | . 26 |
| Solicitation                                    | . 27 |
| Copyright Compliance                            |      |
| Conflicts of Interest                           |      |
| Policy Regarding Use of Social Security Numbers |      |
| CONFIDENTIALITY POLICY                          |      |
| HIPPA - NOTICE OF PRIVACY PRACTICES             |      |
| GENETIC INFORMATION NON-DISCRIMINATION ACT      |      |
| Social Media Policy                             |      |
| ELECTRONIC MONITORING POLICY                    | 31   |
| RULES OF CONDUCT.                               | 32   |
| Working Together                                | 32   |
| CORRECTIVE ACTION                               |      |
| COMMUNICATIONS                                  | 34   |
| INFORMATION TECHNOLOGY                          | 34   |
| Outside Requests for Information                | 34   |
| PROBLEM SOLVING PROCEDURE                       |      |

| EALTH AND SAFETY     | 36   |
|----------------------|------|
| Employee Injuries    | 36   |
| BLOODBORNE PATHOGENS | 36   |
|                      |      |
| WORKPLACE VIOLENCE   | 30   |
| EFINITIONS           | . 37 |

## **Rewards and Recognition**

Your paycheck is one of the most tangible rewards for your work at NHA. Yet, it is only part of the many benefits available to you. You also may enjoy a broad range of benefit programs, including group health and dental insurance, paid vacation, disability benefits and a retirement savings plan.

This portion of the Employee Handbook contains a general description of our pay policies and the benefits to which you may be entitled. Please understand that this general explanation is not intended to provide you with all the details of these programs. Therefore, this Handbook does not change or otherwise interpret the terms of any summary plan descriptions. Your rights can be determined only by referring to the full text of the summary plan descriptions. To the extent that any of the information contained in this Handbook is inconsistent with the plan documents, the provisions of the plan documents will govern in all cases. NHA may amend, modify or delete, in whole or in part, the provisions of the described benefit plans. We also may require or increase employee premium contributions towards any benefit at our discretion. If you need more information on a specific program, or if you have any questions regarding specific state laws that may apply to you, please contact the People Services Department.

#### **Pay Systems**

NHA is committed to rewarding employees for their performance. We assess results and how you get results. This performance-related information influences your total compensation package.

When determining an employee's compensation, NHA takes into account the **person** who holds the position, the **position** itself, and the person's **performance**. We assess your performance through quantitative measures and qualitative observations, and this information impacts your total compensation package.

We call this "pay for performance," and it is a fundamental part of our culture. Our ability to pay for wages and benefits is directly linked to our success. We believe that every employee can contribute to our long-term success. That is why your wage is directly linked to your performance on the job.

#### **Salary Pro-ration**

Teachers or other salaried school staff (excluding Principals, Assistant Principals, Deans, and Lead Secretaries/Office Managers/Registrars) who are hired after the start of the school year (i.e., after the first day students report to school) will receive a pro-rated salary. This means that a teacher hired after the start of the school year will receive a biweekly salary that is less than what it would have been had the teacher started at the beginning of the school year. The pro-rated biweekly salary is calculated based on the percentage of the school year the employee works, in accordance with the school vacation policy. For more information on salary proration, please contact the People Services Department.

#### **Performance Appraisals and Pay Increases**

Performance appraisals serve as an important communication link regarding your work performance. We use the performance appraisal process to evaluate individual accomplishments and also to identify areas of performance that do not meet expectations. Your job classification determines the timing of your performance appraisal and which format is used. Normally, pay increases, if warranted, will be granted on an annual basis.

#### Bonuses

NHA may pay bonuses to job classifications of employees on a periodic basis in recognition of their achievements and as an incentive for future performance. Decisions as to whether to award a bonus, as to the amount of any bonus, and as to who receives a bonus rests entirely with NHA. Bonuses, when given, are paid only to eligible employees who are actively employed on the last day of the bonus plan year (July 1 - June 30). To be eligible for the bonus, the employee must be in good-standing, report to work and work the last day of the bonus plan year unless on an approved leave.

#### **Direct Deposit**

NHA offers direct deposit of your paycheck (automatic payroll deposit). You may deposit your paycheck in up to two (2) checking accounts and three (3) savings accounts. If you want direct deposit of your paycheck, please see your school secretary, the People Services Department, or consult my.heritageacademies.com for a Direct Deposit Authorization form. Sign-up for the program or changes can be done at any time. We encourage you to use this convenient benefit.

#### **Deductions from Wages of Exempt Employees**

Exempt employees will generally be paid a guaranteed salary each week if they perform any work in the week, regardless of the number of hours worked. NHA will only make deductions from an exempt employee's pay as permitted under the Fair Labor Standards Act and its regulations.

If an exempt employee believes that NHA has made an impermissible deduction from his or her pay, the employee should immediately report the matter to the Payroll Department. NHA will investigate the matter and, if the deduction was improper, NHA will promptly reimburse the employee and will take steps to ensure that improper deductions are not made in the future. NHA will not retaliate against an employee who makes a complaint under this policy.

#### Overtime

Overtime applies to all non-exempt employees. Overtime pay is based on actual hours worked. Time off on vacation leave, holiday leave, or any leave of absence will not be considered hours worked for purposes of performing overtime calculations.

Overtime hours that you work must be pre-approved by your Principal/manager and documented on your timesheet in Lawson Employee Self-Service. You will be paid 1-½ times your regular hourly base rate for hours worked over 40, or as otherwise required by applicable state law.

#### Paychecks

Paychecks include your gross wages earned minus all legally required deductions (taxes, garnishments) and individually authorized deductions (insurance, flex, direct deposits, etc.).You will receive your bi-weekly paycheck on Thursday for the time worked during the preceding two full weeks. During summer, school employee checks will be mailed to homes. The pay period begins on Sunday and ends on Saturday. Please review your paycheck for errors. If you find a mistake, report it to your Principal/manager immediately. Any error will be corrected as soon as possible.

#### **Timesheets**

All hourly employees need to complete a time sheet in Lawson (People Services / Payroll Software System) to reflect how many hours were worked. The timesheet must be accurate and must include the number of hours worked at all hourly positions. Employees who work in more than one hourly position must separate the hours worked and report them separately.

Please ensure that your actual hours worked and any leave time is recorded accurately. Falsification of a timesheet may result in disciplinary action, up to and including termination from employment.

Time entry and approval must be completed by 5:00 the Friday prior to payday. When a pay period ends on a holiday weekend, changes to time entry requirements will be communicated to hourly employees by the Payroll Department.

Time not entered and approved by the appropriate time will result in a payment delay. The time will be included on the paycheck following this pay period.

Information on entering and submitting timesheets is available through the Payroll Department or Lawson Reference Site.

#### Vacation

Vacation time is provided to encourage employees to balance their work and personal lives. If you are eligible for vacation pay, your vacation pay will consist of your regular rate of pay and is paid with the normal payroll. Due to the differences of laws in the various states, refer to your offer letter and the applicable vacation policy for your specific vacation eligibility. An Annual School Employment Calendar is published each year to detail the employment, holiday, and vacation day expectations for all school employees. Please see your school office, My NHA or the People Services Department for a copy of the vacation policy applicable to your position or for your school's Annual School Calendar.

Vacation time is for those who are actively employed at the time of the designated vacation (i.e. time at the end of the calendar year, mid-winter break, spring break, and the summer weeks between school years). Vacation pay is not accrued based on the length of time you are employed during a school year. If you are not working on the last day of school or scheduled work day (or on an approved leave of absence) prior to the designated vacation period, you will not be eligible for vacation pay.

#### School Leadership (Principals, Assistant Principals, Principals in Residence and Deans):

Principals and Assistant Principals receive vacation time during the school year for break periods specifically designated in the school's Annual School Calendar. In addition to these vacation times, you also are eligible to take two weeks of vacation time during the summer weeks between school years. These two weeks must be approved by your supervisor.

Principals in Residence and Deans receive vacation time during the school year for break periods specifically designated in the school's Annual School Calendar. In addition to these vacation times, you also are eligible to take four weeks of vacation time. The vacation time is intended primarily for the summer break; however, with Principal approval, the time can be taken any time during the school year.

#### Registrars (new office model):

Registrars receive vacation time during the school year for break periods specifically designated in the school's Annual School Calendar. In addition to these vacation times, you also are eligible to take two weeks of vacation time during the year. These two weeks must be approved by your supervisor, and should not be taken during the two weeks preceding the start of the upcoming school year, unless approved by your supervisor. These two weeks of vacation will be pro-rated for part-time Registrars and for Registrars hired mid-year.

#### Office Administrators (new office model):

Office Administrators are non-exempt and are paid on an hourly basis. Vacation pay for hourly employees is based on a vacation pay accrual system. Your pay consists of an hourly pay rate (85%) and a vacation pay accrual (15%). You will receive your vacation pay accrual in each paycheck, but this portion of your total pay is payment for the break periods specifically listed in the Annual School Calendar. You may be ineligible for unemployment compensation during this time. These weeks specifically include designated time at the end of the calendar year, a mid-winter break, a spring break and other times when you are not working. In addition to the vacation times mentioned above, also included in the 15% vacation pay accrual is four weeks of vacation time to be taken during the summer months between school years. All vacation time taken during the summer months must be approved by your supervisor.

#### Lead Secretaries, Assistant Secretaries and Office Assistants (old office model):

Lead Secretaries receive vacation time during the school year for break periods specifically designated in the school's Annual School Calendar. In addition to these break periods, you also are eligible to take four weeks of vacation time during the summer months between school years. These four weeks must be approved by your supervisor and will be pro-rated for Lead Secretaries who have not worked the entire school year preceding the summer break or for a part-time Lead Secretary.

Assistant Secretaries and Office Assistants are non-exempt and are paid on an hourly basis. Vacation pay for hourly employees is based on a vacation pay accrual system. Your pay consists of an hourly pay rate (85%) and a vacation pay accrual (15%). You will receive your vacation pay accrual in each paycheck, but this portion of your total pay is payment for the break periods listed in the Annual School Calendar. You may be ineligible for unemployment compensation during this time. These weeks specifically include designated time at the end of the calendar year, a mid-winter break, a spring break, and the summer weeks between the school years when you are not working.

Schools operating under this office staffing model must have office staff coverage for at least four hours each day during the month of July or another four week period when the Lead Secretary takes vacation. The Assistant Secretary or Office Assistant may be required to work during this time and will be paid his/her normal hourly rate for any hours worked. All time off and scheduling must be approved by the Principal.

#### Teachers, Paraprofessionals and Other School Employees:

Teachers and other *exempt, salaried* school employees receive vacation time during the school year for break periods specifically designated in the school's Annual School Calendar. These weeks specifically include designated time at the end of the calendar year, a mid-winter break, a spring break, and the summer weeks between the school years. Employees may be required to attend occasional meetings and/or professional development during the summer weeks.

Vacation pay for Paraprofessionals and other non-exempt, hourly school employees is based on a vacation pay accrual system. Your pay consists of an hourly pay rate (75%) and a vacation pay accrual (25%). You will receive your vacation pay accrual in each paycheck, but this portion of your total pay is payment for the vacation weeks listed in the Annual School Calendar. You may be ineligible for unemployment compensation during this time. These weeks specifically include designated time at the end of the calendar year, a mid-winter break, a spring break, and the summer weeks between the school years when you are not working.

#### Admissions Representatives:

Admissions Representatives are Service Center employees and should refer to the *Vacation Benefits-Service Center Employees Policy* for information about their vacation benefits. Please contact your supervisor, the People Services Department or consult my.heritageacademies.com for a copy.

#### Service Center Employees:

Service Center employees should refer to the *Vacation Benefits-Service Center Employees Policy* for information about their vacation benefits. Please contact your supervisor, the People Services Department or consult my.heritageacademies.com for a copy.

#### Holidays

<u>School Employees</u>: All regular full-time regularly scheduled school employees will be paid for designated holidays in which they are not required to work. For salaried employees, the holiday pay will be based on your regularly scheduled hours. For hourly employees, the holiday pay will be based on your straight-time pay rate and your regularly scheduled hours. In other words, your payment amount is based on the hours you would have worked had it not been a holiday (not to exceed eight hours). Regular full-time hourly employees must be regularly scheduled to work on the day of the holiday to be eligible for holiday pay. The following holidays are designated for school employees: New Year's Day, Memorial Day, Thanksgiving (2 days), Labor Day, and Christmas (3 days when Christmas falls on Friday, Saturday, Sunday or Monday; 4 days when Christmas falls on Tuesday, Wednesday or Thursday). The 4<sup>th</sup> of July is a designated holiday for Office Administrators and Assistant Secretaries.

<u>Service Center Employees</u>: Service Center employees should refer to the *Holiday Schedule-Service Center* for information about their holidays. Please contact the People Services Department for a copy.

School and Service Center part-time hourly employees working less than 35 hours per week, interns, casual and temporary employees are not eligible for holiday pay.

#### **Travel Expense Reimbursement**

NHA will reimburse employees for reasonable and necessary travel expenses incurred while on NHA business. Appropriate documentation of such expenses must be submitted to NHA to be eligible for reimbursement.

Reimbursements may cover meals, transportation, mileage, lodging and other allowed expenses. Employees are expected to spend the organization's money as if it were your own. To obtain a copy of NHA's Travel Policy, please contact your school secretary, the NHA Financial Services Department, or consult my.heritageacademies.com.

#### **Benefit Elections**

Participation in NHA's health, dental, vision, voluntary life insurance and flexible spending plans is voluntary. Coverage is not automatic. If you do not apply for health, dental, vision, voluntary life insurance or flexible spending plan coverage within 30 days of your hire date, you will not be able to enroll until the next open enrollment period. Open enrollment is held during the summer each year for all health and welfare plans. If elected, coverage for eligible employees is effective on the first day of the month following your hire date or enrollment through open enrollment. All health and welfare benefits end on the date that your employment with NHA ends, although you may be eligible to continue your group coverage under COBRA as discussed below. A People Services Representative can assist you in making the necessary arrangements for enrollment. Please see the Summary Plan Description for requirements and further details.

#### **Group Health and Dental Insurance**

NHA offers regular full-time and part-time (regularly scheduled 20 or more hours per week) employees comprehensive group health and dental insurance plans for those who subscribe to the plan, subject to the terms and conditions of the plan. Our group health and dental plans are self-insured, joint contribution plans that require you to make partial premium payments through payroll deduction. The amount of your partial premium payment may vary, depending on the type of plans we have and the type of coverage you select. Current cost information is available from the People Services Department. The health and prescription plans currently provide coverage for a variety of medical, surgical, hospital, and prescription expenses. The dental plan currently covers preventative, basic, major, and orthodontic (only for dependent children under the age of 19) services. Your deductible payments will be based on the period of September 1 through the end of the following August. If you add dependents (children or a spouse) to your coverage during designated enrollment periods or due to a qualifying event, you are required to supply supporting documentation to verify that each dependent is eligible to be enrolled. You have 60 days from enrollment to provide approved supporting documentation to the People Services Department or your dependent(s) may be removed from the plans.

#### **Group Vision Insurance**

NHA offers regular full-time employees a vision insurance plan for those who subscribe to the plan, subject to the terms and conditions of the plan. Our group vision plan is a contribution plan that requires you make premium

payments through payroll deduction. Current cost information is available from the People Services Department. The plan provides coverage for a variety of expenses and services related to eye exams and the purchase of prescription lenses or contacts. Coverage cannot be voluntarily terminated by the employee during the plan year. The benefit eligibility period is September 1 through the end of the following August.

#### **Flexible Spending Account**

The flexible spending account offers regular full-time employees an opportunity to annually set aside pretax dollars to pay for un-reimbursed medical expenses through a Health Flexible Spending Account and dependent care expenses through a Dependent Care Flexible Spending Account. Once you have made your election(s), your election is generally fixed for the remainder of the plan year. However, if you undergo a change in family status (as defined in the Summary Plan Description), you may change your elections, provided you do so within 30 days from the date of the change in family status. The plan year is from September 1 through the end of the following August.

NHA also offers a transportation and parking flexible spending account. You can enroll in, make changes to, or stop contributions to this plan at any time during the year. Your monthly elections may not exceed the monthly limits set forth by federal guidelines.

#### **Benefits Continuation (COBRA)**

The federal Consolidated Omnibus Budget Reconciliation Act (COBRA) of 1985 gives you and your qualified beneficiaries the opportunity to continue health, dental, vision insurance and/or the medical flexible spending plan coverage under NHA's health, dental, vision and flexible spending plans when a "qualifying event" would normally result in the loss of eligibility. Some qualifying events are resignation, termination from employment, death of an employee, a reduction in an employee's hours (including some types of leaves of absence), an employee's divorce or legal separation, a dependent child no longer meeting eligibility requirements, and an employee's eligibility for Medicare entitlement.

Under COBRA, you or your beneficiary pays for such coverage, as provided by applicable law, at NHA's group rates plus an administration fee.

#### Life Insurance/Accidental Death and Dismemberment

NHA offers regular full-time employees term Life Insurance and Accidental Death and Dismemberment (AD&D) Insurance coverage, subject to the terms and conditions of the plan. Each policy offers a benefit of 1 x base pay with a minimum benefit of \$25,000, up to a maximum benefit of \$50,000. Regular full-time employees also have the option of purchasing voluntary life insurance benefits for themselves and their dependents based on their families' insurance needs.

In the event of the employee's death, life insurance benefits will be paid to the employee's beneficiaries. If the employee's death was caused by an accident, the beneficiaries of the employee also will receive an additional benefit through the AD&D Insurance. AD&D Insurance also provides monetary benefits in the event of a serious injury. Please see the Summary Plan Description for eligibility requirements and further details.

#### **Short-Term Disability**

NHA's short-term disability (STD) plan provides qualifying employees with partial pay during weeks that employees do not work due to a non-work related illness, injury, or medical disability, subject to the terms and conditions of the plan. To be eligible for STD benefits, an employee must: (1) be a regularly scheduled, full-time employee (salaried or hourly) at the time of the leave; (2) have completed one year of active regular full-time service (35 hours or greater per week) in the year preceding the leave of absence or two years of active part-time service (20 hours or greater per week) in the two years preceding the leave of absence; (3) have been diagnosed by a medical

doctor as having a non-work related injury or condition, including pregnancy; and (4) be unable to work for more than three consecutive work days as advised by a medical doctor. This benefit is intended for active employees only. Should employment terminate, benefits would cease as of the date of termination. The benefit begins after a qualifying employee has been off from work for three or more consecutive working days. STD benefits are generally payable for up to a maximum of 26 weeks within a rolling 12 month period. No benefits are payable under this program if your illness or injury is connected to work you do for another employer. The benefit is paid out at 100% for the first week and 67% each week thereafter that an employee is deemed unable to work, up to the maximum of 26 weeks.

Benefits and eligibility may vary based on differences in laws from state to state. <u>New York</u> employees may qualify for STD reimbursement from the state of New York at the rate of 50% of your normal wage up to a maximum of \$170 per week. NHA will make up the difference from what the State of New York covers to equal 100% of your normal wages for the first week you are off work and 67% of your normal wages for the remaining time that you are disabled and under a medical doctor's care for up to 26 weeks.

Intermittent Short-Term Disability – Employees who qualify for an intermittent FMLA leave may be eligible for intermittent STD. An employee who is eligible: (1) must meet the conditions outlined in the STD policy; and (2) have been deemed unable to work intermittently by a medical doctor. The first 5 work days missed will be paid at 100%; each intermittent day after that will be paid at 67%. An intermittent leave may span over a maximum of 26 weeks within a rolling 12 month period. Intermittent STD must be approved by People Services prior to the start of the leave. Approval is contingent upon meeting all outlined factors.

<u>Reduced Work Schedule Short-Term Disability</u> – Employees who are released to return to work at a reduced work schedule or are placed on a medically related reduced work schedule by a physician will be paid regular pay for time worked and 67% of pay for the remainder of their regularly scheduled work day. An employee who is eligible: (1) must meet the conditions outlined in the STD policy; (2) must have been placed on a reduced work schedule per a medical doctor; (3) must not have exhausted his/her 26 weeks of STD eligibility.

Payments you may be entitled to under other NHA benefits/programs (i.e., vacation pay) are reduced by any STD benefits received. In other words, if you are a school employee who is eligible for vacation pay during a week you also are eligible for STD, you may not receive more than 100% of your regular rate of pay for that week. You may not collect <u>both</u> STD benefits and vacation pay. Salaried employees will receive 100% of pay during school vacation periods as outlined in the School Vacation Policy. If you are an hourly school employee, you are not eligible for STD benefits during school vacations listed in the Annual School Employment calendar.

School holidays will be paid out in accordance with the School Holiday policy. Full-time employees will be paid at 100% of pay for the holidays designated in the Holiday section of the Employee Handbook.

Service Center Employees: Full-time Service Center employees will be paid at 100% of pay for the holidays designated in the Holiday section of the Employee Handbook.

If you need STD benefits, please contact a People Services Representative as soon as possible.

#### Long-Term Disability

NHA provides all full-time regular employees with long-term disability (LTD) benefits after they have completed one year of active regular full-time service in the year preceding the leave of absence, subject to the terms and conditions of the plan. Under this program, employees who are disabled for more than 180 days because of injury or illness (in accordance with the definition of "disability" specified in the Summary Plan Description) are eligible to receive a benefit equivalent to 60% of their base monthly earnings to a maximum monthly benefit of \$10,000 (less other income benefits). The benefit may take effect after short-term disability benefits expire. In the event that you have a need for LTD benefits, please contact a People Services Representative 60 days prior to the date that the LTD should begin to ensure no lapse in benefits.

#### 401(k) Savings Plan

For eligible employees, NHA offers a 401(k) retirement savings plan to help eligible employees accumulate the assets the eligible employees will need for retirement on a pre-tax basis. NHA automatically enrolls eligible employees in the 401(k) plan after completing 45 days of employment at NHA, unless the employee declines participation, subject to the terms and conditions of the plan.

With the convenience of regular payroll deductions, you may contribute from 1-80% of your compensation per pay period, subject to the maximum annual limit. Once eligible, you can increase, decrease, start and stop your contributions to the plan at any time by contacting MassMutual at

https://wwwrs.massmutual.com/journey/Login.aspx or 1-800-743-5274. For investment advice at no cost to you, you can contact PearlStreet Investment Management at 1-800-456-2155.

Each year, NHA may, at its discretion, elect to match a portion of employee contributions. Currently the match is \$.50 for each \$1.00 employees contribute up to 6% of pay (maximum employer contribution is 3%). Please see the Summary Plan Description for eligibility requirements and further details.

Ohio employees are not eligible to participate in this benefit and participate in the SERS or STRS retirement plan instead. Likewise, in Georgia, employees in certain positions are not eligible to participate in this benefit and are required to participate to in the Teachers Retirement System of Georgia.

#### SERS/STRS

NHA employees who work in Ohio are required to participate in the state retirement plans: Ohio School Employees Retirement System (SERS) or Ohio State Teachers Retirement System (STRS). Employees are required to contribute 10% of pay pre-tax to either the SERS or STRS retirement systems depending on their job classification. Both Ohio Retirement systems require NHA to make a 14% employer match. Because Ohio requires these State Retirement Systems, employees in Ohio are not eligible to participate in NHA's 401(k) plan.

For more information about SERS contact 1-866-208-7377 or www.ohsers.org and for more information about STRS contact 1-888-535-4050 or <u>www.strsoh.org</u>.

#### TRS

NHA employees in certain positions who work in Georgia are required to participate in the Teachers Retirement System of Georgia (TRS). Employees are required to contribute 5.53% of pay pre-tax to TRS. TRS requires NHA to make a 10.28% employer match. Because Georgia requires this State Retirement System, employees in Georgia are no longer eligible to participate in NHA's 40l(k) plan.

For more information about TRS contact 1-800-352-0650 (outside metro Atlanta area) or 404-352-6500 (metro Atlanta area) or www.trsga.com.

#### **Tuition Reimbursement**

NHA recognizes that the skills and knowledge of its employees are critical to the success of the organization. NHA encourages employees to continue their education in subjects and fields related to their current and future NHA job responsibilities. NHA offers tuition reimbursement to encourage employees to maintain and improve job-related skills, achieve their professional development plans, achieve recertification, and enhance their ability to compete for reasonably attainable jobs within NHA.

NHA will reimburse classes at an accredited college or university that go toward a teaching certification or a higher teaching degree. NHA also will reimburse employees who are gaining expertise in other subject areas related to teaching (e.g., science, math, language arts, etc.) NHA may, in its discretion, approve a non-educational related

degree for secretaries and administrators. Service Center employees will be reimbursed for classes at an accredited college or university that go toward a job related degree, as approved by NHA.

All regular full-time employees who are regularly scheduled to work 35 or more hours per week and part-time employees who are regularly scheduled to work 20 or more hours per week are eligible. Employees must be employed with NHA on the dates the class starts and ends and the date reimbursement is paid. NHA reimburses Principals, Principals in Residence, Assistant Principals, Deans, Service Center Directors and Executives up to 100% of the tuition costs. All other eligible employees are reimbursed up to 50% of the tuition costs. Reimbursement includes up to 9 semester credit hours in each fiscal year if a grade of "B" or better is obtained in the class. NHA's fiscal year runs from July 1 – June 30. The starting date of the class determines which fiscal year into which the class falls. If the school or university does not grade the class on an A through F basis, it is within NHA's discretion to determine if the grade received is equivalent to a "B" or better.

If an employee ends employment with NHA within six months of a completed course for which he/she has received reimbursement, the employee will be responsible to reimburse NHA for the amount of tuition reimbursement received.

Once the class has ended, a tuition reimbursement form, an itemized billing statement from the college or university, and a copy of the grade report from the college or university must be sent to People Services. Reimbursement will be processed through Payroll and will be included on an employee's paycheck. NHA reserves the right to deny tuition reimbursement if proper paperwork is not turned in within six months after the class has ended. NHA does not reimburse for recreation/activity fees, book fees, late registration fees, or any other non-tuition related fees.

According to federal law, tuition reimbursement amounts paid in excess of \$5,250 in a single calendar year must be included as income and are subject to withholding taxes, unless the benefits qualify as a working condition fringe benefit.

Please refer to the Tuition Reimbursement Policy for additional details and for the Tuition Reimbursement Form.

#### **Employee Assistance Plan**

NHA has partnered with ENCOMPASS to offer employees and their family members free, confidential counseling services and tools to help cope with stress, depression, financial problems, legal issues, family situations and more. ENCOMPASS may be contacted via phone at 800-788-8630 or online at www.encompass.us.com (username: nha).

#### **Nursing Mothers**

NHA is committed to supporting nursing mothers by accommodating the mother who wishes to express breast milk during her workday when separated from her newborn child. Employees who are nursing will be provided with reasonable breaks to express breast milk as frequently as needed for up to one year after the birth of a child. NHA will provide a place for the break, other than a bathroom, that is shielded from view and free from intrusion. Employees will not be retaliated against for exercising their right to express milk in the workplace.

<u>Colorado Employees:</u> Employees who are nursing will be provided with reasonable breaks to express breast milk as frequently as needed for up to two years after the birth of a child.

<u>New York Employees</u>: Employees who are nursing will be provided with reasonable breaks to express breast milk as frequently as needed for up to three years after the birth of a child.

#### **Service Recognition**

NHA appreciates the loyalty and service of its employees. Our service award program recognizes this service and the value of the skill, knowledge and judgment developed through years of experience.

Each year, Service Awards are presented to employees whose years of service will or have reached the following milestones: 5 years, 10 years, 15 years, 20 years and 25 years. Qualifying service is based on your adjusted hire date as a regular employee.

#### **Excellence in Teaching Award**

The NHA Excellence in Teaching Award was created to recognize the accomplishments of individual teachers and the impact they have on student success. All active teachers of record are eligible for an NHA Excellence in Teaching Award in one of four categories: grades K-1; grades 2-5; grades 6-8; and co-curricular specialists. Being true to what we value – results and how we get results – we have aligned our criteria for the award with student learning and the behaviors articulated through our six individual success factors.

#### Charter School / Teacher Appreciation Week & American Education Week

To show our appreciation for our educators, school staff and Service Center staff, NHA celebrates Charter School/Teacher Appreciation Week in May and American Education Week in November. NHA recognizes the important contributions of charter schools and is committed to ensuring that all children receive the education they need to lead lives of purpose and success. We celebrate these weeks by providing our staff with luncheons and a variety of gifts.

#### **Discounts for Educators**

As a teacher or someone who works for an educational institution, you qualify for a number of discount programs offered through various companies. For details on the discounts offered, please visit the NHA Benefits website on my.heritageacademies.com.

#### **Adoption Assistance**

NHA believes strongly in the value of family; whether it is the family at work, at home, or at school. To support that belief, we offer up to \$3,000 to any eligible employee to help with the financial burden of the adoption procedure.

To qualify for this benefit, you must be an active regular full-time employee with at least one year of active regular full-time service in the year preceding the adoption. The benefit is provided upon proof that a legal adoption or placement is complete. The costs of adoption also must be substantiated. All necessary documents must be sent to the People Services Department for approval and processing. This benefit does not apply to legal fees for adoption of stepchildren.

## **Leaves of Absence**

During your employment with NHA, there may come a time when it is necessary for you to take a leave of absence. Your manager and the People Services Department must approve all leaves of absence. Please also reference the Short-Term and Long-Term Disability sections of this Handbook.

#### **Bereavement Leave**

All regular NHA employees, regardless of full-time or part-time status, are eligible for up to three days of paid bereavement leave for the death of an immediate family member during the five calendar days immediately following the death. The amount of bereavement pay is determined by your regular schedule. Immediate family is defined as you or your spouse's parents, stepparents, siblings, stepbrother/sister, children, stepchildren, spouse, grandparents, grandchildren, or any relative residing in your household.

If available, you may use vacation days for time off in case of a death outside of your immediate family. You are not eligible for bereavement benefits if you are on disability leave, workers' compensation leave, Family and Medical Leave, scheduled vacation, layoff, or unpaid leave of any kind. Your manager must approve all requests for bereavement leave. Should this policy differ from state law in the state where you work, state law will apply.

#### Family and Medical Leave (FMLA)

In accordance with the Family and Medical Leave Act (FMLA), NHA will provide eligible employees unpaid leave for certain family and medical reasons.

Up to 12 weeks of unpaid leave may be requested for the following reasons:

- To care for your child after birth or for a child placed with you for adoption or foster care
- To care for your spouse, minor or disabled adult child, or parent who has a serious health condition
- For a serious health condition that makes you unable to perform your job
- Because of a qualifying exigency arising out of a covered family member's (spouse, child or parent) active duty or call to active duty in the National Guard, the Reserves, or retired armed forces or Reserves, in support of a contingency operation ("qualifying exigent leave")

Up to 26 weeks of unpaid leave may be requested for the following reason:

To care for an injured or ill service member (spouse, child, parent or next of kin) who is (a) a member of the Armed Forces, the National Guard, or the Reserves and who is undergoing medical treatment, recuperation or therapy, is in outpatient status, or is otherwise on the temporary disability retired list, for a serious injury or illness incurred in the line of duty on active duty ("military caregiver leave"); or (b) a veteran of the Armed Forces, the National Guard, or the Reserves, and who is receiving treatment, recuperation, or therapy for a qualifying serious injury or illness incurred or aggravated in the line of duty provided the military service was within 5 years of the treatment, recuperation, or therapy.

A "serious health condition" involves:

- Inpatient care or follow up directly related to inpatient care
- A period of incapacity of more than 3 consecutive calendar days and two or more treatments from a health care provider
- A period of incapacity of more than 3 consecutive calendar days and at least one treatment by a health care provider resulting in a regimen of prescribed treatment
- Incapacity due to pregnancy, childbirth or for prenatal care
- Incapacity due to a chronic serious health condition that requires periodic visits for treatment over an extended period of time (e.g., asthma, epilepsy, etc.)

- A period of incapacity that is permanent/long-term due to a condition that may or may not be treatable (e.g., Alzheimer's, terminal stages of a disease, a severe stroke, etc.)
- Absences to receive multiple treatments for either (1) restorative surgery after an accident or injury, or
   (2) a condition that would likely result in incapacity for more than three consecutive calendar days in the absence of medical intervention or treatment, such as cancer (chemotherapy, radiation), severe arthritis (physical therapy), or kidney disease (dialysis)

Please note that a "serious health condition" does not include cosmetic surgery, the common cold, flu, earache, upset stomach, ulcers, headaches (other than migraines), orthodontia problems, periodontal disease, dental treatments, etc., unless they require inpatient care or if complications arise.

Qualifying Exigency Leave: An employee whose spouse, child or parent is on active duty or has been notified of an impending call or order to active duty in the National Guard, the Reserves, or retired armed forces or Reserves, in support of a contingency operation may take up to 12 weeks of leave for reasons related to or affected by the family member's call-up or service. Qualifying exigencies include such things as attending to legal and financial matters; providing urgent childcare or making alternative schooling or childcare arrangements; attending or participating in certain non-medical counseling sessions; attending military events; attending post-deployment and reintegration briefings; spending time with a military member who is on short term rest and recuperation leave during deployment; and addressing issues that arise due to short-notice call or order to active duty. (Son or daughter for this type of FMLA leave is defined the same as for child for other types of FMLA leave, except that the person does not have to be a minor.) This type of leave would be counted toward the employee's 12-week maximum of FMLA leave in a 12-month period. Employees requesting this type of FMLA leave must provide proof of the qualifying family member's call-up or active military service before leave is granted.

<u>Military Caregiver Leave</u>: This leave be taken up to 26 weeks in a single 12-month period for an employee whose spouse, son, daughter, parent or next-of-kin is (a) a member of the Armed Forces, the National Guard, or the Reserves and who is undergoing medical treatment, recuperation or therapy, is in outpatient status, or is otherwise on the temporary disability retired list, for a serious injury or illness incurred in the line of duty on active duty; or (b) a veteran of the Armed Forces, the National Guard, or the Reserves, and who is receiving treatment, recuperation, or therapy for a qualifying serious injury or illness incurred or aggravated in the line of duty provided the military service was within 5 years of the treatment, recuperation, or therapy. Next-of-kin is defined as the closest blood relative of the injured or recovering service member other than a spouse, child or parent.

Employees requesting this type of FMLA leave must provide certification of the family member or next-of-kin's injury, recovery or need for care. This type of leave must be taken within a single 12-month period beginning on the first date the employee takes leave. This is the only type of FMLA leave that may extend an employee's leave entitlement beyond 12 weeks to 26 weeks. Other types of FMLA leave are counted with this type of leave for purposes of totaling the maximum 26 weeks of leave in a 12-month period.

<u>Eligibility, Leave Entitlements and Limitations, Certification</u>: To be eligible for a leave, you must have worked for NHA for at least 12 months; you must have worked at least 1,250 hours in the 12-month period prior to requesting leave; and there must be at least 50 employees within 75 miles of where you work. Eligibility is determined for each reason for leave.

Except for military caregiver leave, which is counted using a single 12-month period, leave for all other reasons is counted using a rolling 52-week period that looks back from the date each leave is taken. Therefore, each time you take FMLA leave for a reason other than military caregiver leave, your remaining leave entitlement will be the balance of the 12 weeks that has not been used during the immediately preceding 52 weeks. Military caregiver leave is counted separate from all other leave reasons, except that under no circumstances may military caregiver leave and leave for other reasons exceed 26 weeks in a single 12 month period.

Where a husband and wife both work for NHA, they will be limited to a combined 12 weeks (not 24 weeks) of leave due to birth, adoption or placement of a child for foster care, to care for a parent with a serious health condition, or a combined 26 weeks of leave for military caregiver leave. Leave for a newborn or newly placed child

may be taken only within 12 months from the date of birth or placement. This leave may only be taken continuously.

Leave due to a serious health condition (either the employee's or a family member's), qualifying exigency leave, and military caregiver leave may be taken on in a single block of time or on a reduced schedule or an intermittent basis where medically necessary. Where an employee takes leave for planned medical treatment or leave on an intermittent or reduced schedule basis, the employee must attempt to work with NHA to schedule such leave so as not to disrupt operations.

When the need for leave is military-related or due to a serious health condition (the employee's or a family member's), you will be required to provide certification of the need for leave and its duration on a form supplied by NHA. Generally, this certification must be completed and returned within 15 days of when it was provided to you. NHA is not responsible for any charges incurred for completing the certification forms. While off work on leave, verification of continued need for leave and intent to return may also be requested. Failure to provide timely notice or the required certification may result in the leave being delayed or denied, or your absences being counted against our attendance policy or standards.

**Notice:** When possible, employees are required to give NHA at least 30 days' notice of the need for leave. When the need for leave becomes known less than 30 days in advance, you are required to notify NHA promptly when the need for leave becomes known. When the need for leave is unforeseeable, you or someone on your behalf must notify NHA as soon as possible. Subject to the above, you must generally comply with NHA's normal absence notification procedures.

Please provide as much information as possible about the reasons for leave so that we can properly and timely determine whether your leave qualifies under this policy. Calling in "sick" is not sufficient. If your leave has already been approved and you subsequently need time off for that same reason, you must specifically inform us that you are taking leave for that reason or tell us that you are taking "FMLA leave."

Failure to provide timely or sufficient notice may result in your leave being delayed or denied, or your absence may be counted against NHA's attendance expectations.

If you request leave, you will be informed whether you are eligible for leave under the FMLA. If you are, the notice will specify any additional information required as well as your rights and responsibilities. If you are not eligible, you will be notified and given a reason for the ineligibility. You will also be informed if leave will be designated as FMLA leave and if known, the amount of leave counted against your total leave entitlement. If NHA determines that the leave is not covered by the FMLA, you will be notified of this determination.

<u>Coordination with Other Forms of Leave</u>: All time off work that meets the definitions under FMLA will be charged against the FMLA allowances. For example, whenever a workers' compensation leave, a disability leave, or any other leave is due to a serious medical condition, all time off will be charged against your FMLA allowance. This provision applies even when you make no reference to FMLA at the time you request or take time off.

#### Benefit Continuation and Return from Leave:

To the extent they were in effect when the leave began, group health plan benefits are continued during FMLA leave on the same basis as for active employees. To maintain health care coverage during an approved leave, you must continue to pay the regular required employee contribution. If you fail to make the required contributions, health care coverage may be terminated and you will be allowed to continue coverage under COBRA. Employees who fail to return from an FMLA leave or who terminate within 30 days of returning may be required to reimburse NHA for any insurance premiums paid on their behalf during a leave. Non-health benefits (e.g., life, disability, vacation accrual, etc.) will be continued during a FMLA leave. You will not lose any benefit that was earned prior to your leave.

When you return from FMLA leave, you will be reinstated to the same job or to an equivalent job with the same pay, benefits, and terms and conditions of employment. For school employees, the determination of how an employee is to be restored to "an equivalent position" upon return from FMLA leave will be made on the basis of established written policies. Reinstatement to the same or an equivalent position may not be possible if the position or benefits would have been changed or eliminated if you had not been on FMLA leave. In that case, you will be given the same consideration as if you had not taken the leave. Employees returning from a leave due to their own serious health condition may be required to provide NHA with a fitness for duty certification.

**Designation of Leave:** If you return from any period of absence that has not been designated as FMLA leave and you wish to have the leave counted as FMLA leave, you must notify NHA within 2 business days of returning to work that the leave was for FMLA reasons. Failure to provide the necessary notice will prevent any later FMLA protection for that absence. For additional information, contact the People Services Department.

**Kev Employees:** Certain employees may not be reinstated to the same or an equivalent job after an FMLA leave if reinstatement would cause NHA substantial and grievous economic injury. A "key employee" is an employee who is among the highest paid 10 percent of salaried employees employed by NHA within 75 miles of the employee's work site. If you are a "key employee" under this definition and you request FMLA leave, NHA will notify you in writing concerning your reinstatement.

<u>Instructional Employees</u>: Instructional employees requesting FMLA leaves may be granted leaves in accordance with special rules applicable to such employees.

**Termination of Leave:** Your FMLA leave and accompanying benefits will end under any of the following circumstances:

- Your employment would have ended if you had not taken FMLA leave.
- You inform NHA of your intent not to return from leave.
- You fraudulently obtain FMLA leave.
- You fail to return from leave or continue on leave after exhausting your FMLA leave entitlement, unless
  permission to do so has been granted in writing by the Director of People Services.
- You falsify any information or document in connection with the leave request.
- You work elsewhere while on leave without prior approval of NHA.

This is merely a brief description of the Family and Medical Leave Act. To obtain a full copy of the Act, please contact the People Services Department.

Employees in Wisconsin may be eligible for leaves under the Wisconsin Family and Medical Leave Laws. Employees in Wisconsin should consult my.heritageacademies.com to review a copy of the Wisconsin Family and Medical Leave Act policy.

Employees in Louisiana may be eligible for a leave on account of pregnancy/childbirth in accordance with Louisiana law. Employees in Louisiana should consult my.heritageacademies.com to review a copy of the Louisiana Pregnancy/Childbirth Leave policy.

#### Unpaid Personal Leave (not covered by FMLA)

You may request an unpaid leave of absence for personal needs **due to extreme situations over which you have no control, such as a medical or family emergency not covered by FMLA**. The leave must be taken continuously and only 1 unpaid personal leave may be granted every 3 years from the date the previous leave began. Once an employee has missed 3 or more consecutive days from work, a formal leave should be requested. The leave will begin effective the first day the employee missed. **Requests for unpaid personal leaves must be approved by the People Services Department.**  Unpaid personal leaves must be substantiated with documentation of the medical necessity or the nature of the emergency. The maximum time an employee may be granted an unpaid personal leave is the time he/she is deemed medically unable to work up to a maximum of 6 weeks. If the leave is for a family emergency, the time granted will be reviewed based on the need up to a maximum of 6 weeks.

Employees must request a leave at least 2 weeks prior to the time the leave is to begin. However, if the leave is due to an emergency, the employee is to notify their manager immediately and to submit a written leave request to People Services within 3 days. Employees are expected to return to work immediately at the end of their leave. If you do not return to work within 3 days after the end of your leave, NHA will consider you to have resigned.

Your elected employee benefits will continue during your leave. You are responsible to reimburse NHA for any missed health premium deductions when you return to work.

Reinstatement cannot be guaranteed to employees returning from a personal leave. However, NHA will attempt to give you the same position or similar position, subject to budgetary restrictions, our need to fill vacancies and our ability to find qualified replacements.

#### **Military Leave**

Leaves of absence without pay for Military or Reserve duty and Military Family are granted to regular full-time and part -time regular employees in accordance with state and federal laws. If you are a reservist or member of the National Guard, you are also granted time off without pay for required training. After returning from military service, you will have rights to reinstatement, vacation, compensation, and length of service as required by law.

#### Jury and Witness Duty

If you are a regular full-time employee who is summoned to jury duty, NHA continues your base wage rate during your active period of jury duty up to a maximum of 15 days per calendar year. You are also permitted to retain any allowance you receive from the court for such service. If you are not a regular full-time employee, you are given time off without pay while serving jury duty. Employees out on any leave are ineligible for jury duty pay.

You must show your jury duty summons to your manager as soon as possible so that arrangements can be made to accommodate your absence. You are expected to report to work whenever the court schedule permits. All employees are allowed unpaid time off if subpoenaed to appear in court as a witness. Volunteer service on a jury, appearances as a voluntary witness, or as a plaintiff or defendant is not covered by this policy. Should this policy differ from state law in the state where you work, state law will apply.

## **Rules of Conduct**

#### Working Together

NHA expects high standards of conduct and performance from all employees. Behavior or performance that is below these standards may result in disciplinary action, up to and including termination from employment, as appropriate. In addition to the policies contained in other sections of this Handbook, the standards listed below represent examples of situations that may occur. NHA may take disciplinary action, up to and including termination of employment, as deemed appropriate if circumstances arise which are not specifically covered by this Handbook.

Violations of the following policies are so serious that, following a thorough investigation, they could result in corrective action up to and including termination from employment for the first offense:

- Violence or Threat of Violence including deliberate property damage, threatening conduct or fighting.
- Dangerous Behavior carrying weapons; using, possessing or distributing intoxicants or illegal drugs; reporting to work under the influence of intoxicants or illegal drugs; refusing to take a drug or alcohol screen when requested; failing to report a work related injury.
- Dishonesty/Unethical Behavior- theft or concealment of stolen property; falsifying NHA records or any employment-related information; originating or spreading false statements concerning employees or NHA; unauthorized disclosure of confidential information.
- Inappropriate Business Behavior gambling; abusive, profane, or threatening language; insubordination; violation of discrimination or harassment policies; misuse of confidential information; conducting personal business during work time; excessive absenteeism or tardiness; showing disrespect for co-workers; improper use of NHA technology or other NHA resources.

#### **Corrective Action**

Corrective action ranges from a documented verbal reminder to immediate termination from employment, depending on certain factors, including, but not limited to, the seriousness of the violation and whether it is a first violation or a recurrence. NHA may suspend an employee from work while investigating a situation to determine the appropriate course of action.

A violation of policy or standards may lead to one of the following disciplinary actions, according to the seriousness of the violation:

#### 1. Documented Verbal Reminder

Your manager will meet with you to discuss a problem with your performance or business behavior. This will be documented in your personnel file.

#### 2. Written Warning

Your manager will put a performance or behavioral issue in writing. This will be reviewed with you. Corrective actions will be discussed. Both you and your manager will sign the written warning. The written warning may include a suspension from work.

#### 3. Final Written Warning

If your performance or behavior does not improve and you have not taken the appropriate action to address the issues, a final written warning will be given. This will be reviewed with you. Corrective actions will be addressed. Both you and your manager will sign the final written warning. The final written warning may include a suspension from work. This is the last step prior to termination from employment.

#### 4. <u>Termination from Employment</u>

Some policy violations may result in immediate termination from employment without any prior warnings.

In addition, in the event of unacceptable or insufficient job performance, NHA may provide coaching or a performance improvement plan to the employee.

NHA has no obligation to use any one or more of these forms of discipline prior to discharging an employee. Any or all of these disciplinary actions may be omitted as NHA deems appropriate, in its sole discretion. Nothing in this policy changes or modifies the at-will employment status of employees.

## ATTACHMENT 9.3

### Licensure Training – Closing Duties Certification of Fiscal Agent, Officer or Services Provider

I, <u>Kathleen (Schmidt) Schaaphok</u>, hereby certify that I am licensed as provided for under R.C. 3301.074 (license attached and made a part of this certificate), or, that I have completed not less than sixteen (16) hours of continuing education classes, courses or workshops in the area of school accounting as approved by the Sponsor, completed an additional twenty-four (24) hours of continuing education classes, courses or workshops in the area of school accounting as approved by the Sponsor, or have completed or will complete eight (8) hours of continuing education classes, courses or workshops in the area of school accounting as approved by the Sponsor during each subsequent year. Official transcripts, licenses and certificates for all hours of training are attached as Exhibit A to this Certification.

If School obtains a new fiscal agent, officer or service provider, then a new Certification of Fiscal Agent, Officer or Service Provider must be signed and sent to Sponsor within three (3) business days.

The written agreement between Fiscal Agent, officer or Services Provider ("Treasurer") and the School requires that Treasurer be retained to also provide all closing responsibilities required by Sponsor, School and ODE should School close for any reason.

Fiscal Agent, Officer or Services Provider

by Schaaplul

-30-12

Date

School Kevin a. Kobie

06,

Witnesses

Witnesses

## Exhibit A to Attachment 9.3

Licenses, Transcripts, Certificates for all hours of training

| Stat   | e of Ohio EMPLOYE   |
|--|---|
| Departmer  | nt of Education   |
| 5 Year School Treasurer  |   |
| This License Awarded To: <b>KATHLEEN E. S</b><br>License Number: OH3103299   | SCHMIDT<br>Issue Date: 09/27/2  |
|  | FROM: 07/01/2011 - 06/30/2  |
| The holder of this credential, having s<br>prescribed by The State Board of Educ<br>or serve in the area(s) listed below for | atisfactorily completed the requirements<br>cation, is authorized to teach the subject(s)<br>• the period specified.  |
| School Treasurer   | ça  |
|  |   |
|  |   |
|  |   |
|  | $\sim \sim $                     |
| Superintendent of Public Instruction 12  | 202372 <u>Michael Jacuyer</u>   |
|  |   |
|  | 如此,而且此,也是是是一些,也是是一些,也是是一些,也是是一些,也是是一些,也是是一些。<br>一般,我们们是一些,你们们们们是一些,你们们们们是一些,你们们们们们是是                                  |
|  | e of Ohio   |
| Departmen  | nt of Education   |
| 5 Year School Treasure   | r License `   |
| This License Awarded To: <b>KATHLEEN E.</b><br>License Number: OH3103299   | SCHMIDT<br>Issue Date: 09/27/   |
|  | FROM: 07/01/2011 - 06/30/   |
| The holder of this credential, having sprescribed by The State Board of Edu or serve in the area(s) listed below fo          | satisfactorily completed the requirements<br>cation, is authorized to teach the subject(s)<br>r the period specified. |
| School Treasurer   |   |
|  |   |
|  |   |
|  |   |
|  |   |
|  |   |
| Superintendent of Public Instruction   | 202372 Michay Januf   |

.

## ATTACHMENT 9.4 Treasurer's Bond



Crime Program Certificate of Coverage

## NATIONAL HERITAGE ACADEMIES

Certificate No. 006

Master Bond No. 104713358

This is to Certify

That the following Fiscal Agent of the Pathway School of Discovery, 173 Avondale, Dayton, OH 45404 is covered under a Crime Program through the master policy of the National Heritage Academies provided by the Travelers Casualty & Surety Company of America. The coverage and limit(s) are set forth below:

Insured:

Kathleen Schmidt

**Terms of Coverage:** 

Policy Term: June 30, 2010 to August 1, 2011

Coverage:

A-Scheduled

Limit of Liability \$25,000.

Deductible \$500.

Policy Term Premium: \$108.50

Travelers Casualty & Surety Company of America

hat Thew Van Conant my

Matthew VanConant, Authorized Representative

## ATTACHMENT 9.5

Projected Budget/Financial Plans Five-Year Projection of Operational Revenues and Expenditures under R.C. §5705.391 Amount of Per-Pupil Expenditure Assumed ODE Per-Pupil Expenditure Worksheet Financial Information Summary (FIS) CCIP, SOES, OEDS-R Forms

#### FIVE YEAR BUDGET PLAN

Community School Budget:

School District:

Fund Name:

## PATHWAY SCHOOL OF DISCOVERY PATHWAY SCHOOL OF DISCOVERY

IRN#

County:

000138

ty: <u>MONTGOMERY</u>

ALL FUNDS

|                                   | 2012-13      | 2013-14  | 2014-15   | 2015-16                                | 2016-17             |
|-----------------------------------|--------------|--|---|--|---------------------|
|                                   | FY - 1       | FY - 2   | FY-3  | FY - 4                                 | FY - 5              |
| Receipts                          | a surra      |  |   |  |                     |
| Beginning Cash Bal.               | ·            | 124.00 × 124   | 1   |  |                     |
| School Foundation                 | \$ 4,658,100 | \$ 4,658,100   | \$ 4,658,100  | \$ 4,658,100                           | \$ 4,658,10         |
| State Sub-Grant                   | 5,300        | 5,300  | 5,300   | 5,300                                  | 5,30                |
| Federal Sub-Grant                 | 244,700      | 244,700  | 244,700   | 244,700                                | 244,70              |
| CCIP Funds                        | 440,500      | 440,500  | 440,500   | 396,100                                |                     |
| Misc from Local Sources           | 61,600       | 61,600   | 61,600  |  | 396,10              |
| Other (Transfers Between Funds)   |              | 01,000   | 01,000  | 61,600                                 | 61,60               |
| Other (Contribution from NHA)     | 218,100      | 308,700  | 405,300   |  | <u> </u>            |
| Total Receipts                    |              | C 500,700  | \$ 5,815,500  | 503,800                                | 604,900             |
| (including start-up)              | 0.020,000    | <u>a</u> ⇒o'\ io'an∩   | 3 5,815,500   | \$ 5,869,600                           | \$ 5,970,700        |
|                                   |              |  |   |  |                     |
| 1100 Reg. Instruct.               |              |  |   |  |                     |
| 100 Salaries-Instruct             | A 101 000    |  |   |  | . *                 |
| 200 Retirement/Insurance Benefits | \$ 1,431,800 |  |   | \$ 1,549,700                           | \$ 1,591,300        |
| 400 Purchased Services            | 637,300      | 654,100  | 671,400   | 689,300                                | 707,700             |
| TO FUICIASED SERVICES             | 193,200      | 193,500  | 195,700   | 164,700                                | 167,000             |
| 500 Supplies and Materials        | 167,400      | 162,200  | 162,400   | 151,500                                | 151,700             |
| 600 Capital Outlay                |              | -  | -   | -                                      |                     |
| 800 Other Objects                 | 19,700       | 19,800   | 19,900  | 20,000                                 | 20,100              |
| Total Reg. Instruct.              | \$ 2,449,400 | \$ 2,499,500   | \$ 2,558,600  | \$ 2,675,200                           | \$ 2637 800         |
| na series a                       |              |  |   |  |                     |
| 1200 Spec. Instruct.              | ter di di    |  | ······································  |  |                     |
| 100 Salaries                      | \$ 133,600   | \$ 137,600   | \$ 141,700  | \$ 146,000                             | \$ 150.300          |
| 200 Retirement/Insurance Benefits | 62,300       | 64,200   | 66,100  | 68,000                                 |                     |
| 100 Purchased Services            | 16,600       | 17,000   | 17,400  |  | 70,100              |
| 500 Supplies and Materials        | 3,300        | 3,400  | 3,400   | 17,800                                 | 18,200              |
| 500 Capital Outlay                | 0,000        |  | 3,400   | 3,500                                  | 3,600               |
| 100 Other Objects                 | •            |  |   | · •                                    | -                   |
| Iotal SE Instruct                 | \$ 215 800   |  |   | -                                      |                     |
|                                   | \$ 215,800   | \$ 222,200   | \$ 228,600  | \$ 235,300                             | \$ 242,200          |
| 2100 Pupil Spprt                  |              |  |   |  |                     |
| 100 Salaries                      |              | ······································   |   |  |                     |
| 200 Benefits                      | \$           | \$ -   | <u>s</u> -  | ·\$                                    | \$ -                |
| 100 Purchased Services            |              | -  | -   | <b>n</b> 2                             |                     |
| 00 Purchased Services             | 221,800      | 226,200  | 230,800   | 235,400                                | 240,100             |
| 00 Supplies and Materials         | -            | •  |   | -                                      |                     |
| 00 Capital Outlay                 | -            | -  | -   |  | -                   |
| 00 Other Objects                  | -            | -  | -   |  |                     |
| otal                              | \$ 221,800   | \$ 226,200   | \$ 230,800  | \$ 235,400                             | \$ 240,100          |
|                                   |              |  |   | •••••••••••••••••••••••••••••••••••••• | Φειακατορίου        |
| 200 Support Srvs.                 |              |  |   |  |                     |
| 00 Salaries                       | \$ 223,500   | \$ 230,400   | \$ 237,500  | \$ 244,800                             | e                   |
| 00 Retirement/Insurance Benefits  | 80,400       | 82,800   | Contraction of the local data and the second data and the second data and the second data and the second data a |  | \$ 252,300          |
| 00 Purchased Services             | 156,760      | 157,760  | 85,400  | 88,000                                 | 90,700              |
| 00 Supplies and Materials         | 12,000       | the second s | 158,860   | 159,960                                | 161,060             |
| 00 Capital Outlay                 |              | 12,200   | 12,400  | 12,500                                 | 12,700              |
| 00 Other Objects                  | · · · · ·    | -  |   | -                                      | -                   |
| otal                              |              | •  | -   | - 1                                    | -                   |
|                                   | \$ 472,660   | <b>\$</b> 483,160  | \$ 494,160  | \$ 505,260                             | \$ 516,760          |
| 300 Gogoral Admit                 |              |  |   |  |                     |
| 300 General Admin                 |              |  |   |  |                     |
| 00 Salaries                       | \$ -         | \$ ~   | \$ ~  | \$ -                                   | \$ -                |
| 00 Retirement/Insurance Benefits  | -            |  | -   |  |                     |
| 00 Purchased Services             | 49,800       | 50,000   | 50,100  | 50,200                                 |                     |
| 00 Supplies and Materials         | -            |  |   | ······································ | 50,300              |
| 00 Capital Outlay                 | -            |  |   |  |                     |
| 00 Other Objects                  | 32,700       | 33,400   | 34,000  | 24 700                                 | -                   |
| otal                              | \$ 82,500    |  |   | 34,700<br>\$ 84,900                    | 35,400<br>\$ 85,700 |

| 2400 Admin.   |  | . <b> </b>                            | <u> </u>  |  |  |
|---|--|---------------------------------------|---|--|--|
| 100 Salaries  | -  |                                       |   |  |  |
|   | \$ 161,700   |                                       | \$ 172,100  | \$ 177,500   | \$ 183,10  |
| 200 Retirement/Insurance Benefits   | 63,800   | ++,444                                | 67,800  | 69,900   |  |
| 400 Purchased Services  | 289,100  | 289,700                               | 290,400   |  | and the second |
| 500 Supplies and Materials  | 14,300   |                                       | and the second se |  |  |
| 600 Capital Outlay  | -  | -                                     | 14,000  | 10,200   | 13,50  |
| 800 Other Objects   | 8,100  |                                       | 9 500   | 0.000  | <u> </u>   |
| Total   | and the second statement of th |                                       | -,  |  | 8,80   |
|   | <u>\$ 537,000</u>  | \$ 545,300                            | \$ 553,700  | \$ 562,200   | \$ 570,900   |
| 2500 Fiscal   |  |                                       |   |  |  |
|   |  |                                       |   |  | 1  |
| 100 Salaries  | -  |                                       | ·   | -  |  |
| 200 Retirement/Insurance Benefits   |  | 1                                     | -   |  | +  |
| 400 Purchased Services  | 66,100   | 66,100                                | 66,100  | 66,100   |  |
| 500 Supplies and Materials  | -  |                                       |   | 00,100   | 66,100   |
| 600 Capital Outlay  |  |                                       | -   | *  |  |
| 800 Other Objects   |  |                                       | ~   |  | -  |
| Total   |  | -                                     | -   |  | -  |
| - Coulter and a second s | \$ 66,100  | \$ 66,100                             | \$ 66,100   | \$ 66,100  | \$ 66,100  |
| 2000 D  |  |                                       | 1   | T  | 1  |
| 2600 Business   | 1  | 1                                     |   | †  |  |
| 100 Salaries  | -  | 1                                     |   | 1  |  |
| 200 Retirement/Insurance Benefits   | 1  | 1                                     |   |  |  |
| 400 Purchased Services  | 2,900  | 0.000                                 |   | -  |  |
| 500 Supplies and Materials  |  | 2,900                                 | 2,900   | 2,900  | 2,900  |
| 600 Capital Outlay  | <u> </u>   | la se a de la sera de                 | •   | -  | -  |
| 800 Other Objects   | <u>.</u>   |                                       |   | · .  | -  |
|   | • • •  |                                       | -   | -  |  |
| Total   | \$ 2,900   | \$ 2,900                              | \$ 2 900  | \$ 2,900   | \$ 2,900   |
|   |  | 1                                     | 1   |  | <u></u>  |
| 2700 Operations   | T  |                                       | [   | <u> </u>   |  |
| 100 Salaries  | 1 .  |                                       |   | <u> </u>   |  |
| 200 Retirement/Insurance Benefits   | <u> </u>   |                                       |   | · · · · · ·  | -  |
| 400 Purchased Services  | 4.044.000  |                                       |   | -  | -  |
| 500 Supplies and Materials  | 1,014,900  | 1,023,300                             | 1,028,000   | 1,032,700  | 1,037,500  |
|   | 8,400  | 8,500                                 | 8,600   | 8,700  | 8,800  |
| 600 Capital Outlay  | -  | -                                     |   |  |  |
| 800 Other Objects   | 5,500  | 5,500                                 | 5,500   | 5.500  | C CAA  |
| Total   | \$ 1,028,800   | \$ 1.037 300                          | \$ 1042100  | \$ 1,046,900   |  |
|   |  |                                       | equest (rotz, roo.  | a 1,040,900  | <u>⊅ 1,051,800</u>   |
| 2800 Transport.   | 1  |                                       |   |  |  |
| 100 Salaries  |  | · · · · · · · · · · · · · · · · · · · |   |  |  |
| 200 Retirement/Insurance Benefits   |  | · · · · · · · · · · · · · · · · · · · |   |  |  |
| 400 Purchased Services  |  |                                       |   |  |  |
| 500 Supplies and Materials  |  |                                       |   |  |  |
| 600 Capital Outlay  |  |                                       |   |  |  |
|   |  |                                       |   |  |  |
| 800 Other Objects   |  |                                       |   |  |  |
| Total   | <b>\$</b>  | <b>S</b>                              | <b>\$</b>   | S  |  |
|   |  |                                       |   |  | <u> </u>   |
| 2900 Central Services   | [  |                                       |   |  |  |
| 100 Salaries  |  | ·····                                 |   |  |  |
| 200 Retirement/Insurance Benefits   | -  | <u> </u>                              |   |  | -  |
| 400 Purchased Services  | -  | -                                     | -   | ~  |  |
| FOO FUICILASED SERVICES   | 225,140  | 225,140                               | 225,640   | 225,640  | 225,540  |
| 500 Supplies and Materials  | i tra pri  |                                       |   | 220,040  | 620,040  |
| 500 Capital Outlay  | · ·  | -                                     |   |  |  |
| 800 Other Objects   |  |                                       |   |  |  |
| Total   | \$ 225,140   |                                       | CONTRACTOR OF THE OWNER   | -  | -  |
|   |  |                                       | . <b>⊅</b> 225,640  | \$ 225,640   | \$ 225,540   |
| 3100 Food Service   |  |                                       |   | 1  |  |
| 100 Salaries  |  |                                       |   |  |  |
|   | \$   | <u>\$</u> -                           | \$ -  | \$ -   | \$ <u> </u>  |
| 200 Retirement/Insurance Benefits   | -  | · •                                   |   |  |  |
| 100 Purchased Services  | 291,600  | 292,100                               | 292,100   | 202.400  | -  |
| 500 Supplies and Materials  |  |                                       |   | 292,100  | 292,100  |
| 600 Capital Outlay  |  |                                       |   |  | *  |
| 300 Other Objects   | 01.000   | -                                     |   |  | -  |
| otal  | 34,600   | 35,600                                | 36,700  | 37,700   | 38,800   |
| And I wanted and the second   | \$ 326,200   | \$ 327,700                            | \$ 328,800  |  | \$ 330,900   |
|   | I  |                                       |   | A CONTRACTOR OF A CONTRACTOR A |  |
| BRAND TOTAL   | \$ 5,628,300   | \$ 5,718,900                          | 5 5 846 500   | \$ 5,869,600   | \$ 5,970,700   |
|   |  |                                       |   |  |  |

# **OFFICE OF COMMUNITY SCHOOLS**

|               |   | Per l     | Public Expenditur            | e Wo            | rksheet           |  |  |  |  |
|---------------|---|-----------|------------------------------|-----------------|-------------------|--|--|--|--|
| Schoo         | School Name: Pathway School of Discovery  |           |                              |                 |                   |  |  |  |  |
| Spons         | or:   |           | Lucas County                 |                 |                   |  |  |  |  |
| IRN #:        | 000138  |           | Cou                          | inty: <u>Mo</u> | ntgomery          |  |  |  |  |
|               |   |           | For Fiscal Year 2            | 2013            |                   |  |  |  |  |
| Basic I       | Formula '13   | x         | Estimated # of Students      | Ξ               | '13 Formula Total |  |  |  |  |
| \$5,653       | 3.00  | x         | 752                          | =               | \$4,251,056       |  |  |  |  |
| Α.            | Estimated R   | evenue    | from PBA* Family Services    |                 | \$305,144         |  |  |  |  |
| 8.            | Estimated Re  | venue fr  | om Special Education Weighte | d Funds=        | \$101,900         |  |  |  |  |
| C.            | Sum of Estim<br>(Totals from A  |           | 13 Annual Revenues           | =               | \$4,658,100       |  |  |  |  |
|               |   | , , , , , |                              |                 |                   |  |  |  |  |
| Total F       | Total Per Pupil Expenditure = <u>Sum of Estimated FY '13 Annual Revenues</u> (Box D)<br>Total # of Students |           |                              |                 |                   |  |  |  |  |
| D.<br>Total F | Total of Stude<br>Per Pupil Expend  |           | \$6,194                      |                 | 752               |  |  |  |  |

\* Poverty Based Assistance

## **ATTACHMENT 10.1** Insurance Binders, Declaration Sheets

| Client#: 51626 NATIHERI  |                            |  |                                |  |                             |  |                    |   |
|--|----------------------------|--|--------------------------------|--|-----------------------------|--|--------------------|---|
| ACORD CERT   | 'IFI(                      | CA                                       | TE OF LIABI                    | LITY INS   | URAN                        | CE [   | DATE (MI<br>8/27/2 | M/DD/YYYY)<br><b>2012</b>               |
| THIS CERTIFICATE IS ISSUED AS A MATTER OF INFORMATION ONLY AND CO  |                            |  |                                |  |                             |  |                    |   |
| CERTIFICATE DOES NOT AFFIRMATIVELY OR NEGATIVELY AMEND, EXTEND OR ALTER THE COVERAGE AFFORDED BY THE POLICIES<br>BELOW. THIS CERTIFICATE OF INSURANCE DOES NOT CONSTITUTE A CONTRACT BETWEEN THE ISSUING INSURER(S), AUTHORIZED<br>REPRESENTATIVE OR PRODUCER, AND THE CERTIFICATE HOLDER. |                            |  |                                |  |                             |  |                    |   |
| IMPORTANT: If the certificate holder is<br>the terms and conditions of the policy  | olicies may require an end | cy(ies) must be er<br>lorsement. A state | ndorsed. If Sl<br>ment on this | JBROGATION IS WAIVE<br>certificate does not co   | D, subj<br>nfer rigł        | ect to<br>nts to the                         |                    |   |
| certificate holder in lieu of such endor<br>PRODUCER   | semer                      | nt(s).                                   |                                | CONTACT Judi He                                  | oth                         |  |                    |   |
| HUB International Midwest  |                            |  |                                | NAME: 5001 He<br>PHONE<br>(A/C, No, Ext): 616-23 |                             | FAX  | 616-23             | 33-4110                                 |
| Commercial   |                            |  |                                | E-MAIL<br>ADDRESS: judi.hea                      | th@hubinte                  |  | 010 -              |   |
| P.O. Box 2167  |                            |  |                                | ADDRESS. J                                       | and the second second       | FORDING COVERAGE                             |                    | NAIC #                                  |
| Grand Rapids, MI 49501-2167  |                            |  |                                | INSURER A : Citizen                              |                             |  |                    |   |
| INSURED National Heritage Academ   | vice la                    |  |                                | INSURER B : Westch                               |                             |  |                    |   |
| 3850 Broadmoor S E, #20  |                            | IC.                                      | _                              | INSURER C : Travele                              | ers Casualty                | / & Surety Co                                |                    |   |
| Grand Rapids, MI 49512   | •                          |  | -                              | INSURER D :                                      |                             |  |                    |   |
| ,,   |                            |  | F                              | INSURER E :                                      |                             |  |                    |   |
| COVERAGES CER  | TIFIC                      |  | NUMBER:                        | INSURER F :                                      |                             | REVISION NUMBER:                             |                    |   |
| THIS IS TO CERTIFY THAT THE POLICIES   |                            |  |                                | E BEEN ISSUED TO                                 |                             |  | POLICY             | PERIOD                                  |
| INDICATED. NOTWITHSTANDING ANY RE<br>CERTIFICATE MAY BE ISSUED OR MAY I<br>EXCLUSIONS AND CONDITIONS OF SUCH   | Pertai<br>I polic          | IN, T<br>CIES.                           | HE INSURANCE AFFORDED          | BY THE POLICIES                                  | DESCRIBED I<br>BY PAID CLAI | HEREIN IS SUBJECT TO                         |                    |   |
| INSR<br>LTR TYPE OF INSURANCE  | ADDL S                     | SUBR<br>WVD                              | POLICY NUMBER                  | POLICY EFF<br>(MM/DD/YYYY)                       | POLICY EXP<br>(MM/DD/YYYY)  | LIMIT  | S                  |   |
| A GENERAL LIABILITY  |                            |  | ZDI1105381                     | 08/01/2012                                       | 08/01/2013                  | EACH OCCURRENCE                              | \$1,000            | 0,000                                   |
| X COMMERCIAL GENERAL LIABILITY   |                            |  |                                |  |                             | DAMAGE TO RENTED<br>PREMISES (Ea occurrence) | \$300,0            |   |
| CLAIMS-MADE X OCCUR  |                            |  |                                |  |                             | MED EXP (Any one person)                     | \$15,00            |   |
|  |                            |  |                                |  |                             | PERSONAL & ADV INJURY                        | \$1,000            |   |
|  |                            |  |                                | ****   |                             | GENERAL AGGREGATE<br>PRODUCTS - COMP/OP AGG  | \$2,000<br>\$2,000 |   |
|  |                            |  |                                |  |                             | PRODUCTS - COMPIOP AGG                       | \$2,000            | ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,, |
|  |                            |  | ADI1105817                     | 08/01/2012                                       | 08/01/2013                  | COMBINED SINGLE LIMIT<br>(Ea accident)       | s1,000             | 0.000                                   |
| X ANY AUTO   |                            |  |                                |  |                             | BODILY INJURY (Per person)                   | \$                 | ,                                       |
| ALL OWNED SCHEDULED<br>AUTOS AUTOS   |                            |  |                                |  |                             | BODILY INJURY (Per accident)                 | \$                 |   |
| X HIRED AUTOS X NON-OWNED AUTOS  |                            |  |                                |  |                             | PROPERTY DAMAGE<br>(Per accident)            | \$                 |   |
|  |                            |  |                                |  |                             |  | \$                 |   |
| A X UMBRELLA LIAB X OCCUR  |                            |  | U7I1112271                     | 08/01/2012                                       | 08/01/2013                  | EACH OCCURRENCE                              | \$15,00            |   |
| EXCESS LIAB CLAIMS-MADE  |                            |  |                                |  |                             | AGGREGATE                                    | \$15,00            | 0,000                                   |
| DED X RETENTION \$10000  |                            |  | ZDI1105381                     | 08/01/2012                                       | 09/04/2042                  | WC STATU- OTH-                               | \$                 |   |
| A AND EMPLOYERS' LIABILITY   |                            |  | Ohio Employers                 | 00/01/2012                                       | 00/01/2013                  | E.L. EACH ACCIDENT                           | s1,000             | 000                                     |
| A ANY PROPRIETOR/PARTNER/EXECUTIVE<br>OFFICER/MEMBER EXCLUDED?   | N/A                        |  | Stop Gap Coverage              |  |                             | E.L. DISEASE - EA EMPLOYEE                   |                    | •                                       |
| If yes, describe under<br>DESCRIPTION OF OPERATIONS below  |                            |  | outh oth country               |  |                             | E.L. DISEASE - POLICY LIMIT                  | \$1,000            |   |
| B D&O w/EPLI   |                            |  | G25021903002                   | 08/01/2012                                       | 08/01/2013                  |  |                    |   |
|  |                            | ł  |                                |  |                             | \$100,000 on EPLI                            |                    |   |
| C Fidelity   |                            |  | 10567717                       |  |                             | \$1,000,000 w/\$2500                         | ) Ded              |   |
| DESCRIPTION OF OPERATIONS / LOCATIONS / VEHIC<br>Education Service Center of Lake Ei   |                            |  |                                |  |                             | d se reenacte                                |                    |   |
| General Liability (CG2029) and D&O   |                            |  | _                              |  |                             | u us respectes                               |                    |   |
| Umbrella is Excess over: General Li  | •                          |  |                                | Ohio Employers                                   | Stop Gap                    |  |                    |   |
|  | -                          |  | -                              |  | • •                         |  |                    |   |
| Alliance Academy of Cincinnati, Apex Academy, Emerson Academy of Dayton, North Dayton School of Discovery,<br>Orion Academy, Pathway School of Discovery, Pinnacle Academy, Winterfield Venture Academy  |                            |  |                                |  |                             |  |                    |   |
| CERTIFICATE HOLDER   |                            |  |                                | CANCELLATION                                     | 1                           |  |                    |   |
| Education Service Cente  | -                          |  |                                |  |                             |  | NCELLE             | D BEFORE                                |
| 2275 Collingwood Blvd  | er of                      |  |                                | THE EXPIRATION                                   |                             | REOF, NOTICE WILL B<br>LICY PROVISIONS.      |                    | /ERED IN                                |

|                           |   | © 1988-2010 ACORD CORPORATION. All rights reserved. |
|---------------------------|---|---|
| ACORD 25 (2010/05) 1 of 1 | The ACORD name and logo are registered marks of | ACORD   |
| #S555530/M552065          |   | JMH   |

La.

Edenal M. B.

# ATTACHMENT 11.19 Board Resolutions Approving Contract

## PATHWAY SCHOOL OF DISCOVERY Board Meeting Minutes DRAFT Wednesday, June 6, 2012 at 4:00 PM

Meeting held at: North Dayton School of Discovery 3901 Turner Road Dayton, OH 45415

BOARD OF DIRECTORS:

PRESENT

UNABLE TO ATTEND

| Paige Bartlett – Secretary           |   | X |
|--------------------------------------|---|---|
| Susan Garretson – Director           | X |   |
| Duane Martin – Vice President        | X |   |
| Kevin A. Roble – President/Treasurer | X |   |
| Kay Wick – Director                  | X |   |

Non-Board Members Attending:

- 1. Anne Barnett ESC of Lake Erie West
- 2 Andrea Townsend Principal at North Dayton School of Discovery
- 3. Keith Colbert Principal
- 4. Brooke Wright 3<sup>rd</sup>-5<sup>th</sup> Grade Dean at North Dayton School of Discovery
- 5. Greg Dennis NHA
- 1. CALL TO ORDER

Kevin A. Robie called the meeting to order at 5:20 PM.

2. ROLL CALL

Roll call was held and a quorum was present.

3. MANAGEMENT REPORTS

Principal Report/School Leadership Team Report Keith Colbert presented the Principal Report and the School Leadership Team Report.

i. Enrollment Update

The Enrollment Update was presented. The school expects to be at or near full enrollment in the fall.

## PATHWAY SCHOOL OF DISCOVERY Board Meeting Minutes DRAFT Wednesday, June 6, 2012 at 4:00 PM

School Performance Dashboard Suite

The School Performance Dashboard Suite was presented.

i. The Proficiency & Growth were presented.

Board Meeting Attendance Report The Board Meeting Attendance Report was reported.

Race to the Top Update Greg Dennis and Keith Colbert presented the Race to the Top Update.

Board Fund Report The Board Fund balance was reported.

4. DISCUSSION ITEM(S)

None.

5. ANNUAL MEETING ACTION ITEM(S)

<u>Approval of the 2012-2013 Board Meetings Calendar</u> A motion was made by Susan Garretson and supported by Kay Wick to approve the 2012-2013 Board Meetings Calendar as submitted. The motion was approved unanimously.

6. ACTION ITEM(S)

<u>Approval of the May 2, 2012 Board Meeting Minutes</u> A motion was made by Duane Martin and supported by Kay Wick to approve the May 2, 2012 Board Meeting Minutes as submitted. The motion was approved unanimously.

Authorization of Board President to Execute Charter Contract A motion was made by Duane Martin and supported by Kay Wick to Authorize the Board President to Execute the Charter Contract upon completion. The motion was approved unanimously.

Approval of the 2012-2013 Parent Student Handbook A motion was made by Duane Martin and supported by Kay Wick to approve the 2012-2013 Parent Student Handbook as presented and to authorize-the principal to finalize the Handbook with the understanding that any substantial changes shall be presented to the Board for approval. The motion was approved unanimously.

7. NEW BUSINESS

Greg Dennis gave a legislative update regarding Senate Bill 316 which aims to make multiple changes to state education requirements.

PATHWAY SCHOOL OF DISCOVERY Board Meeting Minutes DRAFT Wednesday, June 6, 2012 at 4:00 PM

8. PUBLIC COMMENT (2 Minutes)

None was given.

9. MEETING ADJOURNMENT

The meeting was adjourned at 5:55 PM.

Next Meeting: August 15, 2012 at 4:00 PM

#### **OFFICER OF THE BOARD**

SIGNATURE